

UNIVERSITÀ DEGLI STUDI DI TORINO

# EUROPEAN JOURNAL OF ISLAMIC FINANCE (EJIF)

ISSN 2421-2172

**VOLUME 9 - ISSUE N°1 (2022)** 

A JOURNAL DEDICATED TO THE INVESTIGATION OF ISLAMIC BUSINESS AND MANAGEMENT, FINANCE AND BANKING STUDIES

FULLY OPEN-ACCESS AND AVAILABLE ONLINE AT: <u>HTTPS://WWW.OJS.UNITO.</u> <u>IT/INDEX.PHP/EJIF/INDEX</u>





EJIF European Journal of Islamic Finance

## FOCUS AND SCOPE

The European Journal of Islamic Finance is an Open Access online journal that publishes quarterly (every four months) articles and papers that contribute to developing both the theory and practice of research methods employed across the whole field of Islamic Business and Management, Finance and Banking studies. The readers of this journal mainly include scientific workers, college teachers and PhD students and other practitioners.



EJIF European Journal of Islamic Finance

## EDITOR-IN-CHIEF

Prof. Paolo Biancone, University of Turin, Italy

## ASSOCIATE EDITORS

- Prof. Burhan Uluyol, Istanbul Sabahattin Zaim University, Istanbul, Turkey.
- Prof. Kabir Hassan, University of New Orleans, USA.
- Prof. Irum Saba Institute of Business Administration (IBA) - Karachi, Pakistan, Kuala Lumpur, Malaysia.
- Prof. Anwar Hasan Abdullah Othman -International Islamic University of Malaysia, Kuala Lumpur, Malaysia.
- Prof. Hishamuddin Abdul Wahab University Sains Islam Malaysia (USIM), Kuala Lumpur, Malaysia.



## EDITORIAL BOARD

- Prof. Dian Masyita, University of Padjadjaran, Indonesia.
- Prof. Abdulazeem Abozaid, Qatar Faculty of Islamic Studies - Qatar.
- Prof. Ahmad Aref Almazari, King Saud University, Saudi Arabia.
- Prof. Bashar Malkawi, University of Sharjah, UAE.
- Prof. Marco Meneguzzo, Università della Svizzera Italiana, Switzerland.
- Prof. Nidal A. Alsayyed, Inayah Islamic Finance Research Institute, USA.
- Prof. Roberta Aluffi, University of Turin -Italy.
- Prof. Ghassen Bouslama, NEOMA Business
   School Campus de Reims, France.
- Prof. Nazam Dzolkarnaini, Salford University, UK.
- Prof Rusni Hassan International Islamic University of Malaysia, Kuala Lumpur, Malaysia.
- Prof. Khaled Hussainey, University of Portsmouth, UK.
- Prof Mansor Ibrahim INCEIF, Kuala Lumpur, Malaysia.
- Prof. Rifki Ismal, University of Indonesia.



## EDITORIAL BOARD

- Prof. Tariqullah Khan, Hamad bin Khalifa University, Qatar.
- Prof. Ali Khorshid, ICMA Centre Reading University - UK.
- Prof. Amir Kia, Utah Valley University, USA.
- Prof. Laurent Marliere, Université Paris-Dauphine France.
- Prof. Federica Miglietta, University of Bari -Italy.
- Prof. Hakim Ben Othman, University of Tunis -Tunisia.
- Prof. Mohamed Ramady, King Fahd University of Petroleum and Minerals, Saudi Arabia.
- Prof. Mamunur Rashid, Nottingham University, Malaysia.
- Prof. Roberta Ricucci, University of Turin, Italy.
- Prof. Silvana Secinaro, University of Turin, Italy.
- Prof. Younes Soualhi, International Islamic University Malaysia.
- Prof. Metin Toprak Istanbul Sabahattin Zaim University, Istanbul, Turkey.
- Prof. Laurent Weill, University of Strasbourg, France.



EJIF European Journal of Islamic Finance

## JOURNAL MANAGERS

- Dr. Valerio Brescia, University of Turin, Italy.
- Dr. Davide Calandra, University of Turin, Italy.
- Mr. Federico Lanzalonga, University of Turin, Italy.

## COPYEDITING TEAM

- Mrs. Federica Bassano
- Dr. Federico Chmet
- Dr. Daniel lannaci

### NATIONAL RESEARCH ACCREDITATION

 EJIF is listed in the ANVUR (The Italian National Agency for the Evaluation of Universities and Research Institutes) as a scientific journal in both AREA 12 - Legal sciences and AREA 13 - Economic and statistical sciences.



EJIF European Journal of Islamic Finance

## INDEXING

- ACNP
- Arab World Research Source: Al-Masdar
- **BASE**
- DOAJ
- CiteFactor
- EBSCO Discovery Service
- Google Scholar
- Microfinance Gateway (World Bank)
- ROAD Directory of Open Access Scholarly Resources

## PEER-REVIEW PROCESS CERTIFIED BY

ReviewerCredits

# Understanding Volatility dependence between MENA Sukuk, GCC Sukuk and Nifty Shariah Index during Covid-19: A C-vine Copula Approach

Saif Siddiqui<sup>1</sup>, Ziya Batool Rizvi

<sup>1</sup>Centre for Management Studies, Jamia Millia Islamia (A Central University), New Delhi

Contact the Author: ssiddiqui1@jmi.ac.in

Abstract- The study aims to identify the dependence between MENA Sukuk, GCC Sukuk, and Nifty 50 Shariah indices. Further, it finds out volatility patterns among the same indices before and during the COVID-19 pandemic. Daily data (April 2017 to November 2020) of the indices are analyzed using the GARCH model and C-Vine Copula approach. It is done by modelling the returns of the indices. The outcomes may provide a prodigious advantage for portfolio makers and stockholders towards investment policies throughout financial catastrophes like the COVID19 pandemic. In the pre-Covid period- C Vine copula shows that the Sukuk have strong dependence while the Shariah index has rather weak dependence, for MENA Sukuk and Nifty Shariah 50 it increased by a great margin in during-COVID 19 periods. GCC Sukuk and Nifty Shariah indices are positively correlated to each other in comparison to the returns of Mena Sukuk. The results of the GARCH model show asymmetrical co movements for losses and gains. Moreover, conditioned on MENA Sukuk, the GCC Sukuk and Nifty Shariah 50 had a higher negative degree of dependence within and throughout the COVID-19 period. The findings show substantial high and low tail dependency among the Sukuk and Shariah markets before Covid19. GCC Sukuk and Nifty Shariah 50 indices are highly positively correlated to each other in returns of MENA Sukuk. We also find negative dependency among the GCC Sukuk and Nifty Shariah 50 during Covid19 with MENA Sukuk. The outcomes are varied due to the time variation copulas that shows dependency diverges over time for all variables. Furthermore, this study would be helpful to find out the significance of ethical finance functions safe havens for world investors by using the copula model.

Keywords: Covid-19, Dependency, Volatility, Sukuk, Index, Copula, GARCH

INTRODUCTION

I.

The outbreak of the pandemic of coronavirus (COVID-19) is a widespread economic challenge and a major influencing factor in the current time. The novel disease has unfolded quickly athwart boundaries with quite 5,049, 497 individuals confirmed the infection and therefore the deaths of 367,230 people in more than 195 countries around the world, carrying an average death rate of 6.07% below 1% mortality from flu (Gormsen & Koijen, 2020; Sherif, 2020), which has affected the Islamic finance market too.

The major contributor to the recent growth of the Islamic finance market is Sukuk, the Islamic bond. These are structured to generate returns for investors without violating Shariah. The Sukuk market has shown remarkable growth in the first half of 2017. It has developed from just \$ 200bn in 2003 to projected \$ 4 trillion by the year 2030 (Alam & Seifzadeh, 2020). Such exceptional development has been grown up in non-Muslim nations in Asia and Europe (Alam, 2019). This is because of the cultural diversity with enormous numbers of Muslim immigrants (Alam & Seifzadeh, 2020).

Empowered by the above contentions and the possible effect of the current flare-up of the COVID-19 pandemic, this study looks at and gives new proof on the impact of COVID-19 on the S&P Dow Jones MENA Sukuk and GCC Sukuk disparity with the Nifty Shariah 50 record and examines the volatility pattern of the S&P Middle East and North Africa (MENA) Sukuk, Gulf Cooperation Council (GCC) and Nifty Shariah 50 indices.

The key motivation for this research study is to analyze the volatility and dependency structure for Shariah and Sukuk index before and during the Covid-19 pandemic. Shariah market, including Sukuk, is inclined by the situation of financial market circumstances specifically the volatility and returns. The statistic recounts the unrivalled volatility in Sukuk and Shariah indices throughout Covid-19.

European Journal of Islamic Finance - ISSN: 2421-2172

This paper also focuses on the factors of coupling and decoupling of Sukuk and Shariah index before and during the pandemic in association with globally perilous issues given joint dependency of return disseminations together in tails (extremes) and the focus.

Additionally, we enhance contribution and originality via associating the volatility of two Sukuk and a Shariah market, which gives a good insight of their comparative upended before and during the pandemic period. These findings provide better perceptions in understanding the volatilities. The outcomes may provide a prodigious advantage for portfolio makers and stockholders towards investment policies throughout financial catastrophes like to COVID19 pandemic. Apart from this, there is a dearth of studies on the S&P Middle East and North Africa (MENA) Sukuk, GCC Sukuk and Nifty Shariah 50 indices together.

This study adopts a copula approach that offers great flexibility in separating the marginal distributions from the dependence structure and in modeling these distributions independently to provide information on average dependence as well as on the probability that two variables jointly experience extreme upwards or downwards movements.

This paper is organized as follows. Section 2 reviews the documents available in the same way and/or related to Islamic funds. Section 3 describes the data source and method used for the current study. Section 4 shows the strong results of the analysis and finally, section 5 concludes the paper by giving the real and strategy suggestions and delineation for future study.

#### II. LITERATURE REVIEW

The studies that have examined Sukuk indices are few. The empirical studies based on Islamic indices, conducted around the world, collected from various sources, are presented in Appendix 1. From the review of literature, it can be interpreted that the co-integration model is the widely used tool for analyzing Shariah indices, followed by Multivariate GARCH, CAPM and T-test. There are a few articles that have used other tests like Copula, wavelet analysis and GMM. There are a few articles that have used other tests like wavelet analysis and GMM.

This paper subsides the existing literature by investigating the dependency structure among Sukuk and Nifty Shariah. So, we used various copulas which help to find out any variations in dependency structure over time variations. It is notified that most of the outcomes imply that the relationship between Sukuk and Shariah index counterpart by pretentious a continuous and symmetrical collaboration among them, most of the studies reflect the strong dependency between Sukuk and Shariah stocks. In our best information and considering the research gap, there is no research on GCC, MENA Sukuk and Nifty Shariah 50 index based on copula approach. The study intends to find out the modelled return in indexed based investments, primarily in Sukuk before and during COVID-19.

#### III. METHODOLOGY

#### 3.1 Data and Hypotheses

As stated earlier, the data consists of daily returns of three indices, namely, Dow Jones MENA Sukuk (MS), S&P GCC Sukuk (GS) and Nifty Shariah 50 index (NF). The empirical analysis is conducted on daily data obtained for the period 1st April 2017 to 30th November 2020 for benchmark indices. To examine the link between Sukuk and Nifty Shariah, we consider the daily closing prices of the nearest contract to maturity on the Sukuk and Shariah. We choose data from 2017 to 2020 to analyze the pre and during COVID-19 pandemic situation as it was earlier observed that there was the highest issuance of Sukuk after financial crises.

The Dow Jones MENA Sukuk consists of US dollardenominated investment-grade Sukuk (Islamic bonds) issued in the MENA region, which are Shariah-compliant. The S&P GCC Sukuk consists of US dollar-denominated investmentgrade Sukuk from GCC countries of risk. The Nifty Shariah 50 consists of investors with Shariah-compliant investment solutions. The data is taken from the official websites of S&P Dow Jones Indicesand Nifty indices (www.spindices.com and www.niftyindices.com). (See Appendix 1)

#### Null Hypotheses

For this purpose, two null hypotheses are put as follows:

H01: There is no significant difference in dependency structure among GCC, MENA Sukuk and Nifty Shariah 50 before and during the COVID-19 pandemic is not the same.

H02: There is no significant difference in the volatility in Sukuk and Nifty Shariah 50 before and during COVID-19 pandemic.

This includes various copulas with different tail dependency structures like Student-t (symmetric dependency), Rotated Gumbel 180(Lower and Upper tail dependency), Gaussian (Upper tail dependency) and Frank (symmetric tail dependency). Intended for a better understanding, we analyze two global Sukuk indexes namely, MENA and GCC Sukuk and Nifty Shariah 50 index by considering the data from 1st April'17 to 30th November'20. Outcomes represent that both Sukuk and Nifty Shariah50 indices are highly affected by the pandemic. Additionally, results proposed that Sukuk and Shariah index are sturdily related and inclined to co-move during the Covid19 crisis, which proves the coupling hypothesis of Sukuk from Shariah.

Our outcomes have several significant consequences and proposed some prominent contributions to the existing literature by examining the co-movements of the dependency structure of Shariah and Sukuk market through the Copula approach before and during the COVID-19 pandemic. For explaining the associations among Sukuk and Shariah Index we build a dynamic and static dependency structure through Copula. To our best knowledge, this research is the original one that examines the volatility and dependency among Sukuk (MENA and GCC Sukuk) and Shariah (Nifty 50) index before and throughout the COVID-19 crisis.

European Journal of Islamic Finance - ISSN: 2421-2172

#### TABLE I. DATA SUMMARY

Index	Launch date	First date	Countri es conside red	Symb ols used at levels	Symbol s used at first differen ce	
S&P MENA Sukuk Index	September 20, 2013	July 31, 2013	The Middle East and North Africa	MS		
GCC Sukuk	March 22, 2018	December 31, 2012	Bahrain, Kuwait, Oman, Katar, Saudi Arabia, and UAE	GS	RGS	
Nifty Shariah 50	February 19, 2008	December 26, 2008	India	NS	RNF	

All the data series are considered after computing their log returns using the formula for calculating return:

 $rt = \ln (Pt / P(t-1)).$ 

Here, rt and Pt addresses the everyday return at the business day t individually.

All data series are retrieved from Bloomberg, S&P Dow Jones and Nifty Shariah. The explanation behind using the Dow Jones list is for normalization in the total of indices value. Next, the S&P MENA is used for global Islamic fixed income. The S&P index measures the performance of MENA Sukuk around the world which is involved in the Middle East and African market. Then, the Bloomberg index is used to reflect the GCC Sukuk. GCC Sukuktracks the performance of 6 countries Sovereign Sukuk that are globally traded.

Lastly, Nifty Shariah 50 indices are used for reflecting the uncertainty in Sukuk. This study has divided the observation into the Before-COVID 19 and the During-COVID 19 period to examine the difference between Pre and During COVID-19. World Health Organization (WHO) issued the first Disease Outbreak News Reporting 2019 globally.

Hence, we took the beginning of the COVID-19 from November 2019 as global and defined the 1st November 2019-30th November 2020 as during the COVID-19 period. The period from 1st April 2017 to 31st October 2019 is defined as the before COVID-19 period.

In the methodology part, first, we provided a brief introduction of the models used for the distribution of the margins. Then, a brief description of the approach used for dependence.

Figure 1a-c presents pre COVID-19 and 2a-c during COVID-19-time plots of the price series and Figure 3a-c presents pre COVID-19 and 4a-c during COVID-19 presents the time plot of return series for the three sample indices.



Figure 1a: Price series from 1st April 2017 to 31st October 2019. Source: Authors' elaboration



Figures 1a, 1b and 1c (Pre Covid-19 Period) show that NF was at the level of 2150 as of April 1st, 2017, as it was initiated way before the other two indices. It rose and fell due to America's trade war with China and again rose and finished at the rose level from starting (at increasing rate up to 50 per cent)

Authors' elaboration

European Journal of Islamic Finance - ISSN: 2421-2172

after 2.5 years. While, the other two were initiated on April 1st, 2017, at the level of 117.5 and 115. Both GS and MS show a rising trend.

#### **PRICE SERIES DURING COVID-19**



Source: Authors' elaboration

Figures 2a, 2b and 2c show that NF was at the level of 2500 as of November 1st, 2019, as it was initiated way before the

other two indices. It rose and fell due to the Covid-19 pandemic and lockdown situation in the world. Investors do not have the funds to invest in the stock market which affect the global economy but once the lockdown was lifted from the economy and things normalize then again investors invest into the stock market so the price rose and finished more it's starting (increased 20 per cent) level. The other two were initiated i.e., MS and GS on November 1st, 2019, at the level of 128 and 133. Both MS and GS show a rising trend (increase 7.81 and 6.76 per cent).



Figure 3 & 4 shows the return series for the three indices and it can be observed that the return of during Covid-19 of MS (RMS), GS (RGS) and NS (RNF) having higher fluctuations as compared to returns of Pre Covid-19 period of MS (RMS), GS (RGS) and NS (RNF) while, RMS and RGS are leaping around the same path. This was due to the spread of the corona virus pandemic from China into the rest of the world, and the economic shutdown and investors were worried about the impact of the COVID-19. Hence, it can be observed from the graph that NS is the most volatile out of the three indices. Outliers are also visible.

European Journal of Islamic Finance - ISSN: 2421-2172

Tables 3 & 4 reports summary statistics for the daily returns of the indices considered as well as statistics testing for normality and independence.

#### 3.2 Preliminary results

Preliminary analysis and results are presented as follows:

#### Descriptive Statistics before Covid-19

Descriptive statistics of Before Covid-19 period are presented in table 2.

TABLE II.	DESCRIPTIVE	STATISTICS	BEFORE	COVID	PERIOD (3 <sup>F</sup>	D APRIL
$2017 - 31^{st}$ OCT	TOBER 2019)		///	21		

	MS	GS	NF
Mean 0.000173		0.000177	0.000215
<b>Median</b> 0.000171		0.000166	0.000146
Minimum -0.00403		-0.004	-0.05156
<b>Maximum</b> 0.004697		0.004318	0.027081
Std.Deviation	0.001011	0.000998	0.008275
<b>Skewness</b> 0.153036		0.149481	-0.63146
Kurtosis	4.247265	4.153168	6.588454
Jarque-Bera	46.0444	39.61861	404.0084
Probability 0.000000		0.000000	0.000000
Observation	670	670	670

Source: Authors' elaboration

Table 2 premises the descriptive properties before the Covid-19 period. The sample means are positive for MS, GS, and NF in the Pre Covid-19 period, whereas the standard deviation is lowest for GS (00.09 per cent) and highest for MS (00.10 per cent). Thus, MS is found to be the most volatile index followed by GS and NS.

Skewness and kurtosis indicate that all series are completely skewed except Nifty Shariah 50 it skewed negatively and extremely leptokurtic. The values of skewness and kurtosis reveal that all indices follow the traditional distribution that is other verified by the value of Jarque-Bera datum and prospect value. As higher kurtosis relates to the extremity of outliers, which is highest for NF distribution followed by MS and GS, respectively. Descriptive Statistics for During Covid-19

Descriptive statistics of During Covid-19 period are presented in table 3.

TABLE III. Descriptive statistics during covid period (1st november 2019 – 30th november 2020)

4	MS	GS	NF
Mean	0.000251	0.000244	0.000683
Median	0.000375	0.000372	0.001353
Minimum	-0.01172	-0.011702	-0.10902
Maximum	0.006651	0.006697	0.089551
Std.Deviation	0.001935	0.001937	0.018286
Skewness	-1.99655	-1.98831	-0.8989
Kurtosis	13.6984	13.67204	13.90106
Jarque-Bera	1515.907	1507.83	1195.223
Probability	0.000000	0.000000	0.000000
Observation	279	279	235
Source: Authors	' elaboration		

Table 3 summarizes the descriptive properties of the variables during the Covid-19 period. The sample means are positive for MS, GS, and NF during the Covid-19 period, whereas the standard deviation is lowest for MS (00.19 per cent) and highest for NS (01.82 per cent). Thus, NF is found to be the most volatile index followed by GS and MS.

The values of skewness and kurtosis reveal that all indices don't follow the normal distribution. The same is given by Jarque-Bera test. As higher kurtosis relates to the extremity of outliers, which is highest for NS distribution followed by MS and GS, respectively.

After the comparison of Before COVID-19 and During the COVID-19 period, all variables, except MS have a positive return. The volatility of each variable has increased in both periods under study. NS has shown a higher mean return in the COVID-19 period. The maximum and minimum returns during and after the COVID period show the highest peak. Standard deviation is highly volatile during the COVID-19 period in comparison to pre-COVID-19 period. All variables are somewhat asymmetric as shown by their non-zero skewness coefficients.

European Journal of Islamic Finance - ISSN: 2421-2172

#### IV. RESULTS

#### 4.1 GARCH modelling

To comprehend the impulsiveness of the securities market, we will in general allow Exponential GARCH models that area unit broadly utilized in learning the variability of the financial market in money writing have depended on uneven GARCH model developed by Admiral Nelson (1991) recommending the following work of ARMA GARCH model for volatilities (D'Ecclesia & Clementi, 2019). The ARMA approach created by Box and Jenkins (1976) may be a class of random models needed to investigate measurement information considering the ensuing autoregressive moving normal model meant as ARMA (p,q).

 $y_t = \delta + \sum_{i=1}^p \phi_i y_{t-i}$ 

(1)

Where  $\delta$  may be a constant term,  $\phi i$  the ith autoregressive constant,  $\theta j$  the jth moving average constant, and  $\varepsilon$  terror term at time t. p and q are known as the sets of autoregressive and moving normal terms, severally. when the backshift administrator B is applied, Eq. (1) are regularly composed as:

θ.

(2)  

$$(1-\sum_{i=1}^{p}\phi_{j}B^{i}) \quad (y_{t-\mu}) = (1+\sum_{j=1}^{q}\theta_{j}B^{j}) \quad \varepsilon_{t}$$
where B(yt-\mu) =yt-1-\mu and B\varepsilon\_t=\varepsilon\_t -1.

It is used to explore what proportion of Sukuk influenced the market volatility Pre and during the COVID-19 crisis.

In this study, the ARMA-GARCH method is used to assess the parameters of the marginal. The best-fitting model was selected among completely different lag orders (ARMA (0, 0), ARMA (0,1), ARMA (1,0), ARMA (1,1)) with the error term following the traditional, student's t or Generalized Error Distribution (GED).

The estimated parameters for the select models are reported in Tables 4 & 5.

TABLE IV.	ESTIMATES OF MARGINAL DISTRIBUTIONS - BEFORE COVID-
19	

	MS	GS	NF
	GED	GED	GED
μ	0.00014***	0.00178** *	0.000237
	(0.000046)	(0.0000)	(0.32956)
λ	-0.146645 ***	-0.149879 ***	0.053997
	(0.000054)	(0.000136)	(0.21395)
δ	-	-	-
	-	-	-

	MS	GS	NF
ω	0.000000	0.000000	0.000004 ***
	(0.958645)	(0.845614)	(0.00000)
α	0.025792 ***	0.033352 ***	0.048261 ***
	(0.002054)	(0.001945)	(0.00000)
β	0.961937 ***	0.950449 ***	0.896015 ***
	(0.00000)	(0.00000)	(0.00000)
Shape	1.378927 ***	1.387345	1.238533 ***
	(0.00000)	(0.518705)	(0.00000)

Source: Authors' elaboration

\* The table gives the constraint estimations of marginal distribution function along with standard error and p-value of GARCH test for the suitability of distribution function. The constraints of peripheral distribution function are given in eq. (1&2). For GARCH testing, p-values which are lesser than 0.05 shows that null hypothesis is rejected. \*, \*\*, \*\*\* shows statistical consequence at 10%, 5% and 1% level correspondingly.

 
 TABLE V.
 ESTIMATES OF MARGINAL DISTRIBUTIONS - DURING COVID-19

19	1 11		
25	MS	GS	NF
2	STD	STD	STD
м	0.000329 ***	0.000313 ***	0.001102 ***
NG4	(0.002897)	(0.004982)	(0.0000)
4	0.856217 ***	0.856441 ***	0.397966 ***
7 7	(0.00000)	(0.00000)	(0.00000)
Δ	0.730880 ***	-0.730492 ***	-0.343426 ***
	(0.00000)	(0.0000)	(0.0000)
Ω	0.000000	0.000000	0.000007
	(0.955101)	(0.958515)	(0.513746)
α	0.140091 **	0.138502 **	0.147472 *
	(0.015616)	(0.018440)	(0.077205)
β	0.829632 ***	0.833673 ***	0.818538 ***
	0.00000)	(0.00000)	(0.00000)

European Journal of Islamic Finance - ISSN: 2421-2172

	MS	GS	NF
Shape	8.602566 **	8.456019 **	0.988159 ***
	0.023592)	(0.021254)	(0.00000)

Source: Authors' elaboration

\* The table gives the constraint estimations of marginal distribution function along with standard error and p-value of GARCH test for the suitability of distribution function. The Constraints of peripheral distribution function are given in eq. (1&2). For GARCH testing, p-values which are lesser than 0.05 shows that null hypothesis is rejected. \*, \*\*, \*\*\* shows statistical consequence at 10%, 5% and 1% level correspondingly.

As shown in the above Tables, all the return series are best captured either by the GED or student t distribution based on AIC. The Autoregressive and Moving Average (ARMA) terms are statistically significant for the selected variables. The ARCH coefficients represented by  $\alpha$ , exhibits the influence of past squared residuals. On the other hand, the GARCH coefficient represented by  $\beta$  shows the impact of lagged forecasted variance. All coefficients are statistically significant at 1% for all series.

Further, the sum of ARCH and GARCH coefficients is closer to unity which showing that the shocks are persistent. Shape parameters are statistically significant indicating that the return series was asymmetric.

Lastly, the diagnostic tests could not reject the null hypothesis of no ARCH effect and no serial correlation.

#### 4.2 C-vine Copula Approach

A Copula is a variable accumulative distribution operate of which marginal distribution is uniform on the interval [0,1] and it captures the dependence structure of a variable distribution. (Rüschendorf, 2009). Multi-data distribution methods have high flexibility as a result of it permits scholars to shape the marginal CDFs aboard a dependable structure expressed in Copula's work. In alternative words, through Copula, we will split the combined allocation operation Fxy into a phase that explains the reciprocity among the random variables X and Y and also the classes solely describe a marginal behaviour.

According to Sklar's Theorem, there exists a copula perform C (.) such for all  $x,y \in [-a,a]$ :

$$F_{XY}(x,y) = C(F_X(x),F_Y(y))$$
(3)

Where Fx(x) and Fy(y) denote the marginal distribution functions

The theory additionally states that joint distribution is given Fxy performance; copula is exclusive for  $Fx \times range$  Fy range, which may be a set of the ranges of the marginal CDFs. This suggests that the verb is going to be distinctive on the off chance that the marginal Fx and Fy are persistent. Depending on the formulation, copula could be used to connect marginal to a multivariate of distribution functions, which could also

deteriorate into its univariate minimal appropriation and copula catches the reliance structure.

On account of bivariate appropriation incorporates a thickness Fxy(x,y), and this is accessible, further:

$$F_{XY}(x, y) = C (F_X(x), F_Y(y)) \cdot f_X(x) \cdot f_Y(y)$$
(4)

Where c(.) is the depth of the copula.

A necessary property of the copula is that it provides important details associated with the mean dependence referenced and tail dependence (excessive dependence), that tests the chance of the two markets can collectively together expertise extreme top or drawback worth movements. Unsurprisingly, the higher tail dependence suggests the relative quantity of size inside the higher (lower) portion of the quartile distribution.

Attained from copula, higher and smaller tail Dependency processes area unit given as:

$$\lambda_{U} = \Pr\left[X \ge F_{X}^{-1}(u) \mid Y \ge F_{Y}^{-1}(u)\right] = \lim_{u \to 1} \frac{\lim_{u \to 1} \frac{(1-2u+\mathcal{C}(u,u))}{1-u}}{(5)}$$

$$\lambda_{L} = \Pr\left[X \le F_{X}^{-1}(u) \mid Y \le F_{Y}^{-1}(u)\right] = \lim_{u \to 0} \frac{\mathcal{C}(u,u)}{u}$$
(6)

where  $0 \le u \le 1$ ,  $0 \le \lambda_u$ ,  $\lambda_u \le 1$ . If  $\lambda_L > 0$  ( $\lambda_L > 0$ ), then variables X and Y will in general be lower (upper) tail subordinate. This suggests a non-zero likelihood of noticing a tiny (huge) esteem for one arrangement with a minuscule (enormous) esteem for another arrangement (Mensi, Hammoudeh, Shahzad, & Shahbaz, 2017).

Further, different copula families are used to study the dependence between the variables. Therefore, standard residuals from the GARCH models are transformed into uniform distribution using ECDF.

#### 4.2.1 Copula Functions

This paper uses a varied family of copula functions that have exceptional dependency structures. It includes Student-t (symmetric dependency), Rotated Gumbel 180(Lower and Upper tail dependency), Gaussian (Upper tail dependency) and Frank (symmetric tail dependency). The characteristics of these Copulas are specified below:

The rotated Gumbel copula has an only a lower tail dependency.

$$CRG(u1, u2; \delta) = u1 + u2 - 1 + CG(1 - u1, 1 - u2; \delta).$$
(7)

Student t-Copula shows only symmetric tail dependency.

Cn.p (u,v) = 
$$\int_{-\infty}^{t^{-1}(u)} n \int_{-\infty}^{t_n^{-1}(v)} \frac{1}{2\pi (1-p^2)^{1/2}} exp \left\{ 1 + \frac{s^2 - 2pst + t^2}{n(1-p^2)} \right\}_{dsdt}^{\frac{-n+2}{2}} dsdt \qquad (8)$$

European Journal of Islamic Finance - ISSN: 2421-2172

$$\lambda L = \lambda U = 2 + [1 - tn - 1\sqrt{(n+1)(1+p,(1-P))}]$$
(9)

Gaussian Copula has only Upper tail dependency.

$$C_R^{Gauss}(u) = \phi_R \left( \phi^{-1}(u_1) \dots \phi^{-1}(u_d) \right)$$
(10)

Frank copula has only symmetric tail dependency.  

$$-\frac{1}{\theta} \log \left[ 1 + \frac{(\exp(-\theta) - 1)(\exp(-\theta u) - 1)}{\exp(-\theta) - 1} \right]$$
(11)

Another phase is to identify a sequence of the variables in the C-Vine Copula structure. Here, MS is placed in the first place in the order of the variables. For the rest of the variables, this study has used the method proposed by (Panagiotelis, Czado, & Joe, 2012)for identifying the maximal tree spanning by arranging the variables following their sum of absolute

pairwise estimated Kendall's tau  $\hat{S}_i = \sum_{j=1}^d |\hat{T}_{i,j}|$  The variable with the highest dependency is placed at the next sequence and so on.

By this method, the order of the variable is decided as MS, GS, and NF. This sequence applies to the Pre and during COVID 19 periods datasets both.





Figure 6: Pairwise Scatter Plot (During Covid 19). Source: Authors' elaboration

Figures 5 and 6 show Kendall's  $\tau$  (above the diagonal) and pairwise scatter plot (below the diagonal) of the copula data. From the graph, it is evident that MS has strong dependence while NF has rather weak dependence on other indices. From the estimated Kendall's  $\tau$ , MS is the highest dependent variable with the rest of the variables as it has the greatest sum of Kendall's tau, which labels it as the second variable.

The next step involves choosing a suitable copula family for each pair and estimating their parameters accordingly. Tables 7 and 8 summarize the results of parameters estimated for the GED copula structure. The first tree reflects the degree and structure of dependence between MS and two other indices- GS and NF. The estimated Kendall's tau evidenced that the degree of dependence has increased between MS and the other two indices the COVID-19 period. This implies that GS and NS are more positively correlated with the movements in the returns of MS.

The dependence structure between GS and NF is shifted from rotated Gumbel 180 degrees to Frank copula. Though, both the copulas measure the lower tail dependence suggesting the downward movements in the index's returns are more correlated than upward movements.

In the view of symmetric tail dependency Student-t, Gaussian, rotated Gumbel and Frank copulas shows the dependency among Sukuk and Nifty Shariah index is optimistic and substantial, opposing the view towards the Sukuk as to be used as hedger and instrument of risk diversification in the risky market situation also. Additionally, the proof of optimistic symmetric upper and lower tail dependency given by Student- t copula suggests that Sukuk and Nifty Shariah index markets change collectively but at different periods before and during Covid19. For GCC and MENA Sukuk, there is a sign of optimistic dependency, even the tail dependency is lower, which shows the optimistic collective response towards the before Covid19 situation. For GS and NF, the symmetric copulas offer the sign of optimistic dependency by supportive sign of upper and lower tail dependency. Though, for GS and NF, we discover a sign of negative lower tail dependency with MS. So, with the features of oblique copulas, it gives the benefit for the diversified portfolio and managing the risk before the crisis.

Next, in the second tree, we tend to explore the dependence between GS and NF conditional on MS. Given the returns of MS, the worth of Kendall's tau between MS and NS increased by a positive margin duringCOVID-19 period. Also, the dependence degree of MS and NF is highest among all the pairs. It is to be noticed that there are asymmetrical comovements as extreme losses are more correlated than extreme gains depicted through lower tail dependence. Furthermore, conditioned on MS,GS and NF had a higher negative degree of dependence during the COVID-19 period. The increased negative dependence could be due to high volatility in the index (during the COVID period).

#### European Journal of Islamic Finance - ISSN: 2421-2172

	Pairs	Copula	Par1	Par2	$\lambda_L$	$\lambda_u$	Kendall's tau
Tree 1	MS-GS	Student t	0.9979992	2.83224	0.9536992	0.9536992	0.9597219
	MS-NF	Rotated	1.161558	-	0.1838131	0	0.1390873
		Gumbel 180					
Tree 2	GS-NF  MS	Frank	-0.101906	-	0	0	-0.0113283
Source: A	uthors' elaboration		1407	1 (			

For GS and MS, there is a sign of optimistic dependency, with symmetric upper and lower tail dependency, that shows the optimistic collective response towards the during Covid19 situation. For MS and NF, the symmetric copulas offer the sign of optimistic dependency by supportive sign of high lower tail dependency. Though, for GS and NF, the symmetric copula gives the sign of negative dependence with a higher and lower tail dependency with MS. It illustrates that dependency with globally risky issues changed but asymmetrical due to Covid19.

CONCLUSIONS

V.

The prodigious possible development and low apparently susceptibility of Nifty Shariah index as compared to Sukuk throughout the strained financial situation. It attracts the curiosity of Muslims, Conventional stockholders as well as portfolio managers globally. Due to the Covid19 pandemic, the relationship between Sukuk and Shariah index is analysed and concurrence rises on the basis of literature related to the dependency among Islamic finance and another index market, specifically in the risky market situation.

We examine the volatility and dependency among the GCC and MENA Sukuk, and Nifty Shariah 50 index market by using the varied family of copula methods having different dependency structures with different periods i.e., 1st April'17 to 31st Oct'19 as before Covid19 and 1st Nov'19 to 30th Nov'20, the practical outcomes show substantial high and low tail dependency among the Sukuk and Shariah market before Covid19.

GS and NF indices are highly positively correlated to each other in returns of MS. We also find negative dependency among the GS and NS during Covid19 with MS. The outcomes are varied due to the time variation copulas that shows dependency diverges over time for all variables. We find that Sukuk has consistent tail reliance as Bahrain, Oman, and Abu Dhabi; these discoveries counsel that these Gulf markets area unit getting constant direction and the same magnitude throughout the Covid-19 era additionally as these three also. There are asymmetrical co movements for losses and gains. Moreover, conditioned on MS, the GS and NF had a higher negative degree of dependence within and throughout the COVID-19 period. The enhanced negative dependence may be because of high volatility in the index (throughout the COVID period).

Hereafter, H01 was rejected because there is higher and lower tail dependency shown in Sukuk and Nifty Shariah 50 before Covid-19 and during COVID-19 but GS and NF indices are highly positively correlated and dependent to each other in returns of MS during COVID-19.

H02 was rejected because before COVID-19 Sukuk and Nifty Shariah 50 indices are less volatile due to the news effect but highly volatile during COVID-19. Sukuk and Nifty Shariah 50 stock market are Conditioned on MS, the GS and NF show high volatility in index.

Lastly, the outcome indicates some consequences and implications for portfolio managers and global stakeholders seeking interest to invest in the ethical stock market. Meanwhile, the ethical stock market is pessimistically impacted by discriminating dread, tension and concern in the conventional stock market, the hedging assets and hedging strategies seem to be safe in the ethical stock market. Furthermore, this study would be helpful to find out the significance of ethical finance functions safe havens for world investors by using the copula model.

Table 7 Pre Covid-19 period (3rd April 2017-31st Oct 2019)							
	Pairs	Copula	Par1	Par2	$\lambda_L$	$\lambda_u$	Kendall's tau
Tree 1	MS-GS	Student t	0.992768	2.0001	0.923475	0.923475	0.9233846
	MS-NF	Rotated	1.045677	-	0.0596446	0	0.043679
		Gumbel 180					
Tree 2	GS-NF  MS	Gaussian	-0.03402	-	0	0	-0.021667
Source: Au	uthors' elaboration						

European Journal of Islamic Finance - ISSN: 2421-2172

#### REFERENCES

- Abadi, R. T., & Silva, F. (2020). Do Islamic fundamental weighted indices outperform their conventional counterparts? An empirical investigation during the crises in the MENA region. Eurasian Economic Review, 1-26.
- [2] Akguc, S., & Al Rahahleh, N. (2018). Effect of shariah compliance on operating performance: evidence from GCC countries. Emerging Markets Finance and Trade, 54(12), 2874-2896.
- [3] Al-Khazali, O., Lean, H. H., & Samet, A. (2014). Do Islamic stock indexes outperform conventional stock indexes? A stochastic dominance approach. Pacific-Basin Finance Journal, 28, 29-46.
- [4] Alam, I. (2019). Interacting with Muslim customers for new service development in a non-Muslim majority country. Journal of Islamic Marketing.
- [5] Alam, I., & Seifzadeh, P. (2020). Marketing Islamic financial services: A review, critique, and agenda for future research. Journal of Risk and Financial Management, 13(1), 12.
- [6] Ali, H. (2020). Sukuk and International Financial Markets: Comovement Dynamics. CAPITAL UNIVERSITY.
- [7] Aloui, C., Hammoudeh, S., & Hamida, H. B. (2015a). Co-movement between sharia stocks and sukuk in the GCC markets: A time-frequency analysis. Journal of International Financial Markets, Institutions and Money, 34, 69-79.
- [8] Aloui, C., Hammoudeh, S., & Hamida, H. B. (2015b). Price discovery and regime shift behavior in the relationship between sharia stocks and sukuk: A two-state Markov switching analysis. Pacific-Basin Finance Journal, 34, 121-135.
- [9] Ashraf, S., & Marashdeh, H. (2018). Efficiency and Memory Dependence of Shariah Equity Markets. Journal of Islamic Economics, Banking and Finance, 113(6219), 1-13.
- [10] Bhuiyan, R. A., Puspa, M., Saiti, B., & Ghani, G. M. (2020). Comparative analysis between global sukuk and bond indices: value-atrisk approach. Journal of Islamic Accounting and Business Research.
- [11] Bhuiyan, R. A., Rahman, M. P., Saiti, B., & Ghani, G. B. M. (2019). Does the Malaysian Sovereign sukuk market offer portfolio diversification opportunities for global fixed-income investors? Evidence from wavelet coherence and multivariate-GARCH analyses. The North American Journal of Economics and Finance, 47, 675-687.
- [12] Boujlil, R., Hassan, M. K., & Grassa, R. (2020). Sovereign debt issuance choice: Sukuk vs conventional bonds. Journal of Islamic Monetary Economics and Finance, 6(2).
- [13] D'Ecclesia, R. L., & Clementi, D. (2019). Volatility in the stock market: ANN versus parametric models. Annals of Operations Research, 1-27.
- [14] Dharani, M., Narayanamoorthy, V., & Natarajan, P. (2011). A Study on the Performance of Islamic Index and the Common Index During Bull and Bear Market Period. Journal of Emerging Financial Markets, 2(1), 29-36.
- [15] El-Khatib, R., & Samet, A. (2020). Impact of COVID-19 on Emerging Markets. Available at SSRN 3685013.
- [16] El Khamlichi, A., Sarkar, K., Arouri, M., & Teulon, F. (2014). Are Islamic equity indices more efficient than their conventional counterparts? Evidence from major global index families. Journal of Applied Business Research (JABR), 30(4), 1137-1150.
- [17] Gormsen, N. J., & Koijen, R. S. (2020). Coronavirus: Impact on stock prices and growth expectations. The Review of Asset Pricing Studies, 10(4), 574-597.
- [18] Grassa, R., & Miniaoui, H. (2018). Corporate choice between conventional bond and Sukuk issuance evidence from GCC countries. Research in International Business and Finance, 45, 454-466.
- [19] Hassan, M. K., & Girard, E. (2011). Faith-based ethical investing: the case of Dow Jones Islamic indexes. Networks Financial Institute Working Paper(2011-WP), 06.
- [20] Irfan, M. (2020). Do Google Trends and Shariah Compliant Stocks Co-Integrated? An Evidence from India. International Journal of Islamic Economics and Finance, 3(2), 227-250.

- [21] Jawad, D., & Faris, H. (2019). SUKUK IN ISLAMIC FINANCE: A COMPARATIVE STUDY BETWEEN S&P MENA SUKUK AND S&P MENA BOND INDEX. European Journal of Economic and Financial Research.
- [22] Kumar, K. K., & Sahu, B. (2017). Dynamic linkages between macroeconomic factors and Islamic stock indices in a non-Islamic country India. The Journal of Developing Areas, 51(1), 193-205.
- [23] Mansor, F., & Bhatti, M. I. (2011). Risk and return analysis on performance of the Islamic mutual funds: evidence from Malaysia. Global Economy and Finance Journal, 4(1), 19-31.
- [24] Mensi, W., Hammoudeh, S., Shahzad, S. J. H., & Shahbaz, M. (2017). Modeling systemic risk and dependence structure between oil and stock markets using a variational mode decomposition-based copula method. Journal of Banking & Finance, 75, 258-279.
- [25] Munusamy, D., & Natarajan, P. (2011). Seasonal anomalies between S&P CNX nifty Shariah index and S&P CNX nifty index in India. Journal of Social and Development Sciences (JSDS), 1(3), 101-108.
- [26] Natarajan, P., & Dharani, M. (2012). Shariah Compliant Stocks in India: A Viable and Ethical Investment Vehicle\Oman Chapter of Arabian Journal of Business and Management Review.-2012, Vol. 1, No. 6, Pp. 50-62: Sohar University.
- [27] Panagiotelis, A., Czado, C., & Joe, H. (2012). Pair copula constructions for multivariate discrete data. Journal of the American Statistical Association, 107(499), 1063-1072.
- [28] Reddy, K., & Fu, M. (2014). Does shariah compliant stocks perform better than the conventional stocks? A comparative study of stocks listed on the Australian Stock Exchange. Asian Journal of Finance & Accounting, 6(2), 155-170.
- [29] Rüschendorf, L. (2009). On the distributional transform, Sklar's theorem, and the empirical copula process. Journal of statistical planning and inference, 139(11), 3921-3927.
- [30] Sadeghi, M. (2008). Financial performance of Shariah-compliant investment: evidence from Malaysian stock market. International Research Journal of Finance and Economics, 20(8), 15-24.
- [31] Sherif, M. (2020). The impact of Coronavirus (COVID-19) outbreak on faith-based investments: An original analysis. Journal of Behavioral and Experimental Finance, 28, 100403.
- [32] Siddiqui, S., & Sheikh, S. (2016). Modelling the return of shariah with underlying indices of national stock exchange of India: A case of 3SLS and GMM estimation. Journal of Emerging Economies and Islamic Research, 4(2), 1-15.
- [33] Tyagi, A., & Rizwan, M. (2012). A study of the movement of BSE-TASIS Shariah 50 index in accordance with Sensex. International Journal of Emerging Research in Management & Technology, 5-13.

Followed Appendix A.

European Journal of Islamic Finance - ISSN: 2421-2172

S. No	Authors (year)	Conceptual/ Data (Index and time period)	Methodology	Findings
1	Irfan (2020)	SENSEX, NIFTY (2010-2012)	Beta (CAPM), CAGR, Correlation, T-test	There is no difference between Shariah index and benchmark indices Return and conjointly in Shariah index and customary index. It found that equity primarily based Shariah index is social accountable to invest in investors especially small and private investors.
2	Sherif (2020)	UK Dow Jones index, FTSE100 index. Time period is from January'20 to May 20.	Regression	A well-structured and important statistical correlation among the COVID-19 epidemic and the performance of a general stock market indices. It has shown a negative impact compared to the market indicator of complaints. It has co-examined the effect of COVID- 19 on the performing of UK 10 sector clusters. The IT revenues of the IT sector have done much better than the market. return of shares in the field of consumer understanding,
	Grassa & Miniaoui (2018)	88 Sukuk and 287 conventional bonds of GCC countries were taken. The time period is from 2000 to 2015 (15 years)	Correlation, GMM	Prefer Sukuk in the issuance of large debt and long tenor, because it gives negative correlation in terms of credit rating.
4	Aloui, Hammoudeh, & Hamida, (2015)	GCC Shariah stocks and Sukuk. The time period is from 2008-2013 (6 years)	Markov, univariate and multivariate EGARCH	It propounds changes in sukuk price index having a considerable impact on chance of transmission across regime. It has many economic and managerial implications for Islamic portfolio managers, Islamic hedge funds, stock market regulators, and policy makers
5	Bhuiyan, Rahman, Saiti, & Ghani (2019)	Bond index (US, UK, Australia, Canada, Germany and Japan) and Malaysia sukuk index. The time period is from 2010 to 2015 (5 years)	Wavelet coherence, multivariate GARCH	Developed market shown lower co movement of returns between bond and Malaysian sukuk. Malaysian Sukuk shown negative correlation with bond market, good sign of diversification. It divulges alluring opportunity with credit quality to invest in fixed securities.
6	Dharani, Narayanamo orthy, & Natarajan( 2011)	Nifty Shariah Index, Nifty Index 2007 to 2010 (4 years)	Sharpe, Treynor, Jensen and T- test	It was found that Nifty Shariah underperformed throughout sample period. Each were underperforming and consistent with respect to risk free return. Nifty shariah is less volatile as compared to Nifty index.

#### APPENDIX A. REVIEW OF THE LITERATURE

European Journal of Islamic Finance - ISSN: 2421-2172

7	Aloui, Hammoudeh, & Hamida, (2015a)	GCC Sukuk indices and Shariah stocks. Daily data of 240 days.	Wavelet, VaR	It was found that there is strong dependency between shariah stocks and sukuk index. Portfolio diversifications also vary with frequencies and time.
8	Bhuiyan, Puspa, Saiti, & Ghani, (2020)	Sukuk and Bond indices. The time period is from 2010-2015 (6years)	VaR CCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCC	It analyses that VaR amount of sukuk indices is way under VaR of bond indices. Sukuk index with bond indices can cut back the VaR portfolio by 30-50%
9	Abadi & Silva, (2020)	MENA Region 2007-2015 (8 years)	Sharpe ratio, Multifactor Model	Performance measure and period of analysis affects the performance of Islamic and non-Islamic portfolios. Each portfolio found to be underperforming against the benchmark. Islamic portfolios underperformed in global financial crisis. non-Islamic counterparts throughout the Arab Spring their performance was at par
10	Jawad & Faris, (2019)	Mena region, S &P Sukuk and Bond index	GARCH, EGARCH, ARCH-LM	It founds that Sukuk reacts lesser to shocks in comparison to bond. Sukuk are more stable and have less risk.
	Ali, (2020)	Mena Sukuk, Mena Bond	ADRL, DCC GARCH, Copula	It founds that Mena Sukuk is correlated with Mena equity and Mena bond in short run and in long run it is correlated only with Mena.
12	Boujlil, Hassan, & Grassa, (2020)	143 Sukuk and 602 conventional sovereign bonds of 16 OIC countries were taken. The time period is from 2000 to the year 2015 (15 years)	Correlation and Logit model	National, financial, and macroeconomic indicators decide about the governments' choice of sovereign debt. Countries with developed financial markets, could issue sovereign Sukuk than sovereign bonds, to diversify their monetary markets with new debt products.
13	Kumar & Sahu, (2017)	India region Shariah indices and stock market. The time period is from 2006 to 2015. (9.5 years)	VECM, Granger, VAR, Johansen's	Long run equilibrium relation between Dow Jones index and macroeconomic indicators. In short run, unidirectional causality between Dow Jones index and money supply. Indian Islamic capital market is inefficient due to Co-integration between stock return and macroeconomic indicators.
14	El-Khatib & Samet, (2020)	45 emerging countries stock market index	GARCH	It shows harshly struck and sharp decrease in stock market indices. It causes a growth in volatility levels on sovereign credit default.
15	Ashraf & Marashdeh, (2018)	Shariah Indices from Kuwait, Oman, Qatar, Bahrain, Saudi Arabia and the United Arab Emirates, 5 years 6 months	Breusch Godfrey LM test, KPSS test and GPH (Geweke Porter Hudak) test	It was noticed in this study that during the study period, the returns of all GCC Equity Shariah were not informationally economical however fractionally integrated.
16	Siddiqui &	Nifty 50 Shariah,	Correlation test,	It was seen from the analysis that the returns of

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6067 - Published by University of Turin - <u>https://www.ojs.unito.it/index.php/EJIF/index</u> EJIF content is licensed under a Creative Commons Attribution 4.0 International License 12

[	C1 11-1		Colota and in the i	Charles Indiana and Kathan (1911)
	Sheikh, (2016)	Nifty 500 Shariah,	Cointegration test,	Shariah Indices are better, and it is less risky as
	(2010)	Nifty 50 and Nifty	Granger Causality	compared to underlying indices.
		500, 9 years	test, GMM and	
10	Reddy & Fu	Shariah and	3SLS Bick adjusted	The study shows that Charles have diadians are batter
19	(2014)	Shariah and	Risk adjusted.	The study shows that Shariah based indices are better
	(2014)	conventional	return, Sharpe	investment options while both indexes have a
		stocks for	rating, Treynor	tendency towards act in the same way.
		companies listed	rating, Jensen alpha	
		on the	rating, and Multiple	
		Australian Stock	Reduction Test	
		Exchange (ASX),		
		13 years		
20	El	Islamic Indices and	Cointegration test	The study concluded that Shariah guidelines work
	Khamlichi,	conventional	and Variance ratio	other than in the case of MSCI and FTSE. Whereas the
	Sarkar,	benchmarks of	test	Islamic Indices and conventional indices of Dow Jones
	Arouri, &	Dow Jones (12	244	and S&P are found to be integrated and hence,
	Teulon,	years 2 months),	8038	provide opportunity of investment diversification.
	(2014)	Financial Times (3	25	
		years 4 months),		
///		Standard & Poor's		
		(4 years 2 months)	7/1911 11	
		and Morgan		
		Stanley (3 years 2		
		months)		
21	Akguc & Al	Standard and Poor	Multi variate	Shariah firms are more profitable than Non Shariah
	Rahahleh,	Compustat global	regression analysis	firms
	(2018)	database of GCC		
		region. The time		
		period is from	. 52	
	$\geq$	2000 to 2014	154	
			25	
			1 20	
24	Al-Khazali,	Nine Dow Jones	Correlation, Capital	The results indicate that traditional indexes
	Lean, &	indices, 17 years	Asset Pricing Model	overshadow Shariah index all markets with the
$N \setminus \mathcal{U}$	Samet,	maleco, 17 years	(CAPM) statistic	exception of European market. Whereas Islamic
1//	(2014)	//	and Davidson–	indices for world, European and US markets overlook
			Duclos (DD) tests	their traditional equivalents.
26	Tyagi &	BSE Tasis Shariah	Graphical method	The study revealed that Sensex and Tasis Shariah
20	Rizwan,	and Sensex, 1 year	oraphical method	behaved similarly for the period under study.
	(2012)	6 months		benaved similarly for the period under study.
27			t tost CADM4+0	It was determined that the median worlds of the
27	Natarajan & Dharani,	Nifty Shariah	t-test, CAPM to	It was determined that the median yields of the
	(2012)	index, Nifty index	estimate beta, and	sharia Compliant Stocks and benchmark index were
	(2012)	and BSE Sensex	correlation Matrix	comparable. The returns from the sharia index were
		index, 5 years		just like the returns attained from the common index.
				Thus, sharia Compliant investment is taken into
				account a viable and ethical investment opportunity.
28	Munusamy &	Nifty Shariah	T-test	No difference was found between average day-wise
	Natarajan,	index and Nifty		returns of the Nifty Shariah index and Nifty Index
	(2011)	index, 4 years		during the period of study. The effect of Ramazan
				was also observed in the Indian stock market. Also,
				seasonal variation exists in the Shariah Index.

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6067 - Published by University of Turin - <u>https://www.ojs.unito.it/index.php/EJIF/index</u> EJIF content is licensed under a Creative Commons Attribution 4.0 International License

29	Mansor &	Malayasian	T-test and	The results revealed that Shariah portfolio earns less
	Bhatti,	Shariah Mutual	Correlation analysis	returns in regard of traditional equivalents and riskier
	(2011)	Fund and		than conventional one.
		traditional		
		portfolios, 13		
		years 4 months		
30	Hassan &	Dow Jones Islamic	Sharpe, Treynor,	No distinction was found among Islamic and non-
	Girard,	Market Index	Jensen and Fama's	Islamic files. The Dow Jones Islamic lists surpass their
	(2011)	(DJIM) and their	selectivity, four-	ordinary partners from 1996 to 2000 and fail to meet
		non-Islamic	factor pricing	expectations them from 2001 to 2005. Generally,
		counterparts, 10	models and	comparable award to hazard and enhancement edges
		years	cointegration test	exists for each the Islamic and regular lists.
31	Sadeghi,	Daily stock prices,	Event Study	The results showed that the introduction of Shariah
	(2008)	bid ask to spread	Methodology	index had a positive impact on the performance of
		and volume of	5 22	financial performances of stock included in the study.
		trade for 188	5	
		publicly traded	2035	
		companies on	25	
		Shariah Index of		
// *	$\gg$	Bursa Malaysia, 7		
		years	7/7/1/ //	
A V R				
		610		SET EN
			SA	

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6067 - Published by University of Turin - <u>https://www.ojs.unito.it/index.php/EJIF/index</u> EJIF content is licensed under a Creative Commons Attribution 4.0 International License

# The Sharia Compliance Level of Islamic Banks in Asia and Its Implications on Financial Performance and Market Share

Dodik Siswantoro<sup>1</sup>, Adelia Nidyanti

<sup>1</sup>Department of Accounting, Faculty of Economics and Business, University of Indonesia, Depok (Indonesia)

Corresponding author: kidod25@yahoo.com

Abstract— This study aims at showing the influence of the Sharia compliance level of Islamic banks in Asia and how it affects their financial performance and market share. It analyzing whether the average level of the Sharia compliance of Islamic banks in Asia is adequate. It incorporates a descriptive quantitative research design. The sample comprises 62 Islamic banks located in Asia with the observation period ranging from 2016 to 2018. However, following the analysis, the results of the study could not show whether the Sharia compliance level of Islamic banks in Asia has an influence on the financial performance and market share. The limiting factors behind this could be attributed to the limited sample and the short observation period of this study. However, the results of this study can be considered by stakeholders when investing their funds. In this regard, the regulators in Indonesia are expected to make deeper regulations for Islamic banks. This study contributes to the Islamic economic literature because the number of studies that look at the level of Sharia compliance with one measurement in the Asian region and see the implications of the level of this compliance on financial performance and market share is still limited.

Keywords- Sharia; compliance; financial; performance; Islamic; bank

#### I. INTRODUCTION

Islamic banking operates side by side with conventional banks. It makes the competition between them very challenging [1]. The competition drives Islamic banks (IBs) to work harder and be more creative and innovative in order to gain adequate market shares [2]. The main challenge of Islamic banking and finance is to provide products and services that are diverse, innovative, and competitive with the conventional financial instruments while still being in accordance with Sharia [3]. In carrying out its operations, IBs base their rules on the Qur'an and Hadith. All Islamic banking activities must comply with the Islamic Sharia principles and regulations made by the Sharia supervisors [4]. In this regard, shareholders and Islamic scholars are entitled to the services provided by Islamic banks in accordance with Sharia law [5].

There are several studies which show that Sharia compliance is the most important factor for prospective customers in choosing an IB [6] [7] [8]. Sharia compliance also guarantees an IB's credibility and inspires trust in their shareholders and stakeholders. To achieve uniformity of Sharia compliance among financial institutions, the Islamic Financial Services Board (IFSB) and the Accounting and Auditing Organization of Islamic Financial Institution (AAOIFI) were formed. Their responsibilities include forming the accounting, auditing, and government standards while also implementing the ethics that are in accordance with the Islamic law for financial institutions. An IB is determined by its commitment in following Sharia in all aspects [9]. The compliance of IBs and financial institutions to AAOIFI and IFSB standards is very important to manage the Sharia compliance risk, mitigate operational risks, and ensure transparency in financial statements [10].

Islamic financial institutions are exposed to Islamic disobedience risk which is a unique risk for these institutions [8]. Several cases of IBs and financial institutions violating Sharia principles raise serious doubts about Sharia compliance [7]. Examples include the Dubai Islamic Bank (DIB) scandal, Sunrise Equities Inc. case, and Dana Gas sukuk scandal. IBs and conventional banks are very similar in how they operate, due to which the former often ends up violating the Sharia principle's goals [11] [12]. Many of the financing offered by IBs have similarities with debt instruments rather than profit and loss sharing (PLS) practices. These practices negatively affect public perception [7]. They make customers skeptical of an IB with regards to its Sharia compliance [13].

In this regard, a Sharia compliance analysis would be beneficial for prospective customers and bank customers [8]. Most shareholders and investors are very concerned about the funds they invest and whether it is in alignment with Sharia principles or not [14] [7]. Therefore, there is an obligation on

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6025 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index

banks to reflect Islamic values in all their activities [9]. Sharia compliance is also important to promote trust, as it will increase customer protection. Customers trust an IB with their investment because they believe that it truly adheres to Sharia principles that prohibit usury, speculation, and undue risk [7]. In this regard, IBs have claimed that they are complying with the Sharia principles [15] [10] [16].

However, the fact remains that the level of Sharia compliance cannot be qualified as compliant or non-compliant; there must be a ranking system that regulates compliance with the law, including levels/categories such as high, satisfying, weak, and non-compliant. The different levels of Sharia compliance between banks in different countries cannot be compared because they use different measurements. There is no certain model that can describe the Sharia compliance of each IB [13]. Therefore, it is important to use a uniform measurement model so that the Sharia compliance levels of IBs in Asia can be compared fairly. The Sharia compliance ranking system is very helpful in measuring the progress toward achieving the goals of the Islamic Financial System and in gaining the stakeholders' trust. In addition, high compliance can help and support AAOIFI and IFSB in improving improve their standards so that they can be applied by Islamic financial institutions globally [15] [8].

Market forces should obey Sharia principles to increase the efficiency and competitiveness of the Islamic banking industry [17]. A majority (75%) of respondents agree that high Sharia compliance guarantees more financial benefits in Bangladesh [10]. A study using five categories of financial performance, namely profitability, efficiency, risk, asset quality, and liquidity, to examine the financial performance of IBs and conventional banks in Pakistan was conducted. The result shows that IBs are relatively better in terms of profitability, efficiency, risk, and liquidity management, while conventional banks are superior in asset quality [18]. On the other hand, Sharia compliant companies underperform compared to non-Sharia compliant companies. The Sharia compliance also has an impact on a bank's market share. The higher the level of a customer's or prospective customer's trust in IBs, the more people will invest their funds in these banks [19].

The aim of the study is to also expand the scope of or improve upon the previous research. The relationship between Sharia compliance and an IB's performance in Asia is an interesting issue. No previous paper discussed this topic and case before. The structure of the paper starts from the introduction and then proceeds to the literature review and hypothesis development. This is then followed by the research method and the analysis, the last of which is the conclusion.

#### II. LITERATURE REVIEW AND HYPOTHESIS DEVELOPMENT

#### A. *Corporate governance theory*

Corporate governance is a set of internal corporate arrangements that defines the relationship between managers and shareholders and is aimed at supporting financial stability, sustainable growth, and economic efficiency. The OECD principles can be adapted by Islamic financial institutions so that they can conduct business according to Sharia principles which are fair and honest with others. The scope of governance in Islamic principles is broader, as it not only prioritizes the interests of stakeholders but also the socio-economic interests. Based on the OECD principles, the only party entitled to decision making and internal business transaction participation is the Board of Directors (BOD) and senior management. However, in the case of in Islamic principles, anyone related to the organizational stakeholders is given the full right and responsibility to participate and convey their thoughts and ideas in reforming corporate governance [20].

The distribution of profit sharing is carried out fairly in accordance with the agreement of the two parties (bank and customer), which is determined at the time of the contract. Disclosure and transparency are emphasized in governance according to the Islamic principles. According to these principles, banks or financial institutions are not only responsible to the BOD, but also to Allah SWT. Therefore, it is important for IBs to pay attention to the Sharia compliance, as it is one of the good governance principles. By complying with Sharia principles, an IB has achieved certain key goals, namely carrying out activities that are free of usury and oriented to profit and social welfare.

#### B. Financial performance

Financial performance is a company's ability to manage and control its resources [21]. The measurement of financial performance is carried out by looking at the profitability level of the company, such as Return on Assets (ROA) [22] [23]. This ratio measures the company's ability to use its assets to make a profit. Financial ratios are taken from financial statements and made available to the public [24] [25]. A financial ratio analysis is useful for the management and interested parties to evaluate financial conditions expeditiously.

#### C. Market share

Market share is a proxy to see market power [26]. Companies that have a high market share tend to have more innovation. Companies with high market shares that continue to innovate also get higher valuations in the stock market than companies that do not [27].

#### D. Sharia compliance

In carrying out its operations, IBs are required to comply with laws, regulations, and Islamic principles because they are fundamental factors that distinguish IBs from conventional banks. Despite close supervision being carried out by various parties in this regard, various Sharia principles have been violated in the banking industry. Islamic scientists says that in conducting Sharia banking transactions, an IB does things that are contrary to Islamic regulations [28]. This is caused by ineffective governance, human resources, the role of the Sharia supervisory board, and unscrupulous people [29]. The implementation of Sharia compliance in Islamic banks must be reported and requires standards in order to produce a systematic financial report that can be understood by all parties who need information. AAOIFI has developed standards on

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6025 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index

auditing, governance, and accounting for Islamic financial institutions. In this regard, the Sharia compliance can be influenced by various factors, namely bank size [10], a Sharia internal audit and audit committee, profitability, board independence, and liquidity [30].

#### E. Hyphothesis development

Identify applicable sponsor/s here. If no sponsors, delete this text box. (sponsors)

Most IBs adhere to Islamic principles, such as those in Bahrain and Malaysia [16]. However, it is worth noting that not all banks with Islamic labels follow Sharia principles [7]. One reason or this is that the modern Islamic banking financial system is similar to conventional banking [11] [12]. Only a small portion of financing that is actually based on PLS and Islamic deposits is not interest free so it is as same as conventional deposits too. This certainly violates Sharia and hinders the achievement of Islamic principles. In addition, there are several cases of Sharia violations committed by IBs that occurred during the mid-1980s. One example of this is how Islamic banks in Turkey have deviated from the practice of Islamic banks [31]. In this context, the question on whether IBs in Asia comply well with Sharia remains to be answered. This has led to the formulation of the first hypothesis:

H1: The average level of the Sharia compliance of Islamic banks in Asia is compliant.

Increasing the competitiveness and efficiency of the Sharia banking industry can be done if the market force complies with the Sharia principles [17]. The greater the compliance level of IBs with Islamic law, the greater the customer's intention to get funds from these banks so that the latter can sustain their customer base [32]. The majority (75%) of respondents agree that better Sharia compliance guarantees more financial benefits in Bangladesh [10]. Islamic banks are relatively better at profitability, efficiency, risk, and liquidity management than conventional banks [18]. On the other hand, companies that are Sharia compliant [19]. It is on this basis the second hypothesis has been formed:

H2: The more compliant the Islamic bank is with Sharia principles, the better the Islamic bank's financial performance.

The more compliant the IBs are with Islamic principles and regulations, the greater the customer's intention to continue to obtain funds from them [32]. Customers will check for the Islamic compliance of an IB before deciding to choose it [7]. This will certainly increase the number of customers and the market share of the bank. In this regard, the third hypothesis has been formed:

H3: The more compliant the Islamic banks are with the Sharia principles; the market share of the Islamic banks will increase.

#### III. RESEARCH METHOD

This study follows a descriptive quantitative research design that uses secondary data, namely the annual financial statements published by Islamic bank. The sample for this study comprises all the IBs of Asia that are registered in ORBIS from 2016 to 2018. The annual reports for each IB are obtained through its official website. Out of the 113 IBs in the Asian region, only 62 fulfilled the sample selection criteria. Following this, the researchers tested the Sharia compliance with financial performance using a simple linear regression analysis with the following model.

$$\begin{split} &\text{ROAit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \text{Branchit} + \text{Ageit} + \epsilon \\ &\text{ROEit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \text{Branchit} + \text{Ageit} + \epsilon \\ &\text{OERit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \text{Branchit} + \text{Ageit} + \epsilon \\ &\text{NPFit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \text{Branchit} + \text{Ageit} + \epsilon \\ &\text{CTDit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \text{Branchit} + \text{Ageit} + \epsilon \\ &\text{CARit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \text{Branchit} + \text{Ageit} + \epsilon \\ &\text{FDRit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \text{Branchit} + \text{Ageit} + \epsilon \\ &\text{MSit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \text{Branchit} + \text{Ageit} + \epsilon \\ &\text{MSit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \text{Branchit} + \text{Ageit} + \epsilon \\ &\text{MSit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \text{Branchit} + \text{Ageit} + \epsilon \\ &\text{MSit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \text{Branchit} + \text{Ageit} + \epsilon \\ &\text{MSit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \text{Branchit} + \text{Ageit} + \epsilon \\ &\text{MSit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \text{Branchit} + \text{Ageit} + \epsilon \\ &\text{MSit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \text{Branchit} + \text{Ageit} + \epsilon \\ &\text{MSit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \text{Branchit} + \text{Ageit} + \epsilon \\ &\text{MSit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \text{Branchit} + \text{Ageit} + \epsilon \\ &\text{MSit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \text{Branchit} + \text{Ageit} + \epsilon \\ &\text{MSit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \text{Branchit} + \text{Ageit} + \epsilon \\ &\text{MSit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \text{Branchit} + \text{Ageit} + \epsilon \\ &\text{MSit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \text{Branchit} + \text{Ageit} + \epsilon \\ &\text{MSit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \text{Branchit} + \text{Ageit} + \epsilon \\ &\text{MSit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \text{Branchit} + \beta 0 \\ &\text{MSit} = \beta 0 + \beta 1 \text{ Complianceit} + \text{Sizeit} + \beta 0 \\ &\text{MSit} = \beta 0 + \beta 1 \text{ Complianceit} + \beta 0 \\ &\text{MSit} = \beta 0 + \beta 1 \text{ Complianceit} + \beta 0 \\ &\text{MSit} = \beta 0 + \beta 1 \text{ Complianceit} + \beta 0 \\ &\text{MSit} = \beta 0 + \beta 1 \text{ Complianceit} + \beta 0 \\ &\text{MSit} = \beta 0 \\ &\text{$$

#### Descriptions:

ROAit: Return on Assets of Islamic bank i in t ROEit: Return on Equity of Islamic bank i in t OERit: Operating Expense to Revenue of Islamic bank i in t NPFit: Non-Performing Financing of Islamic bank i in t CTDit: Cash to Deposit of Islamic bank i in t CARit: Capital Adequacy Ratio of Islamic bank i in t FDRit: Financing to Deposit ratio of Islamic bank i in t MSit : Market Share of Islamic bank i in t Complianceit: Sharia compliance of Islamic bank i in t Sizeit : The assets value owned by Islamic bank i in t Branchit : Number of branches owned by Islamic bank i in t Ageit : Period years of bank operates

The level of Sharia compliance from IBs is measured by using the Sharia compliance index developed by Ashraf and Lahsasna [13]. There are 14 indicators that look at this. They are grouped under five categories: support from regulators, quality of Sharia supervision, business structure, composition of assets and deposit base, and capital adequacy standards. Each indicator is assessed according to a predetermined score. A bank can get a maximum risk-weighted score of 150 and a minimum risk-weighted score of -138. The formula used to get the percentage of each IB is

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6025 - Published by University of Turin https://www.ojs.unito.it/index.php/EJIF/index

% Sharia compliance= (Achieved score)/(Maximum achievable score)

After the IB compliance percentage value is obtained, the IB is classified as shown in Table 1.

Achieved degree of compliance	Rating	Interpretation
≥80%	SSS	High Sharia compliance
50%-80%	S+	Satisfactory Sharia compliance
0–50%	S-	Weak Sharia compliance
Nilai negatif	SN	Sharia non-compliant
Source: [13]	0	N

	~ .			
Table 1.	Sharia	comp	iance	levels

The dependent variables in this study are financial performance and market share. Financial performance is measured using several financial statement ratios, namely ROA, ROE, OER, NPF, CTD, FDR, and CAR. The market share variable is measured using the formula from Berger [26]:

Market share= (total assets of Islamic banks/ total assets of national Islamic banks) x 100%

The total IB national asset is obtained from the stability report issued annually by the IFSB. In the stability report, the total IB assets in the world per region and their presentations for each country are given. Based on this percentage, the total assets of the national Islamic banking in each country are calculated, while the total IB assets are obtained from each IB's annual financial statements.

#### IV. RESULTS

Table 2. Desc	Table 2. Descriptive statistics					
Variables	Average	SD	Max	Min		
ROA	0,0101	0,0294	0,1842	-0,1077		
ROE	0,0802	0,1347	0,3685	-0,9401		
OER	0,7223	0,5764	6,2130	-0,2039		
NPF	0,0579	0,1079	0,8910	0		
CTD	0,4709	3,9200	53,529	0,0015		
CAR	0,1847	0,2579	1,6307	-1,2508		
FDR	0,8410	0,5613	4,7373	0,0190		
MS	0,2228	0,2739	1,0000	0.0156		
SC	0,6982	0,0736	0,84	0,44		
	Descriptions:					
ROA (Ret	ROA (Return On Asset); ROE (Return On Equity); OER					
(Operating Expense to Revenue); NPF (Non Performing						
Finance); C	CAR (Capital i	to Adequacy	Ratio); CTD	(Cash to		
Danagit): E	EDD (Financia	na to Donosi	+ Patio). MS	(Markat		

Deposit); FDR (Financing to Deposit Ratio); MS (Market Share); SC (The value of Sharia compliance level ) Source: Authors' elaboration

The value of each variable in Table 2 shows the effectiveness of the investment management (ROA) and capital (ROE) conducted by an IB, the amount of profitability of IBs (OER), the quality of financing provided by IBs (NPF), the

amount of IB funds used for financing (CTD), IB capital adequacy (CAR), and the amount of financing provided for deposits received by the banks (FDR). The highest ROA value of 0.1842 is owned by the Bank of Punjab Taqwa Islamic Banking (Pakistan, 2018) while the lowest ROA value of 0.1077 is owned by Bank Mega Syariah (Indonesia, 2017). The highest ROE value of 0.3685 is owned by Al-Hilal Bank (Kazakhstan, 2018) while the lowest ROE value of -0.9401 is owned by Bank Panin Syariah (Indonesia, 2017). Meanwhile, the highest OER value of 6.2130 is owned by ICB Bank (Bangladesh, 2017) while the lowest OER value of -0.2039 is owned by ICB Bank (Bangladesh, 2018). The highest NPF value of 0.8910 is owned by the Muamalat Malaysia Berhad Bank (Malaysia, 2017) while the lowest NPF value of 0 is owned by the ABC Islamic Bank (Bahrain, 2016). The highest CTD value of 53,529 is owned by Maybank Syariah (Indonesia, 2018) while the lowest CTD value of -0.0015 is owned by the Alliance Bank (Malaysia, 2016).

The highest CAR value of 1.6307 is owned by Maybank Syariah (Indonesia, 2018) while the lowest CAR value of -1.2508 is owned by the ICB Islamic Bank (Bangladesh, 2018). The highest FDR value of 4.7373 is owned by the Khaleeji Bank (Bahrain, 2017) while the lowest FDR value of 0.0190 is owned by Bank Muamalat Malaysia Berhad (Malaysia, 2018). The highest MS value of 1,0000 is owned by Al-Hilal Islamic Bank (Kazakhstan), Islamic Bank of Thailand (Thailand), and the Maldives Islamic Bank (Maldives) while the lowest MS value of 0.0156 is owned by Maybank Syariah (Indonesia, 2018). The Islamic bank that got the highest value of Sharia compliance (0.84) is the Meezan Bank (Pakistan) while the one who got the lowest Sharia compliance value (0.44) is the Gulf International Bank (Bahrain). The average value obtained was 0.6982.

In determining the level of Sharia compliance, a scoring is conducted on the annual financial statements of IBs in Asia. Based on the scoring result, there are three IBs in the satisfactorily compliant category, 61 IBs in the high-compliant category, and one IB in the less-compliant category with none in the non-compliant category. The scoring results show the average Sharia compliance level of Islamic banks in Asia is compliant. This proves H1 that IBs in Asia comply with Sharia principles. There are only three IBs in the high-compliant category, namely the Aljazeera Bank, Jordan Islamic Bank for Finance and Investment, and Meezan Bank. These banks have very high Islamic compliance because their amount of distribution of financing in the form of mudarabah and musyarakah is very high compared to other Islamic banks. However, what should be underlined here is that this assessment is based on the annual financial statements of the IBs. The less compliant Islamic Bank is the Gulf International Bank (GIB) because the Sharia supervisory board reports the shariha compliance opinion to the CEO so its independence is less than the Islamic Bank reporting to the BOD or directors. Moreover, GIB also does not have its own Sharia supervisory board report although it has its own risk management report. In this regard, it is more focused on risk management compared to Sharia compliance. Moreover, it does not make a breakdown of the income earned (murabahah, musharaka, mudarabah or

DOI: 10.13135/2421-2172/6025 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index

European Journal of Islamic Finance - ISSN: 2421-2172

ijarah). Instead, it has a pool of funds mixed with conventional banks because it is a business unit of conventional banks.

Some other interesting things were also discovered by researchers in scoring the IBs' annual financial statements. First, the separation of the Islamic banking law from conventional banking law is carried out by three countries, namely Bahrain, Iraq, and Kuwait. In contrast, other countries have one single banking law that regulates Islamic Bank with conventional banks. However, Indonesia and Malaysiadespite having a single law governing Islamic banks with conventional banks—have separate guidelines and regulations for IBs. Second, there are some Sharia supervisory boards that are in charge of conducting Sharia audits, while there are those who only serve as advisors of Sharia banks. Based on the scoring, most countries require Islamic audits. Countries that assigned Sharia supervisory boards as advisors to Islamic banks include Indonesia, Sri Lanka, Iraq, and Brunei Darussalam. Third, a majority of the sample reports the opinion results to the BOD, compared to the CEO and directors. The IBs, in general, also report the results of the assessment (opinion) of the Sharia compliance in the financial statements very well, namely by notifying what is done. Following this, further advice or recommendations are given on the basis of these findings, in addition to explaining that the IB is compliant with Sharia. Fourth, a majority of the sample have separate legal entities within conventional banks. In addition, certain IBs have very little Profit Equalization Reserve, namely those in Bahrain (3 Islamic banks), Kuwait (1), Oman (2), and Pakistan (1). Fifth, the majority of IBs in the sample have separate but not independent charity funds (charity funds are obtained from non-halal income or obtained from zakat). Only a few IBs have foundations or special programs for their charity funds such as Al-Bilad Bank (Saudi Arabia). Sixth, the sample mostly has 6 to 10 types of financing products. The majority of third-party fund structures owned by IBs are obtained from mudarabah and musyarakah rather than murabahah. Finally, the sample in this study has a very good capital adequacy in accordance with IFSB rules which is above 12%. Some even have CARs above 20%.

Based on the scoring results, it can be concluded that IBs in Asia have a level of Sharia compliance at 50–80% which is 69.82%. This is in line with Vinnicombe which states that political, economic, social, and geographical factors have an impact on the level of compliance with AAOIFI standards [9]. In most countries, IBs are placed under the supervision of a central bank and treated the same as conventional banks.

Table 3. Rest	Table 3. Results of H2					
	Moo	del 1	Model 2			
Variable	Coef.	Prob.	Coef.	Prob.		
Constants	-0,0157	0,3005	-0,2303	0,0410		
SC	0,0168	0,4470	0,1052	0,2265		
Size	0,0008	0,7420	0,0115	0,0190		
Branch	-0,0036	0,7440	0,0007	0,1020		
Age	-0,0005	0,5440	0,0009	0,2310		
R2	0.0127		0,1930			
Overall	0,0127		0,1930			
F-	0,8664		0,0007			

statistics						
	Model 3 Model 4					
Constants	24,535	0,0000	0,0966	0,308		
SC	-10,301	0,0435**	0,0356	0,381		
Size	-0,0487	0,0200**	-0,0036	0,380		
Branch	0,0001	0,6230	0,0001	0,655		
Age	-0,0073	0,0320**	0,0001	0,988		
R2	0,1796		0,0093			
Overall						
F-	0,0013		0,9209			
statistics						
		Descriptions:				
SC: Sharia compliance level of Islamic bank; Size:						
Logarithm of total assets; Branch: Total branches of Islamic						
banks; Age: The period of bank operate						
Coef: Variable coefficient						
***: sig	nificant level	1%; **: sign	nificant level	5%; *:		
		ficant level 1				
Source: Auth	ors' elaborat	rion				

Source: Authors' elaboration

The result of testing model 1 shows that the independent and control variables have no significant effect on the dependent variable. The R2 overall value in model 1 is 0.0127. This shows that high or low ROA value does not have a relationship with Sharia compliance. IBs, especially in Indonesia, have a small ROA compared to conventional banks because the former bear a greater cost than the income earned. The model 2 test results show that the independent variable and control variables do not have a significant effect on the dependent variable. The R2 overall value in model 2 is 0.1930, i.e., the level of ROE does not affect the level of Sharia compliance. The ROE shows how the more efficient Islamic bank managers manage their capital to conduct financing. The results from testing model 3 show that the independent variable has a significant influence on SC at a significance level of 5%. The control variables, namely size and age, significantly influence the dependent variable at a significance level of 5%. The overall R2 value in model 4 is 0.1796. That is, the more compliant the IB, the lower the OER ratio. This is contradictory to Lassoed's findings that IBs have to bear higher costs when providing quality services [33]. This is because quality services are supported by diverse facilities and friendly services, which, in turn, require huge expenses to sustain it. Qualified human resources are needed due to which, it is necessary to recruit and provide adequate training to the employees.

|--|

Variable	Model 5		Model 6	
Variable	Coef.	Prob.	Coef.	Prob.
Constants	0,7534	0,8250	-0,1129	0,6080
SC	-62,865	0,0685	0,5109	0,0031**
Size	0,2426	0,1010	0,0011	0,9070
Branch	-0,0003	0,8310	-0,0006	0,4840
Age	-0,0158	0,5070	-0,0038	0,0120**

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6025 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index

R2	0,0452		0,1181	
Overall	-,		•,•-	
F-	0,3405		0,0188	
statistics	,		,	
	Mo	del 7	Mo	del 8
Constants	0,5008	0,2950	-0,2036	0,3810
SC	0,9303	0,0585*	-2,942	0,1540
Size	-0,0190	0,3580	0,0346	0,0010***
Branch	-0,0006	0,0010**	0,0001	0,2810
Age	-0,0064	0,0520*	-0,0008	0,6140
R2	0,1285	1	0,1452	
Overall				
F-	0,0120	5	0,0058	
statistics	$// \langle$			
		Descriptions:	PIN	
SC: Sh	aria complia	nce level of I	Islamic banl	k; Size:
Logarithm	of total asset	s; Branch: To	otal branche	s of Islamic
b	anks; Age: T	he period of	bank operat	te
	Coef:	Variable coe	efficient	
***: sig	nificant leve	l 1%; **: sig	nificant leve	el 5%; *:
		ficant level 1		
Source: Auth				

Source: Authors' elaboration

The model 5 test result shows that the independent variable does not have a significant effect on the dependent variable. The R2 overall value in model 5 has a value of 0.0452. In other words, Sharia compliance does not affect CTD. A large amount of cash violates the Maqasid Syariah principles because the cash is idle money that should be used to distribute financing to customers. The more cash in the bank, the more it implies that the money is buried in the bank. Fewer IBs give financing to customers. The model 6 test result shows that the independent and control variables significantly influence the SC variable at a significance level of 5%. The overall R2 value in model 6 is 0.1181. This shows that Sharia compliance has a relationship with the CAR. The more compliant the IBs, the higher the CAR value. This ratio shows the adequacy of the capital owned by IBs in bearing the risk of losses that may occur due to mudarabah, musharakah or other transactions.

The result from testing model 7 shows that the FDR variable significantly influences the dependent variable at a significance level of 10%. There are two control variables that have a significant influence on Sharia compliance, namely the number of branches and age of IBs at the significance level of 5% and 10%, respectively. On the other hand, the overall R2 value in model 7 is 0.1285. From this, it is quite evident that the Sharia compliance affects the FDR. The more compliant the IBs, the higher the FDR. IBs which have a lot of financing are getting better because they achieves their goal of providing financing to their customers. Significant control variables operate behind an IB's size and number of branches in affecting the level of Sharia compliance [10]. The model 8 test result shows that the MS variables and the majority of control variables do not significantly influence the dependent variable. There is one control variable that influences Sharia compliance. namely an IB's size with a significance level of 1%. The R2

overall value on model 8 is 0.1452 which implies that the Sharia compliance does not affect market share. All in all, from the market share, one can see the development of IBs.

Based on the regression results above, it can be concluded that H2 was rejected. This finding is in accordance with the fact that Sharia compliant companies have lower performance compared to non-Sharia compliant companies [19]. Although Sharia compliance is a governance principle, financial performance depends on not only Sharia but also the internal control systems, risk management systems, and services provided by employees. Considering all this, the results of the regression test failed to prove that better Sharia compliance guarantees more financial benefits [10]. However, the study supports Bizri (2014) in the fact that prospective customers prefer IBs based on security, trustworthiness, quality of service rendering, and price of financing compared to Sharia compliance [7].

#### V. CONCLUSION

The study aimed to determine the Sharia compliance level of IBs in Asia and its influence on their financial performance and market share. The level of Sharia compliance of IBs in Asia on average can be classified under the high-compliant category. This is due to the people in the sample countries upholding Islamic principles in various aspects of life including banking. This opinion is consistent with previous studies that investors in the sample countries prioritize Sharia bank compliance before investing their funds [34] [35]. Based on the regression results, Sharia compliance affects the OER, CAR, and FDR of IBs. However, the ROA, ROE, NPF, and CTD are proven not to be affected by Sharia compliance. The establishment of IBs in countries that do not have a majority Muslim population is hindered by regulations. Moreover, they also do not have the support of the central bank due to which, the level of compliance remains low. In this regard, the IBs in Indonesia must emulate the IBs in Asia so that the level of Sharia compliance is maintained and further enhanced. This is because an adequate level of Sharia compliance is the goal of IBs. The results of this study can be considered by stakeholders before investing their funds. They can find out whether IBs actually comply with Islamic regulations because they are the basis that distinguishes IBs from conventional banks. The regulators in Indonesia are expected to be able to make deeper regulations for IBs, as has been done in several countries, such as Bahrain, Iraq, and Kuwait. These three countries have Islamic banking laws that are separate from conventional banks.

However, there are certain limitations to the study due to which, its scope can be extended further in future studies. For instance, the sample is limited and the observation period is quite short. Moreover, the measurement of Sharia compliance is not the best one. Future research could keep these limiting factors in mind while deciding on research designs and components that would be more efficient. It is also worth noting that the results of the study are subjective as well, as the assessment is based on the annual Islamic financial statements. In future, it is recommended that case studies be conducted for

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6025 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index

each IB to better ensure a sufficient level of Sharia compliance on the part of IBs.

#### REFERENCES

[1] Waemustafa W., 2013, The emergence of Islamic banking: Development, trends, and challenges, IOSR J. Bus. Manag., 7, 2.

[2] Yumanita D. and Ascarya, 2005, Bank Syariah: Gambaran Umum, Seri Kebanksentralan, 14.

[3] Venardos A. M., 2005, Islamic banking and finance in South-East Asia: Its development and future.

[4] Archer S., Karim R. A. A., and Sundararajan V., 2010, Supervisory, regulatory, and capital adequacy implications of profit-sharing investment accounts in Islamic finance, J. Islam. Account. Bus. Res., 1, 1.

[5] Alnasser S. A. S. and Muhammed J., 2012, Introduction to corporate governance from Islamic perspective, Humanomics, 28, 3.

[6] Metawa S. A. and Almossawi M., 1998, Banking behavior of Islamic bank customers: Perspectives and implications, Int. J. Bank Mark., 16, 7.

[7] Bizri R. M., 2014, A study of Islamic banks in the non-GCC MENA region: Evidence from Lebanon, Int. J. Bank. Mark., 2, 2.

[8] Hanif M., 2018, Sharī ah-compliance ratings of the Islamic financial services industry: A quantitative approach, ISRA Int. J. Islam. Financ., 10, 2.

[9] Vinnicombe T., 2010, AAOIFI reporting standards: Measuring compliance, Adv. Account., 26, 1.

[10] Ullah M. H., Khanam R., and Tasnim T., 2018, Comparative compliance status of AAOIFI and IFSB standards: An empirical evidence from Islami Bank Bangladesh Limited, J. Islam. Account. Bus. Res., 9, 4.

[11] Kameel M. M. A. and Larbani M., 2009, Ownership effects of fractional reserve banking: An Islamic perspective, Humanomics, 25, 2.

[12] Chong B. S. and Liu M. H., 2009 Islamic banking: Interest-free or interest-based? Pacific Basin Financ. J., 17, 1.

[13] Ashraf M. A. and Lahsasna A., 2017, Proposal for a new Sharīʿah risk rating approach for Islamic banks ISRA, Int. J. Islam. Financ., 9, 1.

[14] Chapra M. U. and Ahmed H., 2002, Corporate governance in financial institution.

[15] Sarea A. and Hanefah M., 2013, The need of accounting standards for Islamic financial institutions: Evidence from AAOIFI, J. Islam. Account. Bus. Res.

[16] Eddy Y. and Ezry F., 2010, Are Islamic Banks in Malaysia Really "Islamic"? Munich Pers. RePEc Arch., 20901.

[17] Ariff M. and Rosly S. A., 2011, Islamic banking in Malaysia: Unchartered waters, Asian Econ. Policy Rev. 6, 2.

[18] Khan H. and Bashar O. K. M. R., 2008, Islamic finance: Growth and prospects in Singapore, High. Educ., 1.

[19] Farooq O. and Alahkam A., 2016, Performance of Sharia-compliant firms and non-Sharia-compliant firms in the MENA region: Which is better? J. Islam. Account. Bus. Res., 7, 4.

[20] Abu-Tapanjeh A. M., 2009, Corporate governance from the Islamic perspective: A comparative analysis with OECD principles, Crit. Perspect. Account., 20, 5.

[21] Ikatan Akuntan Indonesia, 2016, Standar Akuntansi Keuangan, IAI Global.

[22] Mudjijah S., Khalid J., and Astuti, D. A. S., 2019, Pengaruh Kinerja Keuangan dan Struktur Modal Terhadap Nilai Perusahaan yang Dimoderasi Variabel Ukuran Perusahaan, J. Akunt. dan Keuang., 8, 1.

[23] Sartono A., 2015, Manajemen Keuangan Teori dan Aplikasi, Edisi 4, Yogyakarta: BPFE.

[24] Galant A. and Cadez S., 2017, Corporate social responsibility and financial performance relationship: A review of measurement approaches, Econ. Res. Istraz., 30, 1.

[25] Brigham E. F. and Houston J. F., 2010, Dasar-Dasar Manajemen Keuangan, Edisi 11, Salemba Empat.

[26] Berger A. N., 1995, The relationship between capital and earnings in banking, J. Money, Credit Bank., 27, 2.

[27] Blundell R., Griffith R., and Van Reenen J., 1999, Market share, market value and innovation in a panel of British manufacturing firms, Rev. Econ. Stud., 66, 3.

[28] Sjahdeini S. R., 2014, Perbankan Syariah: Produkproduk dan Aspek-aspek Hukumnya, Perbankan Syariah.

[29] Mardian S., 2019, Tingkat Kepatuhan Syariah di Lembaga Keuangan Syariah, J. Akunt. DAN Keuang. Islam, 3, 1.

[30] Segarawasesa F. S., 2018, Determinan Tingkat Kepatuhan Syariah pada Bank Syariah di Indonesia.

[31] Yanīkkaya H. and Pabuçcu Y. U., 2017, Causes and solutions for the stagnation of Islamic banking in Turkey, ISRA Int. J. Islam. Financ., 9, 1.

[32] Anshori A. G., 2008, Sejarah Perkembangan Hukum Perbankan Syariah di Indonesia dan Implikasinya bagi Praktik Perbankan Nasional, La\_Riba, 2, 2.

[33] Lassoued M., 2018, Comparative study on credit risk in Islamic banking institutions: The case of Malaysia, Q. Rev. Econ. Finance, 70.

[34] Awan H. M. and Bukhari K. S., 2011, Customer's criteria for selecting an Islamic bank: Evidence from Pakistan, J. Islam. Mark., 2, 1.

[35] Komijani A. and Taghizadeh-Hesary F., 2019, An overview of Islamic banking and finance in Asia, in Routledge handbook of banking and finance in Asia.

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6025 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index

# Mapping Islamic finance and new technologies: research and managerial perspectives

Davide Calandra<sup>1</sup>, Roberto Marseglia<sup>2</sup>, Vincenzo Vaccaro<sup>3</sup>, Federico Chmet<sup>1</sup>

- 1. University of Turin, Department of Management, Italy
- 2. University of Pavia, Department of Economics and Management, Pavia, Italy
- 3. Georgetown University, Mccourt school of public policy, USA

#### Corresponding author: davide.calandra@unito.it

Abstract- New technologies promote radical changes in how banking services are delivered. The field of Islamic finance is not exempt from radical changes and is gaining increasing attention from academics and practitioners. Although several scientific studies have been conducted on the impact of technologies on specific Islamic finance instruments, no research has examined bibliometric variables in this area. This study aims to cover this gap by conducting a bibliometric and open coding analysis on 170 sources of published studies on Islamic finance and technology. Our study detects the most influential authors, journals and countries of publication that currently prioritise research in this field. In addition, the thematic analysis reveals that among the niche themes are applications of technology to Takaful. Among the motor themes is the opportunity for technology to facilitate decision-making in Islamic banks. Finally, in terms of originality, this study highlights the field's current state by combining methodological approaches and providing valuable insights for future research. Moreover, it is also a starting point for practitioners to fully understand the characteristics and potential of technology in Islamic finance. Finally, the article provides researchers with a research agenda to guide future research ideas.

Keywords: Islamic finance; Blockhain; Cloud; Artificial intelligence; FinTech; business implications; Future research agenda

#### I. INTRODUCTION

New technologies are gradually changing various businesses and activities. [1], [2]. Islamic finance is also looking with interest at critical global trends. In fact, among the most-watched developments today are cloud, blockchain and artificial intelligence [3]–[5]. According to several international authors, these technologies could change the business models of Islamic banks and promote new virtuous economies to make them more attractive from the economic point of view than traditional finance banks. [6]–[8]. Therefore, this literature sees a decisive advantage for Islamic finance from applying new technologies.

Alongside the technologies applied to the business of Islamic financial institutions, numerous publications are dealing with Islamic Fintech and its compatibility with Sharia'h [9].

Today, the main areas of study in the business sector are undoubtedly Sukuk with process robotisation applications [10], Takaful with automatic management applications and fraud reduction thanks to the blockchain [11] and FinTech [12].

Despite the widespread interest shown by the growth in publications, as defined by Delle Foglie et al. (2021) and Uluyol (2021) [13], [14], more research will be needed in this initial area to systematise publications and identify any increased scope for global competition. This can only be the case if the level of knowledge on the subject is precise. Therefore, precisely on this point, as suggested by Zupic and Cater (2015) and Secinaro et al. (2021) [5], [14], timely analyses of knowledge flows with bibliometric and open code analyses could provide researchers and practitioners with a state-of-the-art on what has been published so far.

All these premises give us the idea of an embryonic stream of knowledge with few authors and citations looking for its own space within the more significant strand of research on Islamic finance.

Therefore, this paper proposes a broad investigation, including studying primary bibliometric data on peer-reviewed sources in the international Scopus database [16], [17].

The main findings denote a multidisciplinary view of the research field studied. The literature includes an interesting discussion on possible new technological tools for Islamic

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6257 - Published by University of Turin <a href="https://www.ojs.unito.it/index.php/EJIF/index">https://www.ojs.unito.it/index.php/EJIF/index</a>

finance, the development of new technical processes to increase the operational efficiency of the Islamic financial system, and the presence of alternative payment and investment methods based on Fintech. Furthermore, we note that the strand of research looking at technological solutions consistently looks at cost reduction as a sign of operational efficiency vis-à-vis traditional type banks. Our analysis is innovative for the topic of interest. It applies a hybrid methodology that includes, without distinction, new technologies and not only a specific tool such as Sukuk or Takaful or FinTech as a financial innovation [12], [18].

Therefore, our article aims to map, discuss, and critique the research debate on these issues by answering the following questions: (1) What are the main characteristics of these research streams considering authors, citations, and geographical interest? (2) What are the most frequent topics in this literature? (3) What are the possible implications for future research in this field?

The rest of the article is structured as follows. Section 2 sets out the methodological flow. Section 4 shows the results of our research. Section 5 provides an in-depth interpretation of the data, commenting on and critiquing the main results. Section 6 concludes this article by considering the current implications and limitations and building on these by suggesting future research paths.

#### П. METHODOLOGY

This paper uses a hybrid method joining the bibliometric analysis and the available coding analysis [5]. As the first research step, we create the review protocol. According to Massaro et al. (2016) [17], authors should include clear and direct data replication processes. Additionally, as Tranfield et al. (2003) [19], in wide-ranging literature reviews, research teams should put transparency measures to allow for high reliability and replicability of the research, even at different times. We implement a six-step research protocol from other published papers with known elements, motivation in conducting this research paper, the topic under discussion, the research limitations, tools used, and the research framework followed. The following Table I shows the research protocol followed by the researchers. Finally, the next sub-paragraphs describe the dataset creation phase, all the tools used to implement the analysis, and the open coding analysis protocol.

#### TABLE J. **RESEARCH PROTOCOL**

Review protocol elements	Author's considerations
What is already known? (Step 1)	Technologies promote a change in Islamic financial institutions' working and business models [20]. More and more researchers are fostering research in this area because of the opportunity for cost reductions and operational modifications [5], [21].
Motivation (Step 2)	A rising research stream analyses how new technologies will impact Islamic finance. There is currently no unified mapping in the literature despite numerous research studies. This creates fragmentation and difficulty in interpretation for researchers who must verify multiple sources of knowledge.

European Journal of Islamic Finance - ISSN: 2421-2172	
DOI: 10.13135/2421-2172/6257 - Published by University of Turin	
https://www.ojs.unito.it/index.php/EJIF/index	
FIJE content is licensed under a Creative Commons Attribution 4.0 Is	nto

**Review protocol** Author's considerations elements Fragmentation promotes possible bibliometric analyses in this emerging Research topic (Step 3) research area [15]. Following Massaro et al. (2016) and Secinaro et al. (2021) [5], [16], we decided not to limit the search to a single scientific journal. Moreover, as Dumay and Cai (2014) Research limitations [20], we included peer-reviewed journal (Step 4) articles, book chapters, and conference proceedings in the analysis. Finally, book chapters and white papers were excluded as nonpeer-reviewed sources. Scopus database, Bibliometrix R package Research tools (Step 5) and Atlas. Ti Cloud. Years, documents' information, sources, authors, keywords, citations, countries, Research framework theories analysed, methodologies used and (Step 6) technologies under discussion. Source: Author's elaboration

#### A. Dataset creation

The critical sources under analysis were collected, starting with studying the keywords. According to Chen and Xiao (2016) [23], keywords selection can be made using different methods. The first is to use wide-ranging words that encompass a large, high-level search domain. The second is to analyze smaller and perhaps rising search domains. Therefore, this study utilizes the second mode of research by analysing the micro-domain of new technologies and their use in Islamic finance.

Additionally, starting from the analysis conducted by Massaro et al. (2020), de Mem Machado (2021) [24] and Rosa et al. (2020) [25], we have selected which keywords: "Islamic finance" AND "Tech\*" OR "Blockchain" OR "Artificial intelligence" OR "Cloud" OR "Smart".

As shown in Table 1, we used the multidisciplinary and international Scopus database for data collection [26]. Indeed, several studies published in leading international journals use this database as it also includes data from other registries such as Web of Science [5], [27]-[29]. Despite the certainty of the previous studies, we decided, in any case, to compare the results of the two databases to avoid the loss of relevant data. In the outcome, we can confirm that the coverage of articles was more excellent on Scopus.

The dataset's creation includes only documents published until 2021 [30]. The application of the search requirements returned 179 documents. However, the research team could not access PDFs of 9 documents, so the final sample analysed was 170 documents. Figure 1 below, using the methodological paper by Liberati et al. (2009) [31], shows the research steps used by the authors.



Figure 1. Prisma Workflow, Source: Author's elaboration using Liberati et al. (2009) [31]

#### B. Tools of analysis

The research used several analysis tools. First, the quantitative and qualitative data analysis was done using the Bibliometrix and Biblioshiny package through the R Studio application [32]. Similarly, as evident from the research gap, few studies deal with providing a map of the theories and, at the same time, systematising the technologies that to date are applied to Islamic finance. Therefore, our research uses Atlas. Ti Cloud software will provide valuable insights regarding technologies applied by researchers in this research area [33].

III. RESULTS

This section aims to illustrate the bibliometric and open code analysis results. In the following sub-sections, the reader will see the research variables such as types of documents, years of publication and scientific production, scientific reference sources, growth of sources, number of articles per author, author productivity and the area related to keyword analysis. Finally, the section will end with a specific analysis of the theories used by researchers and the study of methods and technologies.

#### A. Main information

Table II shows the information of the 170 papers selected between 1997 and 2021. The research topic appeared in 112 scientific sources with an average annual publication rate of 3.7 articles. However, as shown in Figure 2, the figure could be misleading as most published papers are after 2013.

Further on, the most published type of document is the scientific article in a peer-reviewed journal (119), followed by books (15) and peer-reviewed book chapters (14). Finally, the analysis of authors and collaborations is also interesting. What emerges from these first data is a research field with a still low level of collaboration and single authors or an average number

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6257 - Published by University of Turin https://www.ojs.unito.it/index.php/EJIF/index

EJIF content is licensed under a Creative Commons Attribution 4.0 International License

of authors equal to 2.28, a symbol of a growing area that is still little explored by researchers worldwide.

#### TABLE II. MAIN INFORMATION

Data	Results
Timespan	1997-2021 (September)
Documents	170
Sources	112
Average years from publication	3.7
Average citations per document	7.3
Average citations per year per document	1.224
Number of references	10.367
Document types	
Articles	119
Books	15
Book chapters	14
Conference papers	13
Conference review	1
Editorial	1
Review	7
Document contents	
Keywords Plus	183
Author's keywords	515
Authors	
Authors	387
Author appearances	435
Authors of single- authored documents	38
Authors of multi- authored documents	349
Authors collaboration	7/9//
Single-authored documents	44
Documents per Author	0.439
Authors per document	2.28
Co-Authors per documents	2.56
Collaboration Index	2.77



Figure 2. Annual scientific production. Source: Author's elaboration using Biblioshiny

#### B. Sources' analysis

Table III identifies the top 10 sources of publication. Interestingly, the unitary distribution of the sources published both in reference and monothematic journals on Islamic finance and journals with a wide range of topics. For example, the Journal of Islamic Accounting and Business Research is one of the most critical publications (21), followed by the International Journal of Islamic and Middle Eastern Finance and Management (9). On the other hand, among the journals with a broader subject matter is Qualitative Research in Financial Markets (6), aiming to publish contemporary research on corporate and international finance issues.

TABLE III. RELEVANT SOURCES

Sources	Number of publications
Journal of Islamic Accounting and Business Research	21
International Journal of Islamic and Middle Eastern Finance and Management	9
Qualitative Research in Financial Markets	6
Islamic Fintech	4
Al-Shajarah	3
Borsa Istanbul Review	3
Journal of Islamic Marketing	3
Research in International Business and Finance	3
Arab Law Quarterly	2
Contemporary Studies In Economic And Financial Analysis	2

Source: Author's elaboration using Biblioshiny

Figure 3 identifies the cumulative frequency distribution of articles published by Journals. Therefore, the names of the journals that have published research contributions on this topic to date are shown. Interestingly, the publication flow in the most relevant scientific journals began in 2009, a year that is also relevant for technologies such as blockchain [5]. These results align with the growth shown in Figure 1 and the interest in economic research approaches.



Figure 3. Source Growth. Source: Author's elaboration using Biblioshiny

#### C. Authors' analysis

Table IV shows the top 10 most relevant authors. The authors with the most publications are Rabbani MR. from the University of Bahrain, a very active centre for publications on technology, and Professor Hassan M.K. from the University of New Orleans and Editor-in-Chief and editorial member of several journals on Islamic finance. His research has helped broaden the research horizons of Islamic finance by providing interesting and up-to-date ideas and research objectives for other fields. Finally, Khan S. and Saiti Buerhan follow with 6 and 5 publications.

Finally, what emerges from the analysis is a research topic trying to find its place within the macro-discussion on Islamic finance. The first authors per publication indicated here are experts in the field who, with their pilot research projects, are promoting an open and unconditional discussion on the topic.

TABLE IV. MOST RELEVANT AUTHORS

Number of publications
8
8
6
5
4
3
3
3
2
2

Source: Author's elaboration using Biblioshiny

Table V shows the analysis of the most promoting research on Islamic finance and new technologies. The International Islamic University Malaysia is the leading institution in the number of articles published (17). Next, we discover how the significant universities in the world dealing with research in this field reside in the Middle East. A clear vision can explain these results that states like Bahrain promote funding for new technologies [34].

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6257 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index

TABLE V. MOST RELEVANT AFFILIATION

Articles
17
8
6
6
6
5
4
4
4
3

#### D. Citations' analysis

The analysis of citations can yield interesting drivers of observation (Table VI). We will first use concerns by studying the most cited sources at the local level, i.e., within the research area under study. This approach is beneficial in omitting references with greater scientific relevance [28], [35]. In particular, one of the most cited sources is a literature review by Rabbani, Khan and Thalassinos [32], which aims to identify the level of knowledge of Islamic finance within Fintech, cryptocurrency, and blockchain. Furthermore, the study classifies Islamic financial instruments that are Shari'ah compliant. Another interesting study is Khan et al. (2020) [10], which introduces the concept of Sukuk tokens by showing that blockchain can decrease managing costs'.

TABLE VI. Mos	T RELEVANT LOCAL CITATIONS
---------------	----------------------------

Author	Total citations
Gheeraert L. (2015), Econ Model [36]	54
Rudnyckyj D. (2014), Am Ethnol [37]	31
Syed MH. (2020), Int J Econ Bus Adm [38]	17
Rabbani MR. (2020), Int J Econ Bus Adm [18]	16
Diaw A. (2015), J Islamic Account Bus Res [39]	14
Hassan MK. (2020), J Econ Coop Dev	9
Rabbani MR. (2020), Int J Sci Technol	4
Khan N. (2020), Global Financ J [10]	4
Salim BF. (2016), Int J Econ Financ Issues	
Khediri KB. (2015), Res Int Bus Financ [8]	2

Source: Author's elaboration using Biblioshiny

#### E. Keywords' analysis

Analyzing keywords allows us to study the concepts most frequently addressed by researchers [23]. As shown in figure 4,

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6257 - Published by University of Turin https://www.ojs.unito.it/index.php/EJIF/index

EJIF content is licensed under a Creative Commons Attribution 4.0 International License

we do not find uniform examples of technologies except for blockchain and Fintech. We also see that Sukuk is one of the most used tools applied. Finally, a geographic fact that will be confirmed later in the analysis is that Malaysia has the highest level of research in this area.



Figure 4. Trend topic. Source: Author's elaboration using Biblioshiny

Figure 5 below illustrates the thematic map of trending topics based on keywords plus analysed. These are words or phrases that often appear in the titles of article references but not in the primary title of the article itself. Such extractions are possible thanks to an algorithm developed by Clarivate Analytics that allows extending the scope of information by looking at the article's content and not only at the title, abstract and external keywords [40]–[42]. Applying the data reading to the Bibliometrix package makes it possible to obtain a graph by relevance and development. Inside it is possible to notice a distinction by topics:

- Niche themes;
- Motor themes:
- Emerging/declining themes
- Basic themes.

Starting with the first one above, we find that the topic of blockchain is applied to the takaful insurance industry among the niche themes. We see interest from researchers in this topic, especially for data management between different insurance locations and fraud prevention by policyholders [11]. Despite the interest and potential, this topic is not yet widely developed.

The second quadrant of the graph shows the motor themes. They represent the topics on which the scientific discussion is based today. We discover how technology can support the decision-making processes of Islamic financial institutions. Among the most curious applications is pioneering research that identifies technology as an opportunity to increase the economic sustainability of Islamic banks in case of banking disputes and controversies [43].

The third quadrant of the graph identifies emerging issues in which researchers may have future interests. These include climate change. Although widely developed in several businesses and traditional finance [44]–[46], its treatment is still far from widespread in Islamic finance. Finally, the last part of the graph shows the basic themes, including Islamic Fintech, which has had its scientific history with numerous contributions investigating its compatibility with Sharia'h [7], [9], [47], [48].



Figure 5. Thematic map. Source: Author's elaboration using Biblioshiny

#### F. Open coding analysis

The objective of this brief section will be to show the results of the open code analysis conducted by the researchers on the selected documents. The research uncovers several interesting points.

First, the most addressed technologies that meet the definition of Adomavicius (2006) [1] are only blockchain and artificial intelligence.

Second, documents that deal with Fintech do not focus on technology but rather on the applications it allows concerning means of payment.

Third, although reported in the sample, we find data noisiness as many papers use the word technology without going into detail but only into hypothetical applications.



Figure 6. Coded technologies. Source: Author's elaboration using ATLAS.Ti Cloud

#### G. Geographical and collaboration analysis

Table VII identifies the frequency of publication and the countries with the most significant interest in technology in Islamic finance. Malaysia (79) is the country with the highest scientific output on the topic, followed by Bahrain (26), Indonesia (26) and the UK (21). The analysis shows that Muslim countries primarily analyse the research topic. However, the topic is also of interest in countries whose financial sector is traditional, such as Italy, the USA, and the UK.

Finally, interesting research insights can be derived from the analysis of Figure 5, which identifies collaborations between academics worldwide.

As can be seen, there is a strong cluster of collaboration between Indonesia and Australia with the USA and between, Turkey and Brazil. These results also represent active research collaborations and discoveries in other research topics such as halal food or Waqaf [27], [29] and reflect faculty teams currently working to increase knowledge in this interesting area.

Country	Frequency
Malaysia	79
Bahrain	26
Indonesia	26
UK	21
USA	21
Italy	19
Pakistan	10
Turkey	9
Nigeria	8
Saudi Arabia	8

Source: Author's elaboration using Biblioshiny



Figure 7. Thematic map. Source: Author's elaboration using Biblioshiny

#### IV. DISCUSSION AND CONCLUSION

Our study used bibliometric analysis to investigate the broad literature about technologies and Islamic finance. We conducted a bibliometric analysis using Bibliometrix R-

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6257 - Published by University of Turin <u>https://www.ojs.unito.it/index.php/EJIF/index</u> EJIF content is licensed under a Creative Commons Attribution 4.0 International License package and an open coding analysis using ATLAS to answer the three research questions Ti Cloud accurately.

Starting from the gap in the literature with numerous studies in this direction but without a systematisation [7], [49], this article provides an in-depth qualitative and quantitative analysis of bibliometric variables. Some valuable theoretical insights can be made.

Firstly, although sustained in the last three years, the scientific production in this field has not had constant levels so far. Moreover, the increase in interest by researchers is primarily sanctioned to seek solutions and answers to the compatibility of Fintech with Shari'ah [9], [48] and to understand the relationships between blockchain and artificial intelligence in the business of Islamic banks [8], [21], [50]. This evidence is demonstrated, for example, in Figure 5, which shows the primary topics of discussion.

Secondly, from a theoretical point of view, the analysis aims to criticise what is reported in the literature constructively. The research shows noisiness in the data collected, which is to be outlined in the Scopus analysis conducted. However, what is noticeable in the study is a homologation of possible technologies for Islamic finance. For example, according to Adomavicius (2006) [1], Understanding the dynamics of technological evolution is the first point of reference for verifying the state of cyclicality and application of many technologies. In the context of Islamic finance, we have observed a stage of embryonic development on a purely scientific level. Moreover, it is unclear what evolutionary state we are in many works, and applications such as Fintech are likened to technologies like blockchain and artificial intelligence. Therefore, this evidence confirms the novelty of scientific literature and the great space it may have for future researchers.

Moreover, our analysis also contains some practical insights. First, Islamic managers and banks should increasingly open up to the technological wave by aspiring, as demonstrated in the literature, synergies to improve and evolve their business models [47], [50]. Second, despite the expansion of technology, as admitted by Adomavicius, the wave of new technologies needs to be managed and controlled in its life cycle. Therefore, financial managers should understand to what extent a technology in use is in its initial or final life cycle. This is instrumental in activating, if necessary, further developments to remain competitive.

Like all research, this one has some limitations. First, the current work does not consider the technical issues of the technologies analysed. Future research could combine a more careful analysis of recent technological problems and variables. Second, the study might be biased using a single research database. Therefore, future research stimuli could conduct integrated, multi-database analyses.

The following table VIII aims to provide researchers with new avenues of research for the future.





[1] G. Adomavicius, J. Bockstedt, A. Gupta, and R. J. Kauffman, "Understanding Patterns of Technology Evolution: An Ecosystem Perspective," in *Proceedings of the 39th Annual Hawaii International Conference on System Sciences (HICSS'06)*, Jan. 2006, vol. 8, pp. 189a–189a. doi: 10.1109/HICSS.2006.515.

[2] V. Jafari-Sadeghi, A. Garcia-Perez, E. Candelo, and J. Couturier, "Exploring the impact of digital transformation on technology entrepreneurship and technological market expansion: The role of technology readiness, exploration and exploitation," *Journal of Business Research*, vol. 124, pp. 100–111, Jan. 2021, doi: 10.1016/j.jbusres.2020.11.020.

[3] D. Calandra and M. Favareto, "Artificial Intelligence to fight COVID-19 outbreak impact: an overview," *European Journal of Social Impact and Circular Economy*, vol. 1, no. 3, Art. no. 3, Dec. 2020, doi: 10.13135/2704-9906/5067.

[4] S. Secinaro, D. Calandra, and P. Biancone, Blockchain, trust, and trust accounting: can blockchain technology substitute trust created by intermediaries in trust accounting? A theoretical examination," *IJMP*, vol. 14, no. 2, p. 129, 2021, doi: 10.1504/IJMP.2021.113824.

[5] S. Secinaro, F. Dal Mas, V. Brescia, and D. Calandra, "Blockchain in the accounting, auditing and accountability fields: a bibliometric and coding analysis," *Accounting*, *Auditing & Accountability Journal*, vol. ahead-of-print, no.

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6257 - Published by University of Turin https://www.ojs.unito.it/index.php/EJIF/index

ahead-of-print, Jan. 2021, doi: 10.1108/AAAJ-10-2020-4987.

[6] N. M. Ariffin, S. Archer, and R. A. A. Karim, "Risks in Islamic banks: Evidence from empirical research," *J Bank Regul*, vol. 10, no. 2, pp. 153–163, Mar. 2009, doi: 10.1057/jbr.2008.27.

[7] M. Hassan, M. Rabbani, and M. Ali, "Challenges for the Islamic Finance and banking in post COVID era and the role of Fintech," *Journal of Economic Cooperation and Development*, vol. 41, pp. 93–116, Aug. 2020.

[8] K. B. Khediri, L. Charfeddine, and S. B. Youssef, "Islamic versus conventional banks in the GCC countries: A comparative study using classification techniques," *Research in International Business and Finance*, vol. 33, pp. 75–98, Jan. 2015, doi: 10.1016/j.ribaf.2014.07.002.

[9] M. I. B. Ab Razak, N. A. Mohd Dali, G. Dhillon, and A. W. Manaf, "Fintech In Malaysia: An Appraisal to the Need of Shariah-Compliant Regulation," *Pertanika Journal of Social Science and Humanities*, vol. 28, pp. 3223–3233, Dec. 2020, doi: 10.47836/pjssh.28.4.40.

[10] N. Khan, B. Kchouri, N. A. Yatoo, Z. Kräussl, A. Patel, and R. State, "Tokenization of sukuk: Ethereum case study," *Global Finance Journal*, p. 100539, Jun. 2020, doi: 10.1016/j.gfj.2020.100539.

[11] M. Radwan, D. Calandra, and P. Koumbarakis, "Takaful industry and Blockchain: challenges and opportunities for costs' reduction in Islamic insurance companies," *European Journal of Islamic Finance*, Oct. 2020, doi: 10.13135/2421-2172/4926.

[12] R. Hasan, M. K. Hassan, and S. Aliyu, "Fintech and Islamic Finance: Literature Review and Research Agenda," *International Journal of Islamic Economics and Finance* (*IJIEF*), vol. 3, no. 1, Art. no. 1, Jan. 2020, doi: 10.18196/ijief.2122.

[13] A. Delle Foglie, I. C. Panetta, E. Boukrami, and G. Vento, "The impact of the Blockchain technology on the global Sukuk industry: smart contracts and asset tokenisation," *Technology Analysis & Strategic Management*, vol. 0, no. 0, pp. 1–15, Jun. 2021, doi: 10.1080/09537325.2021.1939000.

[14] B. Uluyol, "A comprehensive empirical and theoretical literature survey of Islamic bonds (sukuk)," *Journal of Sustainable Finance & Investment*, vol. 0, no. 0, pp. 1–23, Apr. 2021, doi: 10.1080/20430795.2021.1917224.

[15] I. Zupic and T. Čater, "Bibliometric methods in management and organization," *Organizational Research Methods*, vol. 18, no. 3, pp. 429–472, 2015.

[16] P. Mongeon and A. Paul-Hus, "The journal coverage of Web of Science and Scopus: a comparative analysis," *Scientometrics*, vol. 106, no. 1, pp. 213–228, 2016.

[17] M. Massaro, J. Dumay, and J. Guthrie, "On the shoulders of giants: undertaking a structured literature review in accounting," *Accounting Auditing & Accountability Journal*, vol. 29, no. 5, pp. 767–801, 2016, doi: 10.1108/AAAJ-01-2015-193.

[18] M. R. Rabbani, S. Khan, and E. I. Thalassinos, "FinTech, Blockchain and Islamic Finance: An Extensive Literature Review," *International Journal of Economics and Business Administration*, vol. VIII, no. 2, pp. 65–86, Apr. 2020.

[19] D. Tranfield, D. Denyer, and P. Smart, "Towards a Methodology for Developing Evidence-Informed Management Knowledge by Means of Systematic Review," *Br J Management*, vol. 14, no. 3, pp. 207–222, Sep. 2003, doi: 10.1111/1467-8551.00375.

[20] H. Zarrouk, T. E. Ghak, and E. A. A. Haija, "Financial development, Islamic finance and economic growth: evidence of the UAE," *Journal of Islamic Accounting and Business Research*, Feb. 2017, doi: 10.1108/JIABR-05-2015-0020.

[21] M. Kamdzhalov, "Islamic Finance and the New Technology Challenges," *European Journal of Islamic Finance*, Mar. 2020, doi: 10.13135/2421-2172/3813.

[22] J. Dumay and L. Cai, "A review and critique of content analysis as a methodology for inquiring into IC disclosure," *Journal of Intellectual Capital*, vol. 15, no. 2, pp. 264–290, Jan. 2014, doi: 10.1108/JIC-01-2014-0010.

[23] G. Chen and L. Xiao, "Selecting publication keywords for domain analysis in bibliometrics: A comparison of three methods," *Journal of Informetrics*, vol. 10, no. 1, pp. 212–223, Feb. 2016, doi: 10.1016/j.joi.2016.01.006.

[24] A. de Bem Machado, S. Secinaro, D. Calandra, and F. Lanzalonga, "Knowledge management and digital transformation for Industry 4.0: a structured literature review," *Knowledge Management Research & Practice*, vol. 0, no. 0, pp. 1–19, Dec. 2021, doi: 10.1080/14778238.2021.2015261.

[25] P. Rosa, C. Sassanelli, A. Urbinati, D. Chiaroni, and S. Terzi, "Assessing relations between Circular Economy and Industry 4.0: a systematic literature review," *International Journal of Production Research*, vol. 58, no. 6, pp. 1662– 1687, Mar. 2020, doi: 10.1080/00207543.2019.1680896.

[26] M. Massaro, S. Secinaro, F. Dal Mas, V. Brescia, and D. Calandra, "Industry 4.0 and circular economy: An exploratory analysis of academic and practitioners' perspectives," *Business Strategy and the Environment*, vol. n/a, no. n/a, 2020, doi: https://doi.org/10.1002/bse.2680.

[27] S. Secinaro and D. Calandra, "Halal food: structured literature review and research agenda," *British Food Journal*, vol. 123, no. 1, pp. 225–243, 2020, doi: 10.1108/BFJ-03-2020-0234.

[28] C. Okoli and K. Schabram, "A Guide to Conducting a Systematic Literature Review of Information Systems Research," *SSRN Electronic Journal*, vol. Sprouts: Working Papers on Information Systems, 2010, doi: 10.2139/ssrn.1954824.

[29] M. Massaro, J. Dumay, and A. Garlatti, "Public sector knowledge management: a structured literature review," *Journal of Knowledge Management*, vol. 19, no. 3, pp. 530–558, Jan. 2015, doi: 10.1108/JKM-11-2014-0466.

[30] B. Uluyol, S. Secinaro, D. Calandra, and F. Lanzalonga, "Mapping waqf research: a thirty-year bibliometric analysis," *Journal of Islamic Accounting and Business Research*, vol. 12, no. 5, pp. 748–767, Jan. 2021, doi: 10.1108/JIABR-01-2021-0031.

[31] R. N.-U.-D. Jalal, I. Alon, and A. Paltrinieri, "A bibliometric review of cryptocurrencies as a financial asset,"

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6257 - Published by University of Turin <a href="https://www.ojs.unito.it/index.php/EJIF/index">https://www.ojs.unito.it/index.php/EJIF/index</a>
*Technology Analysis & Strategic Management*, vol. 0, no. 0, pp. 1–16, Jun. 2021, doi: 10.1080/09537325.2021.1939001.

[32] A. Liberati *et al.*, "The PRISMA statement for reporting systematic reviews and meta-analyses of studies that evaluate health care interventions: explanation and elaboration," *PLoS medicine*, vol. 6, no. 7, p. e1000100, 2009. [33] M. Aria and C. Cuccurullo, "bibliometrix: An R-tool for comprehensive science mapping analysis," *Journal of Informetrics*, vol. 11, no. 4, pp. 959–975, 2017, doi: 10.1016/j.joi.2017.08.007.

[34] S. Hwang, "Utilizing Qualitative Data Analysis Software: A Review of Atlas.ti," *Social Science Computer Review*, vol. 26, no. 4, pp. 519–527, Nov. 2008, doi: 10.1177/0894439307312485.

[35] J. Al-Ammary and S. Hamad, "Information technology for enhancing NGO performance in the Kingdom of Bahrain," *International Journal of Electronic Commerce Studies*, vol. 3, no. 1, Art. no. 1, Aug. 2012, doi: 10.7903/ijecs.1017.

[36] S. Secinaro, V. Brescia, D. Calandra, and P. Biancone, "Employing bibliometric analysis to identify suitable business models for electric cars," *Journal of Cleaner Production*, vol. 264, p. 121503, Aug. 2020, doi: 10.1016/j.jclepro.2020.121503.

[37] L. Gheeraert and L. Weill, "Does Islamic banking development favor macroeconomic efficiency? Evidence on the Islamic finance-growth nexus," *Economic Modelling*, vol. 47, pp. 32–39, Jun. 2015, doi: 10.1016/j.econmod.2015.02.012.

[38] D. Rudnyckyj, "Economy in practice: Islamic finance and the problem of market reason," *American Ethnologist*, vol. 41, no. 1, pp. 110–127, 2014, doi: 10.1111/amet.12063.

[39] Mohammad, S. Khan, Mustafa, and Yannis, "An Artificial Intelligence and NLP based Islamic FinTech Model Combining Zakat and Qardh-Al-Hasan for Countering the Adverse Impact of COVID 19 on SMEs and Individuals," *IJEBA*, vol. VIII, no. Issue 2, pp. 351–364, Apr. 2020, doi: 10.35808/ijeba/466.

[40] A. DIAW, "The global financial crisis and Islamic finance: a review of selected literature," *Journal of Islamic Accounting and Business Research*, vol. 6, no. 1, pp. 94–106, Jan. 2015, doi: 10.1108/JIABR-03-2012-0015.

[41] P. P. Biancone and M. Radwan, "Social Finance And Unconventional Financing Alternatives: An Overview," *European Journal of Islamic Finance*, vol. 0, no. 10, Jul. 2018, doi: 10.13135/2421-2172/2818.

[42] S. Secinaro, D. Calandra, D. Petricean, and F. Chmet, "Social Finance and Banking Research as a Driver for Sustainable Development: A Bibliometric Analysis," *Sustainability*, vol. 13, no. 1, Art. no. 1, Jan. 2021, doi: 10.3390/su13010330.

[43] E. Garfield, "Historiographic Mapping of Knowledge Domains Literature," *Journal of Information Science*, vol. 30, no. 2, pp. 119–145, Apr. 2004, doi: 10.1177/0165551504042802.

[44] U. A. Oseni and S. O. Omoola, "Banking on ICT: The relevance of online dispute resolution in the Islamic banking industry in Malaysia," *Information & Communications Technology Law*, vol. 24, no. 2, pp. 205–223, May 2015, doi: 10.1080/13600834.2015.1067968.

[45] V. Bakhshi and A. Krajeski, "Accounting for climate change: A window on the future," *Harv. Bus. Rev.*, vol. 85, no. 10, pp. 36-+, Oct. 2007.

[46] S. Secinaro, V. Brescia, D. Calandra, and B. Saiti, "Impact of climate change mitigation policies on corporate financial performance: Evidence-based on European publicly listed firms," *Corporate Social Responsibility and Environmental Management*, vol. n/a, no. n/a, 2020, doi: 10.1002/csr.1971.

[47] M. Oppenheimer, "Climate change impacts: accounting for the human response," *Climatic Change*, vol. 117, no. 3, pp. 439–449, Apr. 2013, doi: 10.1007/s10584-012-0571-9.

[48] P. P. Biancone, S. Secinaro, and M. Kamal, "Crowdfunding and Fintech: business model sharia compliant," *European Journal of Islamic Finance*, vol. 0, no. 12, Apr. 2019, doi: 10.13135/2421-2172/3260.

[49] N. Naifar, Impact of Financial Technology (FinTech) on Islamic Finance and Financial Stability. IGI Global, 1AD. Accessed: Oct. 05, 2021. [Online]. Available: https://www.igiglobal.com/book/impact-financial-technology-fintechislamic/www.igi-global.com/book/impact-financialtechnology-fintech-islamic/225847

[50] P. P. Biancone, B. Saiti, D. Petricean, and F. Chmet, "The bibliometric analysis of Islamic banking and finance," *Journal of Islamic Accounting and Business Research*, vol. ahead-of-print, no. ahead-of-print, Jan. 2020, doi: 10.1108/JIABR-08-2020-0235.

[51] F. H. L. Chong, "Enhancing trust through digital Islamic finance and blockchain technology," *Qualitative Research in Financial Markets*, vol. 13, no. 3, pp. 328–341, Jan. 2021, doi: 10.1108/QRFM-05-2020-0076.

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6257 - Published by University of Turin https://www.ojs.unito.it/index.php/EJIF/index

# Trust Enhancement in Zakat Institutions using Blockchain Technology: A Qualitative Approach

Zulfikri<sup>1,2</sup>, Auwal Adam Sa'ad<sup>2</sup>, Salina Kassim<sup>2</sup>, Anwar Hassan Abdullah Othman<sup>2</sup>

<sup>1</sup> Department of Management, Universitas Muhammadiyah Sumatera Utara, Indonesia <sup>2</sup> IIUM Institute of Islamic Banking and Finance, Malaysia

Contact Author: zulfikri@umsu.ac.id

Abstract- There is an indication that zakat institutions are inefficient and lack transparency in terms of distributing and managing their funds. This leads to a lack of trust between Muzakee and zakat institutions; hence, they prefer to pay zakat to Mustahiq directly. This means that zakat institutions may be collecting far less in zakat funds than the maximum amount that is potentially available. Meanwhile, the invention of blockchain technology has disrupted the financial sector as it offers transparency in transactions, meaning intermediaries are no longer needed. The use of blockchain in zakat management is important for improving transparency in the distribution of zakat. The objective of this paper is to explore the determinant factors that influence trust in zakat institutions using blockchain technology as a mediating effect. The results of the literature review revealed that these factors are reputation, satisfaction with zakat distribution, service quality and disclosure practice. This study expects that trust in zakat institutions can be enhanced through the use of blockchain technology.

Keywords: Blockchain, Zakat, Trust, Zakat institution

### INTRODUCTION

Zakat is one of the financial tools and resources used in Islamic economic development. Zakat funds are designed to help people or communities in need and every Muslim's responsibility to pay zakat forms part of their worship and duties. The goal of the Zakat Fund is to minimize economic gaps among communities, which is only achieved if zakat institutions succeed in collecting large amounts of Zakat. According to Act No. 23, 2011, Zakat management in Indonesia is carried out by two institutions, namely The National Amil Zakat Agency (Badan Amil Zakat Nasional – BAZNAS) and The National Amil Zakat Institution (Lembaga Amil Zakat – LAZ). BAZNAS manages zakat at the national level and is formed by the government, while LAZ is formed at the community level and is responsible for the collection, distribution and utilization of zakat. To assist in the collection of zakat, BAZNAS formed a separate organization, the Zakat Collection Unit (UPZ) as noted by Andayani, Hanum, Zaenal, Asmita, Damayanti, Fahrudin & Kardiman [1]

Poverty is one of the many problems facing Indonesia. Zakat has been proven to be effective in reducing or eradicating poverty, with prominent examples from the era of the Second Caliph of Islam, Hazrat Umar bin Khattab (R.A) and of Umar bin Abdul Aziz, who was the First Caliph of Islam for just over one year (99-101H) as noted by Hudayati & Tohirin [2]. As such, Zakat, along with Infaq, Waqf and Sadaqah, is an example of Islamic philanthropy that has a significant impact on alleviating poverty. As set out in Law No. 23 of 2011, BAZNAS is the agency or institution responsible for the overall management of zakat in Indonesia.

As the most prominent Muslim country globally, Indonesia has the potential to collect a large amount of zakat funds. According to the BAZNAS report of 2017, Indonesia boasted a potential zakat fund of Rp 286 trillion. BAZNAS collects many types of zakat, including from households, private companies and state enterprises, in the form of deposits and savings. However, the entire zakat operation collected only Rp 3.7 trillion in 2015, thus indicating a significant difference between the actual and potential amounts collected as noted by Firdaus, Beik, Irawan & Juanda [3], meanwhile, found that the total size of the zakat fund could potentially reach Rp 217 trillion (combined amount from households, industries and deposits). This figure equates to 11.45% of Indonesia's national income from 2018. However, data from BAZNAS in 2017 show that the total zakat collection in 2016 amounted to only Rp 5 trillion.

Currently, the amount of zakat funds collected lags far behind its potential, which in turn means that the amount of

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6312 - Published by University of Turin <a href="https://www.ojs.unito.it/index.php/EJIF/index">https://www.ojs.unito.it/index.php/EJIF/index</a>

funds distributed automatically to eight Asnaf (*Mustahiq*) is not optimized. Many factors explain why BAZNAS has not received the maximum amount of funds available. One issue is a distrust of zakat institutions among the *Muzakee* as reported by BAZNAS [4]. This distrust stems partly from skepticism among the *Muzakee* concerning the ability of zakat institutions to distribute and manage their funds efficiently, due to a lack of transparency as reported by BAZNAS [4]. Thus, many *Muzakee* continue to pay zakat directly to the *Mustahiq*. These issues must be urgently addressed to further optimize the zakat collection.

Several previous studies have examined the use of blockchain in zakat management as a way of improving trust in zakat institutions. These studies include Beik, Nurzaman & Sari [5]; Rejeb [6]; Nienhaus [7]; Nor, Abdul-Majid & Esrati [8]; Zulfikri, Kassim, & Hawariyuni [9]. These studies focused predominantly on the creation of a model and mechanism for implementing blockchain in zakat management, but none measured the effect of blockchain technology on influencing trust in zakat institutions. Hence, this study attempts to fill this gap by measuring the effect of blockchain technology in improving trust in zakat institutions using a qualitative approach and by developing a conceptual model.

# II. LITERATURE REVIEW

# A. Blockchain technology

Bitcoin is a type of digital currency based on cryptography that was developed by Satoshi Nakamoto in 2008 as noted by Abubakar, Ogunbado, & Saidi [10]. However, Kamaruzaman, Yassin, Zabidi, Zaman, Rizman, Baharom & Wahab [11] argued that Satoshi Nakamoto is a fictitious name. Nakamoto published a paper entitled 'Bitcoin: A Peer-To-Peer Electronic Cash System' describing a peer-to-peer version of electronic cash that would enable online payments to be sent directly from one party to another, with no need to pass through a financial institution or intermediary as concluded in Crosby, Nachiappan, Pattanayak, Verma, & Kalyanaraman [12]

While the bitcoin innovation was thrilling and groundbreaking, it was the mechanism behind its flawless operation that was the real story. Soon after Nakamoto's paper was released, it became clear that the core mechanical invention was not the digital currency but the technology operating it. As part of the bitcoin operation, Nakamoto also developed a ledger called 'A chain of Blocks', which later became the more familiar 'blockchain' as noted by Morkunas, Paschen, & Boon [13]. Blockchain and bitcoin are therefore linked as blockchain is the public ledger used for bitcoin currency as note by Zubaidi & Abdullah [14]

Blockchain is essentially a decentralized transaction ledger that can be used to create, authorize and send transactions to other nodes within the same network as noted by Tama, Kweka, Park, & Rhee [15]. It combines the basics of cryptography, peer-to-peer networking and game theory. While blockchain was created to track the database underlying the cryptocurrency, which is bitcoin, it is now commonly regarded as a distributed ledger with a software algorithm to record transactions as a chain of blocks safely and anonymously as noted by Laroiya, Saxena, & Komalavalli [16]

Although blockchain is often linked solely to bitcoin, many other blockchain applications have been developed since Nakamoto first introduced it. Despite being a ledger for digital currencies, the use of blockchain as an application has extended further to impact the economy, financial sector and beyond as noted by Crosby, Nachiappan, Pattanayak, Verma, & Kalyanaraman [12]; Zubaidi & Abdullah [14]. According to Lakhani & Iansiti [17] blockchain has gained significant recognition to potentially become the new Internet and is proving transformative in terms of enabling organizations to modify how they both generate and take value.

Since its initial development, blockchain has grown in popularity as a tool with a decentralized transaction ledger that can be used to register, confirm and send payments or contracts. The use of blockchain technology has also extended far beyond financial transactions to include a much broader range of transactions and applications such as healthcare, utilities, real estate and government as noted by Christidis & Devetsikiotis [18]. Wang, Zheng, Xie, Dai, and Chen [19] acknowledged that while blockchain is famous for bitcoin, it can also be applied to diverse applications beyond cryptocurrencies as it enables payments to be made with no involvement from banks or any intermediary. With this in mind, Morkunas, Paschen & Boon [13] stated that blockchain is expected to disrupt existing business models and propose new value creation.

# B. Trust in Zakat Institutions

Trust plays a crucial role in society as noted by Abdul-Rahman & Hailes [20]. Previous researchers have described trust as 'general trust for others'. Sahidi [21], meanwhile, stated that the role of zakat institutions in providing a quality service to the community is an essential factor in attracting and encouraging entrepreneurs to pay zakat.

As a non-profit organization, trust is an essential foundation for BAZNAS to achieve its objectives. Handriana [22] reported that the intention of *Muzakee* to maintain LAZ is influenced by their trust in the institution. Yang, Brennan, and Wilkinson [23], meanwhile, asserted that a trustworthy charity meets the public's expectations. Hence, both LAZ and BAZNAS must work tirelessly to both examine and meet the demands of the public in order to gain their trust. In this way, the presence of trust will strengthen the commitment of *Muzakee* to pay zakat to zakat management institutions and thus maintain the sustainability and life of the programme. If they have trust in LAZ, people become more confident that their zakat funds will be paid to zakat institutions.

An examination of the relevant past studies reveals that trust is influenced by reputation, satisfaction with zakat distribution and the quality of the service provided. Zainal, Abu Bakar & Saad [24] showed that the greater the reputation, satisfaction with zakat distribution and service quality factors, the higher the trust of stakeholders in zakat institutions, and vice versa.

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6312 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index

#### C. Reputation and Trust in Zakat Institutions

Reputation has a significant effect on trust in zakat institutions as noted by Mukhibad, Fachrurrozie, & Nurkhin [25]. The concept of reputation in a non-profit organization has been widely explored in the economic and sociological literature and given numerous definitions. Reputation can be defined as the accumulated stakeholder assessment of an organization that develops over time and is a result of consistent performance and communication as noted by Gray & Balmer [26]. Richard and Zhang [27], in their study, defined reputation as the overall perception of stakeholders of the company's performance over time. This is in line with the definition from the Concise Oxford Dictionary, as cited by Zainal, Abu Bakar & Saad [24], that reputation constitutes what is generally said or believed about the character or standing of a person or thing.

Meanwhile, Abratt and Kleyn [28] identified that citizens trust a company based on past product experience and the company's reputation. As such, the company's reputation is the final determinant in meeting consumer expectations and demands as noted by Abratt & Kleyn [28]. Extending this, reputation has a significant influence on trust in zakat institutions.

## D. Satisfaction with Zakat Distribution

Customer satisfaction is the key to an organization's success, which is especially relevant in the case of servicebased organizations. According to Oliver [29], satisfaction can usually be viewed as an evaluation of the market in terms of the extent to which the preferences of the customer match the actual service rendered by an organization. Satisfaction with the distribution of zakat should be improved to maintain the performance of zakat and thus encourage Muslims to pay it through the relevant zakat institutions as noted by Ellany & Lateff [30]. According to Idris & Ayob [31] distrust of zakat institutions leads to non-compliance, especially when that distrust relates to transparency and inefficiencies in the management of zakat distribution. The greater the satisfaction with zakat institutions, the higher the adherence to paying zakat. In accordance with the above claims, a stakeholder who is highly satisfied with the distribution of zakat would be expected to have greater trust in zakat institutions, and vice versa. Therefore, satisfaction with zakat distribution has a significant influence on trust in zakat institutions.

### E. Service Quality and Trust in Zakat Institutions

Service quality is closely related to customer satisfaction. According to Gronroos [32], service quality can be defined as how well a service meets the expectation of the customer. Zeithaml [33], meanwhile, stated that service quality refers to a customer's overall assessment of the advantages and value of the product or service. Service quality is measured using a small number of evaluative dimensions that have met previous research expectations. Parasuraman, Zeithaml, & Berry [34] point to the use of SERVQUAL as a measurement of service quality. In the SERVQUAL model, service quality is made up of the five dimensions of reliability, responsiveness, assurance, empathy and tangibles. To maintain the efficiency of zakat, zakat institutions must increase the amount of zakat payments as noted by Zainal, Abu Bakar, & Saad [24]. Previous studies reported various findings regarding satisfaction with zakat distribution. Among these, Md. Idris & Ayob [31] found that widespread non-compliance was caused by the poor administration of zakat-by-zakat institutions in terms of their transparency and efficiency. As such, greater levels of satisfaction with zakat institutions were more strongly correlated with higher zakat compliance. In line with the above arguments, it is expected that a high level of satisfaction among stakeholders will lead to greater trust in zakat institutions. Therefore, service quality has a significant influence on trust in zakat institutions.

# F. Disclosure Practice and Trust in Zakat Institutions

Financial statements serve as a reliable means of assessing the success of the management and financial condition of an organisation as noted by Samargandi, Tajularifin, Ghani, Aziz & Gunardi [35]. These financial statements contain information that is valuable for various people and it is therefore the responsibility of the zakat board of management to ensure their credibility as documents as noted by Karim [36]. As the primary shareholders in non-profit organizations, donors and investors turn to published financial statements as their main source of information. Zakat institutions must disclose a wide range of financial and other related details related to their expenditure. This information is necessary to demonstrate the credibility of the organization as noted by Zabri & Mohammed [37].

A significant difference often exists, however, between the size of the potential zakat fund and the amount of zakat actually received by zakat institutions in Indonesia. Problems with zakat funds resulted from mistrust in zakat institutions due to the quality of the financial information that they provide. This information is needed when making decisions on whether zakat institutions should be trusted to obtain, use and distribute zakat funds that have been entrusted to them by zakat payers as noted by Mediawati [38].

Meanwhile, Jayanto, Jayanto, & Munawaroh [39] stated that the disclosure of financial statements is expected to enhance the public's reasoning agility and increase their interest in paying the zakat profession. The transparency of zakat institutions has a significant effect on *Muzakee* due to the close relationship between transparency and the receipt of zakat funds as noted by Jayanto, Jayanto, & Munawaroh [39]. Transparency entails the disclosure of information to stakeholders concerning the use of funds. Schnackenberg & Tomlinson [40] identified three main elements of transparency: information disclosure, clarity and accuracy. Based on this, it can be concluded that disclosure practice has a significant influence on trust in zakat institutions.

### G. Mediating Effect of Blockchain Technology

Blockchain is a technology that offers the potential to significantly improve the traceability of food-related products not only for companies but also consumers and policymakers. Blockchain facilitates the establishment of a distributed network that holds records of digital assets in a decentralized

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6312 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index

manner. Abojeib & Habib [41] highlighted the continuous efforts being made to apply blockchain technology in the social and financial system based on the system's good governance, low transaction costs and high transparency.

Given their role in building a strong zakat environment, the amil zakat (zakat officer) is critical to the collection of zakat funds. Hamdani [42] asserted that at least four factors contribute to a plentiful zakat collection that can alleviate poverty and bring prosperity to the people. These factors include the Amil, who must be trustworthy, professional, fair and responsible. Blockchain, for its part, makes processes traceable, auditable and irreversible, which is the most important aspect of successful philanthropy. Based on this, it is expected that blockchain technology has a significant influence on trust in zakat institutions.

# III. RESEARCH METHODOLOGY

This study is qualitative in nature. It utilized an extensive review of the literature to identify and analyze the relevant studies to propose the conceptual model. The systematic review of the literature promotes new ideas, the discovery of new alternatives, and the development of new hypotheses and research proposals, as noted by Fiegen [43].

According to Khan [44], the process of conducting a systematic literature review includes framing questions for review; identifying relevant literature; assessing quality; summarizing evidence; and interpreting findings.

To perform systematic literature review, this study used Google Scholar and Emerald Insight database. The papers retrieved from reputed journal databases range from 2015 to the present, focusing on the use of blockchain in zakat management. Books and book chapters also were included in the analysis. The analytic results of this literature review produce a conceptual framework.

# IV. RESULTS AND DISCUSSION

This study develops an initial conceptual framework of factors influencing trust in zakat institutions with blockchain as mediating variable. We examine the literature in the context of trust and blockchain technology.

### Reputation

Reputation has significant effect on trust in zakat institutions. A non-profit organization like BAZNAS needs such a reputation to stimulate the trust of zakat payer. Furthermore, studies have found a positive relationship between reputation and consumer trust as noted by Oladimeji, Har Sani & Akhyar [45]. Therefore, it is hypothesized that Reputation significantly influences trust in zakat institutions.

### Satisfaction with Zakat Distribution

Trust in zakat institutions is strongly influenced by the satisfaction of distribution zakat. Mohammed Dahan in Saad, Farouk & Abdul Kadir [46] argued that poor zakat distribution management leads to a bad image of zakat institutions, which in turn affects zakat collection. Meanwhile, Zainal, Abu Bakar & Saad [24] stated that low confidence in Zakat institutions can be attributed to a lack of transparency and inefficiency in zakat

distribution. Thus, Satisfaction with Zakat Distribution significantly influences trust in zakat institutions.

### Service Quality

As an institution that manages public funds, zakat institutions' credibility is also affected by how their customers perceive their services. In general, service quality can be described as the difference between what customers expect and what they get as noted by Ghani, Said & Syed Yusuf [47]. According to Parasuraman, Zeithaml, & Berry [34], If service expectations are higher than the service performance, then the perceived quality of the service is less than satisfactory, resulting in customers' dissatisfaction. This finding shows the importance of having a good service delivery (service quality). Therefore, Service Quality significantly influences trust in zakat institutions.

### **Disclosure** Practice

As non-profit organization, BAZNAS need to provide information that is useful to a variety of users including *Muzakee* such as financial statements. The concept of disclosure refers to the perceptions of transparency exhibited by an entity. Information and data produced by companies fall under this category, including financial and non-financial data, quantitative and qualitative data, required or voluntary information, and information produced through formal and informal means. Information may be disclosed through different mechanisms, such as annual reports, press releases, investor relations, periodic reports, brochures, and websites as noted by Hassan & Marston [48].

With regard to zakat institutions, the early practices of Prophet Muhammad (SAW) and his pious successors exhibited transparency with regard to zakat proceeds, particularly regarding their storage and distribution. According to Al-Njrani in Sawmar & Muhammad [49], Zakat payers and authorities should embrace transparency hence, Zakat payers are expected to disclose their wealth subject to zakat to zakat collectors.

It is important that information about the activities of zakat institutions be disclosed through annual reports and verified by trustworthy auditors. By viewing the information disclosed, zakat payers may be more inclined to pay zakat in the future since they will have a sense of mutual trust. Therefore, Disclosure Practice significantly influences trust in zakat institutions.

#### Blockchain Technology

Blockchain technology can optimize trust in zakat institutions by offering transparency. There will only be two parties involved in a blockchain-based transaction. The transaction itself and its decentralized validation. The originality of the blockchain technology is at the level of validation, as there is no third party involved as noted by Chong [50].

Technically, blockchain is distributed, transparent, immutable, validated, secured and pseudo-anonymous. Blockchain data enables one to create an incorruptible and resistant to censorship secure legal identity, referred to as a "trust stamp". Using this technology, smart contracts cannot be

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6312 - Published by University of Turin https://www.ojs.unito.it/index.php/EJIF/index

altered or removed after they are written as noted by Chong [50]

Given the advantage of blockchain technology in term of transparency, thus, we proposed the blockchain technology as the mediating effect that could enhance trust in zakat institutions.

Based on the discussion above, this study proposed research model shown in Figure 1. The model postulates that trust in zakat institutions is influenced by Reputation, Satisfaction with zakat distribution, Service quality and Disclosure practice. The framework also postulates that the independent variables are expected to have a direct positive influence on the dependent variable. Meanwhile, blockchain technology strengthens the direct relationship between the independent variables and dependent variable.



#### V. CONCLUSION, IMPLICATION AND FUTURE RESEARCH

This paper aimed to explore the determinant factors that influence trust in zakat institutions, where the use of blockchain was expected to enhance this trust. From the literature review, reputation, satisfaction with zakat distribution, service quality and disclosure practice are the determinant factors influencing trust in zakat institutions. In addition, this study suggests that blockchain technology has a mediating effect and is expected to enhance trust in zakat institutions. The proposed research model can be used to conduct further research on trust in zakat institutions' issues, while policymakers in zakat institutions can use the findings and analysis of this study to enhance Muzakees' level of trust through the implementation of blockchain technology. This, in turn, would help to maximize the level of zakat funds collected in the near future. This paper is conceptual in nature; therefore, no empirical evidence is provided. Further research could use the survey method to validate and examine the predictive power of the proposed model.

#### REFERENCES

- Andayani, D. R., Hanum, K., Zaenal, M. H., Asmita, B., Damayanti, D. R., Fahrudin, & Kardiman, M. (2019). Statistik Zakat Nasional 2018. Retrieved from <u>https://baznas.go.id/szn2018</u>
- [2] Hudayati, A., & Tohirin, A. (2010). Management of zakah: Centralised vs decentralised approach. Seventh International Conference – The Tawhidi Epistemology: Zakat and Waqf Economy, Bangi 2010 (June), 351–374.
- [3] Firdaus, M., Beik, I. S., Irawan, T., & Juanda, B. (2012). Economic estimation and determinations of Zakat potential in Indonesia. Jeddah: Islamic Research and Training Institute.
- [4] Baznas, P. (2017). Outlook Zakat Indonesia. Jakarta. Baznas Pusat.
- [5] Beik, I. S., Nurzaman, M. S., & Sari, A. P. (2019). Zakat Standard Framework of Halal Cryptocurrency. In Halal Cryptocurrency Management (pp. 275-284). Palgrave Macmillan, Cham.
- [6] Rejeb, D. (2020). Blockchain and Smart Contract Application for Zakat Institution. International Journal of Zakat, 5(3), 20-29.
- [7] Nienhaus, V. (2019). Blockchain technologies and the prospects of smart contracts in Islamic finance. Fintech in Islamic Finance, 183.
- [8] Nor, S. M., Abdul-Majid, M., & Esrati, S. N. (2021). The role of blockchain technology in enhancing Islamic social finance: the case of Zakah management in Malaysia. foresight.
- [9] Zulfikri, Z., Kassim, S. H., & Hawariyuni, W. (2021). Proposing Blockchain Technology Based Zakat Management Model to Enhance Muzakki's Trust in Zakat Agencies: A Conceptual Study. Journal of Accounting Research, Organization and Economics, 4(2), 153-163.
- [10] Abubakar, Y. S., Ogunbado, A. F., & Saidi, M. A. (2018). Bitcoin and its legality from Shariah point of view. SEISENSE Journal of Management, 1(4), 13–21. https://doi.org/10.33215/sjom.v1i4.32
- [11] Kamaruzaman, N. E., Yassin, I. M., Zabidi, A., Zaman, F. H. K., Rizman, Z. I., Baharom, R., & Wahab, N. A. (2018). Blockchain technology for Islamic marriage certificate. International Journal of Engineering and Technology (UAE), 7(4), 193–197. https://doi.org/10.14419/ijet.v7i4.11.20802
- [12] Crosby, M., Nachiappan, Pattanayak, P., Verma, S., & Kalyanaraman, V. (2016). Blockchain technology: Beyond bitcoin. Applied Innovation Review, 2.
- [13] Morkunas, V. J., Paschen, J., & Boon, E. (2019). How blockchain technologies impact your business model. Business Horizons, 62(3), 295–306. https://doi.org/10.1016/j.bushor.2019.01.009
- [14] Zubaidi, I. B., & Abdullah, A. (2017). Developing a digital currency from an Islamic perspective: Case of blockchain technology. International Business Research, 10(11), 79. https://doi.org/10.5539/ibr.v10n11p79
- [15] Tama, B. A., Kweka, B. J., Park, Y., & Rhee, K. H. (2017, August). A critical review of blockchain and its current applications. In 2017 International Conference on Electrical Engineering and Computer Science (ICECOS) (pp. 109-113). IEEE.
- [16] Laroiya, C., Saxena, D., & Komalavalli, C. (2020). Applications of blockchain technology. In S. Krishnan, V. E. Balas, E. G. Julie, Y. H. Robinson, S. Balaji, & R. Kumar (Eds.), Handbook of Research on Blockchain Technology (pp. 213–243). Elsevier https://doi.org/10.1016/B978-0-12-819816-2.00009-5
- [17] Lakhani, K. R., & Iansiti, M. (2017). The truth about blockchain. Harvard Business Review, Jan-Feb 2017, 118–127. Retrieved from https://hbr.org/2017/01/the-truth-about-blockchain
- [18] Christidis, K., & Devetsikiotis, M. (2016). Blockchains and smart contracts for the Internet of Things. IEEE Access. https://doi.org/10.1109/ACCESS.2016.2566339
- [19] Wang, H., Zheng, Z., Xie, S., Dai, H. N., & Chen, X. (2018). Blockchain challenges and opportunities: a survey. International Journal of Web and Grid Services, 14(4), 352. https://doi.org/10.1504/ijwgs.2018.10016848
- [20] Abdul-Rahman, A., & Hailes, S. (2000). Supporting trust in virtual communities. Proceedings of the Hawaii International Conference on System Sciences. https://doi.org/10.1109/hicss.2000.926814

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6312 - Published by University of Turin https://www.ojs.unito.it/index.php/EJIF/index

- [21] Mohamad Noor Sahidi bin Johari. (2013). Penentuan Institusi Zakat Secara Menyeluruh. Jurnal Pengurusan JAWHAR, 7(2)
- [22] Handriana, T. (2016). Analysis of Donor's Trust and Relationship Commitment in Lembaga Amil Zakat in Indonesia. Journal of Economics, Business, and Accountancy Ventura, 19 (1), 59 – 68. doi:10.14414/jebav.v19i1.529
- [23] Yang, Y., Brennan, I., & Wilkinson, M. (2014). Public trust and performance measurement in charitable organizations. International Journal of Productivity and Performance Management, 63(6), 779–796. https://doi.org/10.1108/IJPPM-09-2013-0159
- [24] Zainal, H., Abu Bakar, A., & Saad, R. A. J. (2016). Reputation, satisfaction of zakat distribution, and service quality as determinant of stakeholder trust in zakat institutions. International Journal of Economics and Financial Issues, 6(7), 72–76.
- [25] Mukhibad, H., Fachrurrozie, F., & Nurkhin, A. (2019). Determinents of the intention of muzakki to pay professional zakat. Jurnal Ekonomi Dan Keuangan Islam, 8(1), 45–67. https://doi.org/10.22373/share.v8i1.457
- [26] Gray, E. R., & Balmer, J. M. T. (1998). Managing corporate image and corporate reputation. Long Range Planning, 31(5), 695–702. https://doi.org/10.1016/S0024-6301(98)00074-0
- [27] Richard, J. E., & Zhang, A. (2012). Corporate image, loyalty, and commitment in the consumer travel industry. Journal of Marketing Management, 28(5–6), 568–593. https://doi.org/10.1080/0267257X.2010.549195
- [28] Abratt, R., & Kleyn, N. (2012). Corporate identity, corporate branding and corporate reputations. European Journal of Marketing, 46(7/8), 1048–1063. https://doi.org/10.1108/03090561211230197
- [29] Oliver, R. L. (1980). A cognitive model of the antecedents and consequences of satisfaction decisions. Journal of Marketing Research, 17(4), 460–469. https://doi.org/10.2307/3150499
- [30] Ellany, E., & Lateff, A. (2011). Faktor-faktor yang mempengaruhi pembayaran zakat pendapatan di Malaysia. Persidangan Ekonomi Malaysia Ke VI(PERKEM VI)
- [31] Md. Idris, K., & Ayob, A. (2002). Peranan sikap dalam gelagat kepatuhan zakat pendapatan gaji. Analisis.
- [32] Gronroos, C. (1984). A service quality model and its marketing implications. European Journal of Marketing, 18(4), 36–44. https://doi.org/10.1108/EUM000000004784
- [33] Zeithaml, V. A. (1988). Consumer perceptions of price, quality, and value: A means-end model and synthesis of evidence. Journal of Marketing, 52(3), 2–22. https://doi.org/10.1177/002224298805200302
- [34] Parasuraman, A., Zeithaml, V. A., & Berry, L. L. (1985). A conceptual model of service quality and its implications for future research. Journal of Marketing, 49(4), 41–50. https://doi.org/10.2307/1251430
- [35] Samargandi, N., Tajularifin, S. M., Ghani, E. K., Aziz, A. A., & Gunardi, A. (2018). Can disclosure practices and stakeholder management influence zakat payers' trust? A Malaysian evidence. Business and Economic Horizons (BEH), 14(1232-2019-876), 882-89
- [36] Karim, R. A. A. (1990). Standard setting for the financial reporting of religious business organizations: The case of Islamic banks. Accounting and Business Research, 20(80), 299–305. https://doi.org/10.1080/00014788.1990.9728888
- [37] Zabri, M. Z. M., & Mohammed, M. O. (2018). Examining the behavioral intention to participate in a cash waqf-financial cooperativemusharakah mutanaqisah home financing model. Managerial Finance, 44(6), 809–829. <u>https://doi.org/10.1108/MF-05-2017-0189</u>
- [38] Mediawati, E. (2016). Research article internal control and Sharia Supervisory Board role in zakat management organization. International Journal of Recent Advances in Multidisciplinary Research.
- [39] Jayanto, P., Jayanto, P. Y., & Munawaroh, S. (2019). The influences of reputation, financial statement transparency, accountability, religiosity, and trust on interest in paying zakat of profession. Jurnal Dinamika Akuntansi, 11(1), 59–69. <u>https://doi.org/10.15294/jda.v11i1.18729</u>
- [40] Schnackenberg, A. K., & Tomlinson, E. C. (2016). Organizational transparency: A new perspective on managing trust in organizationstakeholder relationships. Journal of Management, 42(7), 1784–1810. <u>https://doi.org/10.1177/0149206314525202</u>

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6312 - Published by University of Turin <u>https://www.ojs.unito.it/index.php/EJIF/index</u> EJIF content is licensed under a Creative Commons Attribution 4.0 International License

- [41] Abojeib, M., Habib, F. 2019. Blockchain for Islamic Social Responsibility Institutions, FinTech as a Disruptive Technology for Financial Institutions. IGI Global, 221-240
- [42] Hamdani, L. (2020). Zakat Blockchain: A Descriptive Qualitative Approach. EkBis: Jurnal Ekonomi dan Bisnis, 4(2), 492-502.
- [43] Fiegen, A.M. (2010), "Systematic review of research methods: the case of business instruction", Reference Services Review, Vol. 38 No. 3, pp. 385-397.
- [44] Khan, K.S., Kunz, R., Kleijnen, J. and Antes, G. (2003), "Five steps to conducting a systematic review", Journal of the Royal Society of Medicine, Vol. 96 No. 3, pp. 118-121.
- [45] Oladimeji Abioye Mustafa, M., Har Sani Mohamad, M. and Akhyar Adnan, M. (2013), "Antecedents of zakat payers' trust in an emerging zakat sector: an exploratory study", Journal of Islamic Accounting and Business Research, Vol. 4 No. 1, pp. 4-25. <u>https://doiorg.ezlib.iium.edu.my/10.1108/17590811311314267</u>
- [46] Saad, R.A.J., Farouk, A.U. and Abdul Kadir, D. (2020), "Business zakat compliance behavioral intention in a developing country", Journal of Islamic Accounting and Business Research, Vol. 11 No. 2, pp. 511-530. https://doi-org.ezlib.iium.edu.my/10.1108/JIABR-03-2018-0036
- [47] Ghani, E.K., Said, J. and Syed Yusuf, S.N. (2012), "Service quality performance measurement tool in Islamic non-profit organisation: an urgent need", International Business and Management, Vol. 5 No. 2, pp. 71-75.
- [48] Hassan, O. and Marston, C. (2010), "Disclosure measurement in the empirical accounting literature: a review article", Financial Reporting and Reporting Special Interest Group Annual Conference, Glasgow.
- [49] Sawmar, A.A. and Mohammed, M.O. (2021), "Enhancing zakat compliance through good governance: a conceptual framework", ISRA International Journal of Islamic Finance, Vol. 13 No. 1, pp. 136-154. https://doi-org.ezlib.iium.edu.my/10.1108/IJIF-10-2018-0116
- [50] Chong, F.H.L. (2021), "Enhancing trust through digital Islamic finance and blockchain technology", Qualitative Research in Financial Markets, Vol. 13 No. 3, pp. 328-341. https://doi.org.ezlib.iium.edu.my/10.1108/QRFM-05-2020-0076

Received: 14/11/2021 - Accepted for publication: 27/03/2022 - Published: 22/04/2022

# Islamic Banks and the European Banking Union: An Overview of Critical Issues and Perspectives

Sandra Antoniazzi1

<sup>1</sup> University of Rome 2 "Tor Vergata" (Italy)

Contact the Author: sandra.antoniazzi@uniroma2.it

Abstract- The aim of this paper is to provide a general, but indepth overview of aspects of incompatibility and actual interest profiles of Islamic finance in relation to the legal system of the European Banking Union, identifying some prospects for positive future developments. It examines the relevant Islamic principles for banking and the differences from conventional European banks for regulation and supervision, a number of experiences and possible relationships. The modern communication between the Islamic banking scheme and the European (and international) one is certainly allowed by the activity of the IFSB regulator, especially as regards the adaptation to the Basel Committee standards and the related capital requirements, transparency criteria and common references, considering the global application to operate in the international markets. The examination of the relationship with the European Banking Union allows the identification of future profiles for debate and research in economics law, considering the suitability for regulatory adjustments and the interaction between EU and national law and the interesting ethical aims of the socially responsible Islamic finance.

Keywords: Islamic finance, Islamic banking, European Banking Union, Regulation, Supervision, Transparency, Certainty of rules, Comparison, Perspectives

# I. INTRODUCTION

In Europe, for several years now, there has been a significant growth of interest in Islamic finance and banking for investment companies, Islamic bond products and other credit collection instruments. Very significant was the position of the Financial Service Authority which, in 2004, issued an authorisation to operate to the Islamic Bank of Britain [1] and to the first European Islamic Investment Bank. Other European countries have experimented with Islamic financial products, albeit in a limited way [2]. In this context, it is interesting to look further at the possible relations between the conventional bank model that operates in the EU market and the Islamic bank that follows criteria, including ethicalreligious ones (e.g., the prohibitions of riba and gharar), but with issue regarding risk control and data transparency, regulation and prudential supervision [3], especially in relation to the current European system, capital requirements and the Basel III Agreement.

The age-old question of the prohibition of interest is also known in conventional banking, in the sense of considering the difficulties of the weak contractor for elementary forms of lending; moreover, the reference to the prohibition of usury and to the forms of protection provided by the national systems that extend to the various forms of traditional operations is inevitable, while for the issue of supervision and risks, the evolution has been very different until the appearance of European banking supervision. From the Islamic perspective, instead, the mechanism of pecuniary interests has been replaced by the "sharing" of the creditor in the profits and risks of the banking activity. In essence, the result of profit can be achieved through sophisticated legal constructions (shareholdings, investment funds, linked contracts) also to encourage credit, given that banks use credit in commercial operations, or with the affirmation of ethical justifying principles linked to necessity or the common good.

In addition, the prohibition of gain (gharar), based on uncertainty for reasons inherent in the value of the asset for speculative purposes or when the asset is not determined for the contracting parties and, therefore, the contract is qualified aleatory, involves an element of risk and uncertainty that is, instead, central to contracts in European (and international) financial markets [4]. A derivative financial product is, of course, an instrument that "derives" its value (price) from that of an underlying financial asset. Derivatives are intended to

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6424 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index

of guarantee regarding the return or the repayment of capital, but a share of profit paid periodically [6]; the other collection instruments, traceable to forms of co-participation or partnership - mudàraba and mushàraka [7], muràbaha [8] - are clearly different from bank deposits. For the well-known principle of the prohibition of interest, there are further particular rules in the Islamic banking model [9]; it is based on precepts of the Shariah: the prohibition of

hedge business risks (e.g., foreign exchange options to hedge

foreign exchange transactions between importers and

exporters), but are currently mainly used for speculation on

future price movements, as the value of derivative contracts

consists of multiples of the actual economic transaction. These financial instruments certainly do not meet the requirements of

linkage to an asset and the character of "non-speculation"; conservative Shariah scholars argue that the prohibition of gharar eliminates speculation on interest, options and some

conventional insurance contracts because they conflict with

the prohibition of uncertainty and wagering (maysir), even if

The main ways of collecting credit consist of current

accounts for the management of payments, for which interest

payable and receivable are foreseen and repayment is only

guaranteed, and in savings deposits for which there is no form

there is a hedging of an underlying economic transaction [5].

[9]; it is based on precepts of the Shariah: the prohibition of investing in activities forbidden by Koranic law and in speculative activities, the sacredness of the contract and the obligation to pay the zakat, which is a religious tax; moreover, these principles also apply in the securities market [10]. Therefore, the model of the Islamic bank, precisely because of its necessary conformity to ethical-religious values, is conditioned in several respects: in investments, the bank must ascertain the credibility of the projects, the entrepreneurial skills of the client and the ethical character of the financed operation with regard to the aims; thus the collection of savings must be ethically consistent and the objective is obviously to attract the interest of savers, even if no interest is paid, through investment funds with participation in the profits according to pre-established conditions.

As we known, the Islamic banking has adopted the Basel III standards, a solution that revealed a significant regulatory challenge, given that the agreement does not consider the specificities of Islamic banks and Islamic products present various profiles of incompatibility with the European (and international) markets [11]. The context is complex and Islamic banks are required to be adequately capitalised to inspire trust and maintain sufficient liquidity under Basel III, like European banks, in order to protect depositors and avoid expensive governments "bail-outs" in the case of defaults. Besides an increasing number of Islamic funds screen equity investments to ensure they are permitted and there are even indices tracking Shariah compliant stock, most notably the S&P Dow Jones Islamic Indices.

This framework suggests a reflection on the sector in relation to the European Banking Union, in order to identify conflicting legal profiles, and to highlight the need for specific European and national rules and compatibility aspects especially considering sustainable finance. The aim of the research is to outline legal considerations regarding the prospects for the establishment of Islamic banks and the use of financial products; the results that emerge may provide a reference for future research.

### II. LITERATURE REVIEW

The article adopts an interdisciplinary analysis between law and economics, in the sense that the legal reflection cannot disregard decisive economics studies. The literature cited in the "References" section shows jurists have not confirmed a particular interest, beyond the episodic in the context of conferences, especially in the Italian legal system or the initiatives of financial institutions (ECB, Bank of Italy). The results of economic studies make it possible to acquire an adequate awareness of the affirmation of Islamic finance in the international markets and of the reasons for its necessary consideration in relation to the European legal systems.

It should also be pointed out that the role of lawyers depends on the existence and interpretation of disciplines or at least the adoption of bills; as is well known, there are not European rules for Islamic banking, but individual experiences of some European countries. From the bibliographical references emerge a clear minority of law studies and an extensive panorama of in-depth studies by Italian and foreign economists who analyse the relevant presence in international market of Islamic finance from several points of view.

Predieri was the first Italian jurist to deal with Islamic banks in essays in 1996 and 2006 with the broader aim of fostering knowledge of Islamic law and contributing to cultural enrichment; Piccinelli also follows this line, focusing on legal instruments. Since 2006, some authors (Donato, Donini, Freni, Gimigliano, Miraglia, Piccinelli, Porzio, Rotondo, Russo, Santoro, Scolart, Casoria) have explored the topic from an interdisciplinary perspective, above all, in several collective volumes, often containing conference proceedings that highlight the intention to solicit further studies. However, the debate has not had much follow-up among Italian jurists, while there has been more interest in other legal systems. There are also some contributions from foreign lawyers (Aldohni, Al-Rimawi, Ariff, Arnaud, Decock, Engels, Farook, Housby, Lewis, Sagaert, Wilson, Knan, et al.) dealing with the differences between conventional and Islamic banking systems, also with reference to the European Union and the experience of the United Kingdom.

Undoubtedly, the most numerous and specific studies consist of economics studies by many foreign (Abbas, Abiad, Ahmed, Aracil, Ariff, Asad, Ashraf, Asutay, Chazi, Ginena, Gintzburger, Hassan, Khallaf, Khavarinezhad, Jafari-Sadeghi, Khan, Zamir *et al.*) and Italian authors (Biancone, Calandra, Colecchia, Floreani, Miglietta, Paltrinieri, Prandi, Lanzara *et al.*) who addressed various aspects of Islamic finance: the response to the crisis, capital requirements and the application of Basel III, risk exposure of GCC banks and activities in the European context as well as sustainable finance profiles.

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6424 - Published by University of Turin https://www.ojs.unito.it/index.php/EJIF/index

In this survey, the law and economics studies, reports and publications of the Islamic Financial Services Board (IFSB) [12] represent crucial references for an overall knowledge of the sector and regulation aimed not only at Shariah compliance, but also at aligning Islamic banks with international standards acquired by the European Banking Union and the Single Supervisory Mechanism.

This article could help fill a gap in law studies on possible relationships between the European system and Islamic banking industry.

# III. METHODOLOGY

The method of investigation adopted consists in comparing the typical legal principles and the economic aspects of Islamic finance and the European Banking Union and national law, in order to verify whether the establishment of Islamic banks and investment companies is conceivable or whether, for reasons of structural incompatibility, the current European regulations and organisation can only admit the offer of Islamic finance products by conventional European banks.

More precisely, the method of comparative law [13] allows a comparison of the relevant aspects of the two sectors of banking activity that belong to different law systems, examining the European discipline and the principles of Islamic finance in order to identify possible areas of "coexistence" in the same system. The survey draws on the contribution of the extensive economics literature cited in the "References" section which has supported the study of legal profiles, in the light of the modern interdisciplinary research that is indispensable in such complex areas. In fact, for the field of Islamic finance, economic studies are prevalent compared to the limited interest of European jurists.

The difficulty of finding legal scholarship conditions the debate, which focuses mainly on the economic consequences and less on the construction of legal instruments for operational adaptation in the European banking system. In fact, lawyers tend to examine disciplines that already exist in the system of reference, the possible changes or bills; in this case there are no specific rules of European law, but there are general principles and individual experiences of some European countries that have allowed the use, within certain limits, of Islamic financial products.

Economic studies, on the other hand, can evolve by examining global economic phenomena using case study criteria and empirical analysis of the activities of operators according to international standards. The approach is more open, and the object of the research concerns the economic consequences of existing legal schemes, which can provide an interesting basis for legal reflection, especially when a sector, such as Islamic finance, is so widespread that it cannot be ignored by European legal systems.

# IV. FINDINGS AND DISCUSSION

# A. The comparison with the European conventional banking system and differences

Relations with conventional banks are difficult because of the characteristics of Islamic finance and banking activities exercised in the EU context according to different rules for the financial intermediary, above all, considering the management of the risk, to the rules of vigilance and to the regulation of competition. For the intermediary from the Islamic discipline, the implementation of transparency depends essentially on the respect of conduct provided by the supervisory authorities and by the Shariah guarantee committee for an ethical control, while the informative obligations depend on the discretion of the bank with inevitable problems of transparency and certainty of the rules. However, the IFSB has contributed significantly to improving relations with the conventional banking system. A central aspect that must be examined is the compatibility of the Islamic banking activity with the European supervision discipline, on the requirements of liquidity of the capital and on the financial system in order to envisage a framework of admissible relations.

In general, as we have seen, the principle of participation which is linked to the sharing of the risk characterises Islamic banking activity and because of its structure it is not linked to an obligation of guarantee for the capital deposited and funds reimbursement, a substantial aspect which is not compatible with European banking discipline and the requirements of capital liquidity, especially after the innovations of the Banking Union. However, it is necessary to distinguish between the regime of free and participating deposits and, in fact, only the latter can be affected by losses, if the bank is not able to meet them with its capital reserves. Moreover, it is necessary to point out that the location of the bank, be it Islamic or conventional, conditions the applicable discipline in the relative order; the purposes of the bank deposit are different: in the Islamic system, the capitalsation of savings and the guarantee of preservation, while in conventional banks the deposit has represented, at least until the financial crisis and the drastic reduction of interest rates, also a solution of investment and short-term savings [14].

An immediate comparison [15] between the two systems shows the different nature of the discipline: since for conventional banks the functions and operating procedures are provided for by European and national laws and standard rules, while for Islamic banks the reference is the Shariah to which the production of profits must conform, an objective common to the conventional bank, while the main function is different, which in the first case consists in participation in partnership activities and in the second in the intermediation of funds. The relationship with clients is a three-way one, i.e., between investors, partners and intermediaries, which is different from the "traditional" one based on the creditor/debtor dynamic; as for interest, this is not applicable as it is contrary to the rule of risk sharing between depositors and custodians of funds, unlike our system which provides for active and passive interest rates as a return and costs to be

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6424 - Published by University of Turin <u>https://www.ojs.unito.it/index.php/EJIF/index</u> EJIF content is licensed under a Creative Commons Attribution 4.0 International License 39

incurred for the deposit. The approach is also different for the phase of verification of admission to projects financing, since the Islamic bank gives importance to the feasibility, while the conventional one gives priority to the reliability of the applicants and financing based on the provision of guarantees. In the event of defaults or delays, the Islamic bank cannot set interest or other charges, a rule that is, instead, general in the conventional scheme, and neither would there be an obligation to fully guarantee deposits, an issue that has been the focus of attention in the EU and in the choices of the ECB until the introduction of a single mechanism for the resolution of banking crises (Reg. (UE) 15 July 2014, n. 806; Directive (UE) 15 May 2014, n. 59).

In any case, the obstacle of the prohibition of interest has prevented easy access to the interbank market and investment in "conventional" securities precisely because of their intrinsic nature of referring to interest-based return mechanisms. However, specific forms of investment and credit collection have been developed for issuance, consisting of transactions. especially in mutual funds, in participatory corporate contracts and in the issuance of bonds and Sukuk, which are bonds deriving from the securitisation of portfolios of contracts, and in participatory current accounts. Islamic bonds are issued by the state or other public entities according to the scheme of securitisation of public assets through sale and lease Ijarah. In the mutual fund, the holders of the shares entrust the management of a capital to a specialised operator who is entitled to a fixed amount of the profits deriving from the management and the investor is entitled to a share in profits and losses proportional to the amount conferred, based on the trust agreement between bank and depositor. The management of the fund can also take place through a company, which is responsible for the correct administration of the investments in compliance with the Islamic discipline and the directives of the body (committee or council) in charge of monitoring compliance with the ethical rules and for any financial losses from which an obligation of compensation generally derives [16].

# B. Islamic Banking and European Banking Union

The Islamic bond market has gained significant consideration even among conventional investors as well in the last decade precisely because these instruments are linked to material assets and, therefore, with a high degree of security [17]. Not only Muslim investors are involved, as they formally conform to Islamic principles, but also European ones, given the financial crisis caused mainly by the circulation of instruments that have unsafe mechanisms, based as they are on interest and debt. Indeed, these securities are currently traded in the largest secondary markets mainly because of their security qualities.

The growth of Islamic banking can certainly be traced back to the spread of investment funds and Islamic bonds; the establishment of the Islamic Bank of Britain [18], which is an expression of this evolution also by means of innovative management methods in order to acquire credibility in the markets, notwithstanding its very particular and distinct framework compared to conventional banks. Regarding taxes, the legislation in the UK facilitated the practical implementation of traditional models of acquisition avoiding interest, which are compatible with Islamic law. New rules have been implemented to levy the stamp duty only once in cases of Islamic mortgages and at first HSBC Amanah first offered such Islamic mortgages, with other banks following suit.

However, it is necessary to highlight the problematic issues of compatibility with the conventional market, regulation in the classical sense and the necessary transparency of requirements, when Islamic banks operate in European countries. It is necessary to consider the supervision exercised by the ECB and national central banks, according to the approach of European banking supervision (Reg. (UE) 15 October 2013, n. 1024 and Reg. (UE) 16 April 2014, n. 468 and related disciplines) and many rules (Basel III, EBA) that impose obligations of capital requirements and procedures of necessary transparency (e.g., the "stress test" for banks) about aspects of accounting and internal decisions that can only be based on the adoption of technological innovations for communications between the operators involved. Islamic finance schemes present interesting aspects from the perspective of a sustainable and socially oriented finance, but the critical issue is the certainty of the rules and their effective knowability through widely available technological tools.

It is, therefore, necessary to have a high level of transparency in internal controls by the Council of Shariah, which plays a role analogous to the ethics committees, but according to Islamic ethical-legal rules from which derives a certain unpredictability in the application of rules of conduct by way of interpretation. Moreover, a certain uniformity of rules would be necessary, given the obvious differentiation according to the countries of origin, an objective in theory constantly affirmed by the GCC, but difficult to implement even for the discipline of competition [19].

The Islamic bank is, however, for certain activities, certainly comparable to a conventional credit institution, but for further services of a financial nature, problems of vigilance arise and for the collection of credit "at risk", there is a need for greater transparency also for the savers, especially for European investors. From the mudàraba and mushàraka schemes derives a significant involvement in the business risk of the bank and, therefore, adequate supervisory measures are necessary. In the case of musharaka, the partnership contract between bank and client foresees that the capital in goods or liquidity derives from both and allows the control to actively participate in the project [20]. Moreover, the need for an organisational separation of the activities of the bank is also envisaged for the Islamic bank, in order to avoid possible conflicts of interest between the parties involved in the business, like the universal bank scheme [21].

### C. Tasks of Islamic financial and banking authorities

The Islamic financial activity has, therefore, objectives of an economic nature (preservation of the capital, profits,

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6424 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index

EJIF content is licensed under a Creative Commons Attribution 4.0 International License

equilibrium between liquidity and profitability) and ethicalreligious rules (respect for the precepts and standards of behaviour of the Koran and of the Shariah, considering the legitimate goods - halal - and excluding the forbidden haram activities). Moreover, the concept of cooperation replaces the traditional concept of competition in the market mechanism, which is instead central in the EU.

The difficulty of "translating" ethical-religious or, at least, moral principles into rules of prudential supervision, which in the international market must be compatible with the requirements of Basel III, is evident, in addition to the uniform European framework for Member States. Furthermore, there is the necessary availability of information that can be acquired by financial intermediaries and investors, a condition favoured by technological innovation and tools in the most advanced countries along with the establishment of competent authorities.

The schemes of Islamic finance present characteristics that seem, in effect, to favour economic and social sustainability; however, the risk of unpredictable interpretations adopted by the Shariah supervisory committees is still an element of significant perplexity, especially if we consider the essentiality of legal certainty and reliability of the financial products, after various crises for different reasons. The current solution represents a delicate compromise between the acceptance of international standards and the objective of Shariahcompliance activities.

The national Islamic banking supervisory authorities must guarantee the stability of the financial system according to two objectives: the control of capital requirements with international indications and the modalities of activity that must be Shariah-compliant. This stability depends on ensuring that Islamic banks have systems in place to control requirements and internal Shariah governance policies and processes to identify, monitor and control or mitigate Shariah risk as a result of interpretations that may be inconsistent. As regards the relationship between liquidity and credit risk, some studies have compared Islamic banks and conventional ones and revealed that the former are better than the latter in managing risks [22]. In addition, a further objective of financial regulation is the protection of consumer rights that may be affected by the proposed instruments, and it is essential that Islamic banks ensure fairness and diligence in their operations and compliance with Shariah precepts, as depositors are attracted to these institutions primarily for their ethical and religious principles.

Thus, the IFSB advises supervisors to provide banks with a framework of uniform internal rules to enable Shariah compliance, although supervisors will inevitably have differing views on Shariah governance given the different legal systems in which the banks operate. In addition, the IFSB ascertains the different needs of individual banks, which vary depending on different factors such as size and capital requirements; regulators should ascertain whether banks have control systems in place to ensure Shariah compliance. To this end, the IFSB has developed some guiding principles on

requirements for the distribution of Islamic products in the capital market, for Sukuk and collective investment schemes.

The focus issue is the implementation of good practice in financial reporting and the establishment of an independent auditing systems and the task for any supervisory body can be made more manageable by learning from the experience of others and by applying available standards and guidelines. In this sense, the Basel III rules have contributed significantly. At the same time, it is possible to say that Islamic finance schemes may contribute generally to spreading an economic approach of solidarity.

# D. The relationship of the GCC countries with the ECB and some limited experiences of Islamic finance in EU Member States

From 2005 onwards, it is possible to identify frequent contacts between the EU and the GCC, managed by the General Secretariat [23], and requests for specific advice to the ECB on the single currency project, which was planned for 2010, highlighting the European experience as a model to be followed for the establishment of a centralised banking entity, price stability, and fiscal and monetary convergence parameters [24]. However, links and political relations between the GCC and the EU date back to 1988 and a cooperation agreement, which provided for advice, mutual dispatch of delegations and annual meetings of ministers in sectors such as energy, transport, research and innovation, and the EU-GCC Clean Energy Network and the EU-GCC Trade and Business Cooperation Facility. At present, the objective of the single currency is postponed or at least suspended, given that there are no recent definitive decisions in this direction, to which have to be added the problems of agreement between the various countries, especially for the international policy guidelines that have created diplomatic difficulties.

Decisive decisions are still being awaited on monetary union, while for the financial markets the European discipline has already influenced regulation well beyond its natural boundaries of application due to the need for necessary harmonisation in order to encourage investment, despite the critical aspects. In fact, in the adopting standards, the IFSB, which includes GCC banks, has inevitably further rooted the European schemes as an experience of best practice in conventional markets, initially considering the Directive 2003/71/EC (on trading in financial instruments), the Transparency Issuers' Information of Directive (2004/109/EC), the UCITS Directive 2009/65/EC (regulation of undertakings for collective investment in transferable securities) and, subsequently, the most recent regulations in force, as well as the criteria adopted by the Basel Committee for European and international standards.

The cooperation of the ECB with non-EU countries has therefore also involved GCC countries, as it has strengthened relations with the General Secretariat, central banks and monetary bodies of the Member States since 2005, particularly with a view to introducing a single currency. Instruments have also been developed to enhance cooperation with the major

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6424 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index EJIF content is licensed under a Creative Commons Attribution 4.0 International License central banks of many non-European states, including those of the GCC and their monetary authorities, through specific seminars [25], technical assistance and guidance, including through the contribution of expertise in the conduct of monetary policy, payment systems and banking supervision, as a "source of inspiration" for various systems. This advisory activity is part of the institutional tasks as provided for by Article 5 of the Statute of the ESCB-ECB (Protocol no. 4); for the collection of statistical information, the ECB "shall cooperate with the institutions, bodies, offices or agencies of the Union and with the competent authorities of the Member States or third countries and with international organisations" (Article 5.1.) while Article 6 refers to international cooperation and participation in international monetary institutions.

The ECB cultivates international relations [26] for cooperation with countries outside the EU; in particular, the Article 8 of Reg. (EU), 15 October 2013, no. 1024, on banking supervision, provides for the possibility for the ECB "to establish contacts and enter into administrative arrangements with supervisory authorities, international organisations and administrations of third countries, subject to appropriate coordination with the EBA. These agreements do not create legal obligations for the Union and the Member States". This provision, which also allows for forms of cooperation in prudential supervision, may help to expand relations with other systems, especially in the exchange of information and the adaptation to common international standards.

In addition, a regulatory convergence between the EU, the GCC and other Islamic countries is necessary for the establishment of Islamic banks in Europe regarding the prerequisites for banking authorisation (Articles 4.1 (a) and 14, Reg. (UE), no. 1024/2013), capital requirements, safety nets, independence and powers of supervisors. In fact, EU law and national legislation provide for rules on administrative and decision-making bodies in a way that is far removed from the presence of the Shariah board.

Islamic financial instruments have been offered on the market for some time; in 2004, in the United Kingdom the Islamic Bank of England obtained its authorisation, in 2006 Sukuk was listed on the stock exchange and in 2014 saw the widespread issuance of Islamic bonds with a specific discipline, as in Luxembourg; while in 2004, Germany, the Land of Saxony-Anhalt decided on the issuance of Sukuk as in France [27].

In Europe, the financial sector considers the needs of Muslims with some solutions, apart from the older case of the United Kingdom, now no longer part of the European legal framework, due to the significant presence of a very traditionally oriented Muslim population. In Germany, there is a significant experience: the Kuveyt Türk Beteiligungsbank (Frankfurt) which obtained a licence as a deposit and investment bank in 2015 after an experience that began in 2004 [28], through a German bank that allowed the knowledge of Islamic finance, along with some national regulatory innovations. Other jurisdictions have introduced a special tax regime for Islamic investors for specific products.

Stock exchange values are currently listed in the Dow Jones Islamic Market and FTSE Global Islamic Indexes. European banks already distribute shares in Islamic equity funds for financial investments that exclude companies involved in gambling, alcohol, tobacco or interest-bearing loans.

The development of Islamic finance has depended on its significant reliability and resistance to financial crises (2007/2008 financial crisis period) due to its low indebtedness and use of operational instruments that are based on real assets. Certainly, financial products are more expensive as sharing entails control costs, problems of information asymmetry and property management. For an effective contribution to sustainable finance, a common regulatory and certainty, as well as contracts with clear content, especially considering the link between law and moral principles in order to limit legal risk as much as possible.

However, there are critical views regarding the tendency of Islamic financial institutions to offer products and services that are in essence similar those offered by conventional institutions and to try to come up with new ideas to remain competitive as a viable alternative to conventional finance; at the same time Islamic financial experts must ensure their Islamic character [29].

Financial transactions are guaranteed by assets and risksharing and transactions have real purposes (concrete and productive activities, real estate), limiting financial speculation; moreover, some recent reports [30] on Islamic finance reveal prospects for growth, despite the difficulties caused by the pandemic [31]. These aspects are of considerable interest to European legal systems as an opportunity and alternative source of financing. In Europe, there is a significant presence of "Islamic counters" that have opened in conventional banks for depositors wishing to implement religious principles, but regulatory innovations must be introduced for supervision and regulation, in the sense that the EU member states where these banks are located should adopt their legislation considering all the potential risks in relation to this new financial activity.

These perspectives also appeal to the Italian system, on the assumption of appropriate regulatory changes necessary to overcome tax issues and incompatibilities with internal rules, in order to encourage the establishment of Islamic banks, especially in the form of investment banks, while European advisory companies are already widespread. In 2010, a seminar, organised at the Bank of Italy, brought together GCC central banks and monetary authorities to examine financial and economic developments in these countries and in the euro area and the consequences of the global financial crisis for stability and monetary policy. A relatively favourable economic experience emerged, especially in comparison with advanced economies and emerging markets. The GCC Region was seen as a major player in international financial markets, engaged in the efforts of the international community to reform the global framework for financial stability and banking supervision [32].

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6424 - Published by University of Turin https://www.ojs.unito.it/index.php/EJIF/index

Later, in 2013 at the IFSB Forum in Rome at the Bank of Italy, the Governor Ignazio Visco focused attention on the interest for Islamic finance and the opportunity to attract foreign capital to underpin economic progress and the intensity of the commercial and financial links with the southern shores of the Mediterranean and make it increasingly important for Italy and its financial system to be equipped with the knowledge and the operational instruments needed to interact with financial systems complying with shariah principles: the principle that profits should be generated from fully sharing in the business risk of an investment, the socalled "profit and loss sharing principle". The asset-banking requirement complements these prescriptions, providing for the link between each financial transaction and an identifiable underlying asset, but it has to be clearly ethical and certainly transparent.

The offer of Islamic financial products is characterised by high growth rates and the most relevant component is banking, but in non-Islamic systems an adaptation of traditional and fiscal schemes is required. The Italian legal system shows a limited and slow interest in these alternative financial instruments, compared to other European countries. Bill no. 4453 of 2 May 2017, presents provisions concerning the fiscal treatment of the operations of Islamic finance, given that a decisive obstacle to the expansion of Islamic finance in Italy consists in the "duplication" of taxation because of the complex structure of the operations to ensure compliance with the Shariah, with an aggravation of fiscal costs in comparison with the instruments of conventional finance [33].

The objective is to "assure a taxation equal to that of the conventional financial operations" (Article 1), in order to attract foreign investments and depositors from the Muslim community in Italy who already hold a high amount of savings. The "constant monitoring" will certainly have to be extended and supplemented by specific rules of regulation and supervision to be included in the current system.

At present, the project is still assigned to the VI Finance Commission, but there have been no further developments. The slowness of the Italian legislator has led some banking operators to consider the possibility of offering Shariahcompliant products through "Islamic windows", i.e., special branches created in conventional banks. Regulation appears to be a very important issue, in order not to exclude from access to credit a sector of the population corresponding to at least one million residents, both for social reasons and for evaluations of credit support and incentives for investment funds.

# E. The role of the IFSB

The context of reform of the international rules and global difficulties have undoubtedly aggravated the relevant complexity of the multiple tasks carried out by the IFSB as regulator and supervisor of the Islamic banks in the system of natural belonging and as an authority that adopts the choices of necessary alignment to the conventional system for obvious interests in international markets. This organisation promotes

and enhances the soundness and stability of the Islamic financial services industry by issuing global prudential standards and guiding principles, broadly defined to include banking, capital markets and insurance sectors.

The respect of punctual constraints of capital is a further controversial aspect, given that for the Islamic banks the prevailing limits consist in ethical rules of the Shariah, in addition to the evaluations of control of merit which depend on the interpretation of the Shariah, exercised by a council (or committee) of the Shariah, and limits for the liabilities in the balance sheet. However, in addition to these forms of ethicalreligious control, there is also the regulation of Islamic banks for the aspects that are linked to risk and capital requirements. In this context, the IFSB has taken on a decisive role, established in 2002 and operational since 2003, precisely because of the particularity of the requirements of financial institutions, which involves several banks in the Arab countries of the GCC and which provides international standards and obligations comparable to those provided by Basel III and the discipline on banking supervision. Indeed, the IFSB has published standards, guiding principles and technical notes covering various regulatory aspects for Islamic banking markets, insurance markets and financial markets.

The IFSB is an international organisation aimed at introducing standards, promoting and enhancing the soundness and stability of the Islamic financial services industry by issuing prudential rules and guiding principles for the financial services industry (banking, capital markets and insurance sectors). The standards developed derive from a complex process specified in guidelines and standard-setting procedures, which are characterised by the issuance of exposure drafts, the conduct of workshops and public hearings. In addition, the IFSB prepares very concise annual reports, conducts research and coordinates initiatives, roundtables, seminars and conferences for regulators and industry participants and for these purposes it cooperates with international, regional and national organisations, research institutes and market participants, as a "dialogue" entity with the conventional banking system.

The IFSB standards follow a comprehensive due process as outlined in its *Guidelines and Procedures for the Preparation of Standards/Guidelines*; it involves the issuance of exposure drafts and carries out workshops and public hearings, research, conferences for regulators and stakeholders [34]. These activities are carried out in cooperation with relevant international and national organisations, research entities and market players.

The Islamic Financial Services Industry Stability Report 2021 [35] offers a selection of results from the IFSB international research and surveys to achieve a broader and common understanding of critical issues in Islamic finance. In line with governments and regulators in other countries engaged in economic recovery and the adoption of fiscal and monetary measures, the IFSB has been engaged in analysing the most effective interventions to support the banking sector and sustained its growth momentum in 2020 recording a growth rate of 10.7% year-on-year based on significant

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6424 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index

improvement, especially in the Islamic banking and capital markets segments. The financial stability indicators remained satisfactory, especially when compared with previous years' performance, conventional peers, and assessment criteria used by international standard-setting bodies.

The IFSB closely monitors developments in the global financial system generally and specifically in the jurisdictions of its members; it has published a compendium of policy responses and a paper on the impact of the pandemic on the Islamic Financial Services Industry (IFSI) and several standards, guides and technical notes to complement the work of other international standards but also to cater for the specific tasks of the IFSB. The analysis conducted on the Islamic banking segment is based on the Prudential and Structural Islamic Financial Indicators (PSIFI) database of the IFSB, which has extended coverage of the database to both the Islamic capital market and takaful segments from 2020 onwards. The report examines the implications for the soundness and resilience of the global IFSI and recent economic developments and changes in the global financial system, with a particular focus on the pandemic.

The 2021 Report presents an overview of the vulnerability aspects and the outlook of the global Islamic financial services industries and IFSB members regarding the three key sectors: Islamic banking, the Islamic capital market and Islamic insurance that are objects of an in-depth analysis based on data from the Prudential and Structural Islamic Financial Indicators database of the IFSB. This document provides an indicative outlook for the IFSI, which makes it a prime reference for information on the stability of Islamic finance globally and across national systems.

The developments and significant economic impacts depend on the widespread public health measures for the global pandemic, an external shock that did not affect the banking sector. However, the economic and social impacts have affected the stability of financial systems. The Islamic social finance sector can promote stability by diversifying the financial system and indirectly by financial and social inclusion through providing inclusive financial services for poorer people. This financial sector could complement the commercial financial sector through the zakah and waqf that can provide stable funds to tackle the adverse impact of negative shocks such as the pandemic. The zakah is a percentage of wealth a waqf represents in social capital that is relatively stable during negative shocks, and they provide support to people in need [36].

Covid-19 has impeded the achievement of many goals, but it has also highlighted the need to make significant changes. According to the report, in this regard, Islamic social finance needs to transform to become more effective and resilient in order to help promote the development of sustainable and stable economic, social, environmental and financial systems. As in the European system, economic support measures and macroprudential policy measures [37] helped to improve financial conditions and to mitigate any potential financial stability risks that remained relatively resilient with significant differences in several countries. Reading the report reveals the Islamic banking and Islamic capital markets sectors contributed to the increase in total worth of the global IFSI, while takaful sector experienced a slight contraction; otherwise, the IFSI demonstrated overall growth in 2020 despite many systems experiencing economic recessions as a result of lockdowns and heightened uncertainty due to the pandemic [38]. For example, the GCC countries maintained their position as the most important market for Islamic finance activities in 2020 and the Middle East and South Asia region is the second largest share remaining consistent with the previous year.

The data emerging from the report do not seem negative, if compared to the European experience and if we consider the level of transparency adequate; however, adjustments are also needed to participate in the European market. The countries examined, although they share an Islamic approach to the economy, also have very different situations in the financial sectors, due in part to differences in internal regulations, profitability levels and the market for available energy resources, as the Report clearly shows.

For a useful comparison of compatibility with the European system, it is therefore necessary to examine the commitment to the Basel III standards. The Report considers the Basel Committee programme and strategic priorities for 2021-2022 that focused on implementation and evaluation; it will complete existing work on the prudential treatment of cryptocurrency exposures about which a consultation document was published in June 2021 and on disclosure requirements related to market risk and sovereign exposures. In particular, the Report specifies that disclosure requirements may have some impact on future IFSB standards, confirming the direct influence on Islamic banking system.

In 2020-2021, international entities in the banking, securities and insurance sectors issued several documents (reports, consultations, standards, survey reports) that are relevant to the tasks of the IFSB, and show global developments and the impacts of new choices also as a result of the pandemic. The IFSB, "being the complementary global standard setter" for the Islamic banking, capital markets and takaful sectors, monitors standards and guidelines of these global authorities (Basel Committee on Banking Supervision, International Organisation of Securities Commissions, Financial Stability Board, International Association of Insurance Supervisors). Their activities and standards have been seriously affected by the pandemic; in fact, some entities have done significant short-term work in response to it, for example, by providing members with access to information on measures taken in other jurisdictions.

Some postponements of the implementation dates were decided for agreed reforms in order to allow institutions to focus more fully on their responses to the pandemic crisis, for example, the consultation on new Principles for Operational Resilience and revised Principles for the Sound Management of Operational Risk of the BCBS. The IFSB also adopted a consultation on Effective Practices for Cyber incident Response and Recovery, and IOSCO consulted on revisions to its Principles for Outsourcing with operational resilience as a

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6424 - Published by University of Turin https://www.ojs.unito.it/index.php/EJIF/index

driver of several changes. In these documents, there are no references to the specifically Islamic financial and banking regulations, but their influence on the work of the ISFB will emerge in future standards on broader aspects such as risk management or the supervisory review process.

Other topics for specific definitive standards relate to the impact of financial technologies, particularly crypto-assets and "stable coins", but also include the use of artificial intelligence and machine learning by asset managers and other capital markets intermediaries and the use of big data analytics in insurance [39].

The Islamic financial system has shown resilience, and unprecedented interventions by governments and regulators have allowed markets to continue in their tasks [40]. The IFSB has identified a programme focusing particularly on fund liquidity and non-bank financial institutions based on analytical study and a number of changes in standards. In fact, Islamic capital markets have some very positive profiles such as the very limited use of derivatives and the adoption of social projects of financing, while in the Sukuk market a shortage of liquidity has become apparent.

# F. The regulatory criteria, standards and supervision of Islamic banks: the needs for transparency and knowability through technological innovation

The implementation of IFSB standards represents a challenge in some jurisdictions that have a limited experience, because they should have adequate infrastructure to support Islamic finance and a skilled policy team. The guidelines represent an opportunity to improve implementation rates by leveraging on the experience of other members and sharing the basic steps of regulatory and supervisory authorities that have successfully implemented IFSB standards [41].

Most of the regulatory and operational aspects of Islamic banking depend on the respective central banks which develop the legislative framework for competition within the Islamic financial sector; the legislations also contain provisions on governance and provide for Shariah compliance committees or boards for financial institutions, present e.g., in all GCC countries [42]. The main role of the Shariah supervisory boards is to ensure that the instruments offered by financial institutions are coherent with the fiqh (jurisprudence) on ethical-religious rules as a form of supervision and governance with additional tasks, e.g., advising the board of directors, providing guidance to Islamic financial institutions on the limits of activity, without executive functions that fall exclusively to the management of the bank or a strategic role for the board of directors. The Shariah board has no competence regarding the management of the risks, apart from those deriving from its own decisions or from the fatwa [43] which is a legal opinion on any Shariah question, according to a predetermined methodology. In fact, the board establishes rules and principles relative to Shariah, ensures that the policies and procedures of the Islamic financial institutions are compliant and can take decisions to create certainty of compliance. Together with the board of directors and other

operational committees, the Shariah committee can prevent aggressive lending and higher-risk activities, counter unethical practices, and ensure compliance and principles for social justice purposes, although there may be changes of direction in decisions [44]. In this context, Shariah-compliant financing is a contributing factor to the efficiency of the GCC national banks and the functioning of the overall financial sector.

While several banking laws regarding Islamic finance require banks to have a Shariah board in-house, most regulators are reluctant to engage in Shariah matters and this trend is controversial. Malaysia's regulation [45] is an exception, as the central bank has its own Shariah board, whose fatwa may be of a higher level than the Shariah boards of individual Islamic banks for internal control [46]. This ensures consistency and probably a reduction in risk, but it could potentially limit debate in the future by centralising the decision on compliance. Within the corporate governance system, therefore, adequacy to Shariah is essential, not least to help mitigate the effects of potential risks through due diligence respecting the ethical underpinnings of Islamic economics. However, from recent empirical evidence it emerges that the best practices of Shariah governance have not achieved the efficiency envisaged by the banks, which pursue objectives of governance of the activities according to the ethical Islamic economy [47] and the prevention of financial crises.

Compliant financing has demonstrated positive effects, but improving the quality of institutions, the rule of law, and the effectiveness of government action have emerged as equally important aspects as part of strengthening lending by banks in GCC countries. The main policy implication is that governments and policymakers will have to adapt legislation more vigorously and adopt reforms of laws and regulations governing the banking and financial sector and a more open economy. GCC countries have, in fact, long been in the process of continuously reforming their business sectors to make them more competitive internationally, but attention to strengthening the institutional framework in which finance operates is essential [48].

However, the issue of securities regulation cannot be addressed independently of an examination of the legal context, especially since there are no abstract schemes developed in advanced regulatory regimes that can be adequately introduced into systems with different assumptions and ethical-religious limitations that are a source of legal risks inherent in the application of Shariah and constitutive effects for commercial legislation, such as the prohibitions of riba and gharar that limit the modus operandi of stock markets and the choice of securities [49].

The Shariah produces legal and economic risks that are not quantifiable a priori and that can directly influence modern Arab commercial legislation and, in the worst-case scenario, could render the new discipline inapplicable, given the exercise of supervision according to ethical-religious criteria. In essence, the influence of Shariah cannot be fully foreseen in the examination of commercial legislation, as assessments are based on criteria that are subject to interpretation that is often

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6424 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index

not unambiguous and not even accessible through instruments of certain knowability. The need for transparency is particularly evident in order to reduce risks as they can be known and foreseen in advance [50].

## G. The rules on capital requirements

The IFSB institutional website has been enriched with content, and transparency has undoubtedly increased in recent years; however, the difficulty of framing ethical-religious principles in uniformly applicable rules and standards of banking regulation is evident.

The objectives of the guidelines are very similar to those pursued by the EBA and ESMA and, therefore, to the indication of technical standards as a basis for national regulatory and supervisory authorities to define, in turn, specific rules and guidelines on adequacy requirements for the marketing of financial products and investment funds [51], the dissemination of information in the sector and, in particular, greater harmonisation of regulation and practices, in order to encourage investments from third countries. In addition, they include the competence to prepare stress tests [52], similar to the EBA. Among the guiding principles in the offer of Sukuk securities and investment funds, inspired by good practices in conventional and Islamic markets, are the rules of clear, accurate and not misleading information to investors as well as timely disclosure for a reasonable investment decision; in particular, for Sukuk securities, the structure and related contracts must be described with sufficient clarity to allow the investor to understand and assess any risks attached to it [53] and there are similar rules for investment funds adopted by national regulators [54].

The regulatory standards and guidelines for Islamic banks can be divided into two basic categories: the first set of standards relate to Islamic counterparties and consist of guidelines and conventional international standards that include the prudential standards necessary to ensure a stable financial system; in this development the IFSB uses existing international standards issued by relevant bodies, such as the Basel Committee, as the basis for its own standards with adaptations to suit Islamic banking practices. For example, the Capital Adequacy Standard for Institutions (other than Insurance Institutions) Offering Only Islamic Financial Services (2005) is based primarily on the principles of Pillar 1 of the Basel II standards. In particular, the IFSB amended and adapted two Basel Committee documents to develop appropriate capital requirements for Islamic banking practices. Following the financial crisis, IFSB considered the new guidance of Basel III by adopting Revised Capital Adequacy Standard for Institutions Offering Islamic Financial Services (IFSB-15) incorporating the changes in capital adequacy requirements of the new international standards [55]. Thus, the Basel Committee rules are acquired indirectly through the intervention of the IFSB which adapts them to the reality of Islamic banking. The second type of IFSB standard rules is unique to Islamic financial activity and consists of guidelines for an adequate supervisory framework in the application of

Shariah for Islamic financial institutions also through digital solutions [56].

The IFSB, in its publication Guiding Principles of Risk Management for Institutions (other than Insurance Institutions), has defined the various categories of risk (e.g., credit, investment, market, liquidity and operational risk) and clarified that Islamic financial institutions should be characterised by a comprehensive risk management process and transparency and reporting requirements. The process should consider adequate measures to comply with the rules and principles of the Shariah and ensure adequate risk reporting to the supervisory authority; among the forms of risk, credit risk: the Islamic Financial Institutions (IFIs) should have a specific strategy of financing and the instruments used must be in conformity with the Shariah, which recognises the potential credit exposures that can arise in different phases of the various financing agreements [57].

As regards investment risk, IFIs should have appropriate risk management strategies and reporting processes in place in relation to risk characteristics for equity investments, including mudarabah and musharaka investments. With respect to market risk, for all assets, IFIs should have a system of appropriate technology tools with respect to risk management and reporting, to be considered separately, including for those that are not intended immediately for the market and/or exposed to high volatility pricing. In terms of liquidity risk, IFIs should have a liquidity management framework (including reports) in place, considering separately and in total liquidity exposures related to each category of current accounts, and restricted and unrestricted investment accounts. Finally, for operational risk, IFIs should have adequate systems and controls in place, including oversight to ensure compliance with ethical and religious norms and principles, including the risk of non-compliance with Shariah [58].

Thus, Islamic banks face the same risks (dependent on insolvency, commodity price fluctuations, liquidity management, regulatory and market risks) as conventional institutions, but the typical risk is that of not properly fulfilling the principle of risk sharing in financial transactions or religious precepts related to economic activities.

The principles prepared by the IFSB also concern the transparency and necessary detailed information on the financial instruments and the investment funds for conformity to the Shariah, in order to allow informed decisions according to ethical-juridical rules; furthermore, specific regulations are provided for the composition of the board, their roles and responsibilities, the circumstances in which the non-compliance can occur and the procedures to detect and control the risks, the processes of internal and external audit as well as the possible pecuniary sanctions as a measure of "purification" for the non-compliant incomes. There are requirements for transparency and constant updating of data by appropriate electronic reporting systems.

The Shariah-compliant contracts used by Islamic banks modify the risk-return characteristics of the products and have relevant regulatory implications [59]. However, despite the

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6424 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index

particularities, some regulatory standards of the Basel Committee can be applied to Islamic banks, while the specific characteristics in relation to ethical rules are certainly not considered by international regulatory standards. For example, the use of profit-sharing investment accounts (PSIA) raises several issues, e.g., whether to consider them as deposits or treat them as capital, as there is a risk-sharing element to the assets. Moreover, since the returns of PSIAs are based on the principle of profit-loss ratio, the rights of investor-depositors need to be protected and the fiduciary nature of the contract also requires greater transparency in the related banking operations [60].

This is a concrete problem common to other international bodies adopting rules with standard provisions and, in fact, even the IFSB does not have "enforcement powers" [61] to ensure the effective application of its prudential standards in the various countries involved, and it is inevitable that enforcement depends on national regulators; in essence, few supervisors have incorporated these standards into national regulation and it is possible to state that the regulatory regimes of Islamic banks can be traced to three solutions: the first concerns a few countries (e.g., Bahrain, Oman [62]) that have adopted specific regulatory guidelines; the second solution concerns countries that require banks to follow the regulations foreseen for conventional banks, even if adaptations are necessary to respect the Shariah.

The third solution adopted by countries such as Saudi Arabia and the United Kingdom [63] is to subject Islamic banks to the same regulation as conventional counterparties and Islamic banks adjust their operations to the existing regulation. While in the jurisdictions where there is no regulation of Islamic finance, the banks, also Islamic, adapt to the national regulatory provisions based on the standards of the Basel Committee and in these countries the Islamic banks must use products and instruments that conform to these rules and to the ethical-legal rules of the Shariah even if a frequent contrast emerges between products that conform both to those standards and to the ethical rules. Thus, e.g., under the Basel III rules on capital requirements, Islamic banks should ascertain acceptable capitalisation (Tier-1 and Tier-2) and the value of risk-weighted assets in their portfolios with the new risk benchmarks. Since PSIA cannot be accepted as capital under the strict Basel III definitions, one solution may be to issue long-term equity-based Sukuk that meet the definition of capital under the new regulatory requirements. In addition, as Basel III pays more attention to market and other special risks, the risk references for partnership contracts such as mudarabah and musharaka and sales-based instruments such as salaam and istisna'a (a medium-term funding technique) are likely to increase.

The Basel Committee had already published guidelines in relation to liquidity for risk management prior to the crisis (Basel II), but with the crisis, the central role of capital further emerged, and this led to the definition of specific requirements with Basel III. The aim of introducing the appropriate capital liquidity requirements is to promote a more flexible banking sector, improving its ability to withstand shocks from different sources. Basel III introduced the Liquidity Coverage Ratio (LCR) to meet short-term liquidity needs and risks and the Net Stable Funding Ratio (NSFR) to ensure adequate medium and long-term liquid funds. In particular, Saudi Arabia has implemented the requirements set out in Basel III, through the Saudi Arabian Monetary Authority, which has undertaken a number of initiatives to strengthen the prudential framework for bank capital and issued the Basel III risk-based capital framework in December 2012 to implement the first phase, which came into force in January 2013. A number of new rules and policies were also introduced in October and December 2012, in particular on capital requirements for bank exposure to central counterparties, the 3 Pillar and the Liquidity Coverage Ratio. The United Arab Emirates also recently adopted the capital requirements of Basel III through its own regulations [64].

The risk of liquidity arises when banks face problems in acquiring sums of cash at reasonable cost from loans or from the sale of assets. In conventional banks, the needs of financial liquidity can be satisfied by private sources, by other financial institutions and by inter-bank money markets. The Islamic banks, instead, must consider the constraints of the Shariah in accessing liquidity from these sources. Since interest-bearing loans are prohibited, under Shariah it is not possible to borrow funds to meet liquidity requirements in case of need and, therefore, there are limits to borrowing; moreover, debt sales are prohibited by Islamic law in most cases and banks could not sell their debt-based assets to ensure market liquidity.

In this context, the identification of quantitative limits to borrowing is a central issue, as banks cannot receive large sums as interest payments and loans must not consist of more assets than the assets held, and loans involving large debts cannot be used to ensure liquidity; over the years, the general rules have been expressed more clearly as a certain reference for the banks and, in particular, the Shariah-compliant financial parameters [65] of the Dow Jones Islamic Market Indexes [66] are relevant as criteria of reference for the Shariah boards of the financial institutions [67]. For example, it is prohibited to hold shares in corporations which are in debt to the amount of more 33% of their own capital resources, which draw more than 5% of their profits from non-operating interest revenue or whose receivables amount to more than 45% of their own capital resources. Besides, there were several attempts to establish generally binding standards in order to guarantee sufficient legal security such as the Shariah standard of the Accounting Organisation for Islamic Financial Institutions (AAOIFI) based in Bahrain [68].

Without an adequate supply of Shariah-compliant liquidity instruments and active markets in which to trade, it will be difficult for Islamic banks to meet the requirements of Basel III, since they would have to shorten the maturity of their assets and increase the duration of their liabilities and capital, in order to meet the liquidity requirements. This may put Islamic banks at a disadvantage compared to their conventional counterparts and create obstacles to long-term growth; therefore, additional measures and initiatives with innovative interpretations for Shariah compliance need to be

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6424 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index

introduced to address liquidity requirements.

In many contexts, there are no inter-Islamic banks or organised money markets with reference funds in the event of a crisis. Furthermore, due to the lack of liquidity for Sukuk instruments and active Sukuk markets, the Islamic banks present significant risks of liquidity, given that these certificates represent a positive alternative to the traditional bonds (the issuer must return to the buyer the initial countervalue and the agreed rate of interest), but are structured in such a way that the profits are linked to a concrete asset or to an underlying property with a relative right of ownership.

While central banks in some countries play an important role in providing tradable instruments to meet short-term liquidity needs, liquidity quotas are still a major problem, forcing many banks to raise more funds. The prohibition of the interest rate entails difficulties in the management of the liquidity in the absence of a Shariah-compliant inter-banking market and because of the impossibility of investing in government securities at short expiry and at low risk; a solution is represented by the issuance of Sukuk to be understood as share certificates of Islamic investment, even if related, by affinity, to bonds. Therefore, liquidity management is one of the most challenging and relevant objectives, but with a particular difficulty if liquidity instruments and infrastructure do not evolve. A robust liquidity framework for the Islamic financial sector will be needed, not only to meet liquidity requirements under Basel III, but also for the development of private sources of liquidity and public support services.

# V. CONCLUSIONS

# A. Prospects for Islamic finance in the EU and future economics law research

In the European system of conventional banks, the need adapt to the requirements of the Basel standards, to transparency rules and digital transformation and innovation, raise issues that may inhibit the development of Islamic finance. In fact, the Euro-system provides for interest-based instruments, and, for prudential regulation and supervision, all European banks are required to adhere to deposit insurance and financial services clearing systems, especially after the Banking Union. In contrast, Islamic jurisprudence, which has long prevailed in several Islamic countries, has held that the investment deposits of Islamic banks, which follow the "profit and loss sharing principle", cannot be covered by deposit guarantee schemes. However, recently, this aspect has evolved due to the significant role played by the IFSB. Moreover, there is a significant further obstacle to the creation of Islamic banks, as the corporate governance and management board of European banks, which take full responsibility for management decisions based on EU and national laws, can certainly not be shared with the Shariah board [69].

In general, therefore, there are several legal difficulties and complex profiles for the spread of Islamic banks in Europe, precisely because of the prevailing legal schemes. The establishment of banks would require a specific legal rule, not so much because of the European regulation that, in any case, is in line with the well-known international standards, but because of the issue of banking authorisation which still involves the supranational and national regulatory level and above all because of the internal national rules that need to be integrated.

In fact, Article 14.1, Reg. (EU) n. 1024/2013 provides that any application for an authorisation to take up the business of a credit institution to be established in a participating Member State shall be submitted to the national competent authorities of this legal system in accordance with the requirements set out in relevant national law.

Thus, for example, in the case of an Islamic credit institution to be established in Italy, the related application must be consistent with the objective requirements determined by Article 14 of the Italian Banking law (Testo Unico Bancario), in line with the approach introduced by Directive 1977/780/EU now merged into Directive 2013/36/EU [70].

The authorisation is an administrative measure, which is issued when the conditions laid down in Article 14.1 are met. It is issued by the ECB, on the proposal of the Bank of Italy, and is denied by the ECB or the Bank of Italy, when the verification of the conditions, provided for in paragraph 1, does not guarantee sound and prudent management. For the establishment of non-conventional banks, it would be necessary to introduce specific rules on the admissibility of issuing authorisation with additional conditions for carrying out the activity, given the profound differences, just as specific rules are needed for the offering of Islamic financial products by domestic banks.

The study of the German experience could be a useful investigation for new rules and future economic law research, in order to identify a solution compatible with Italian (or other European) banking laws and tax disciplines, on the assumption of a clear desire on the part of the legislator, obviously fallowing a political decision.

On the other hand, it would be easier to foresee European banks offering Islamic financial products or branches, although the problem of taxation or other legal compatibility questions would arise. This solution should be encouraged and studied, in view of the tangible advantages associated with it and for the presence of Islamic savers and investors, but with adequate controls on the purpose of the investments. However, EU and national legislation need a solution for reasons of legal certainty.

Apart from these central aspects, it is necessary to consider the process of profound change, already underway and accelerated by external events, affecting the activities of banks, the acceleration to new technologies and the related and new risks associated with them, issues that require rapid innovations in terms of the tasks of regulators and legal disciplines.

From the European perspective, the pandemic has confirmed the importance of a solid system of rules capable of directing banks towards virtuous and prudent behaviour and strengthening the ability of the system to cope with the crisis. Reforms and the establishment of the European single

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6424 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index

supervisory mechanism go back some ten years to address the financial crisis, but the framework made it possible to respond to the problems of the health crisis with more adequate capital strength and liquidity. Banks during the most acute phase of the pandemic supported the economy and regulators and supervisors strengthened the resilience of the financial system.

Currently, the Basel Committee intends to improve prudential rules based on studies to verify their actual adequacy and effectiveness; the implementation of the "Final Basel III" standards has been postponed to January 2023 due to the pandemic and to increase banking capacity to face the new challenges.

These considerations can only be extended to Islamic banks: their challenge will be the "Final Basel III" implementation for their attractiveness in the international markets, and future success will depend on the imminent further adjustment in the direction of homogeneous standards. Certainly, this evolution will be examined by future legal and economics research.

Thus, several regulatory measures were adopted at international and EU level, allowing banks to absorb losses and lend during a very challenging period. The choices of postponing the implementation of the recommendations to 2023 and not adopting any further reforms (the so-called "Basel Hard Stop"), in order to ensure the medium-term stability of the prudential regulatory framework, are consistent and appropriate, while not undermining the commitment of banks to transpose the new rules. The European Commission has formulated a proposal for a legislative act from which a wide-ranging debate has ensued to ensure the competitiveness European banks, compliance with the of Basel recommendations and the application of the proportionality of the rules, without any substantial weaknesses [71].

This context undoubtedly has international consequences and can only affect Islamic banks and their relations with EU banks. For future research it will be interesting to examine how the forms of risks (credit, market and operational) are dealt with in the European Banking Union, according to the Basel indications, and how the same risks are assessed by the IFSB which, moreover, has adopted the same rules, precisely in order to cultivate profitable relations with conventional banks.

The reform of the Basel III rules has pursued three objectives to improve credibility, simplicity and comparability and has affected all risks of the so-called first pillar (credit, operational, market). In the case of market risk, it has not been easy to reconcile credibility, which is linked to the ability of quantitative measures to correctly reflect risks, with that of simplicity, which entails the rapid and immediate application and interpretation of calculation methods. Consequently, the search for a compromise has prevailed.

The previous reform of the Basel standards took place more than a decade ago, in connection with the 2007/2008 crisis, involving a comprehensive review of prudential standards and confirming the appropriateness of risk-sensitive rules. Indeed, from the outset, the Basel II rules had significant limitations and an inadequate incentive scheme, and so needed to be corrected. The new formulation has improved the credibility of risk measures, including through the reporting system, and has limited non-responsible behaviour in risk management. The pandemic has certainly confirmed the desirability of ensuring the existence of adequate capital for banks that can help to cope with external difficulties as financial support for companies and other private clients.

Regarding credit risk, as the EBA pointed out in the "Call for advice" of the European Commission, numerous revisions to be considered as widespread practices in the EU. Other revisions allow the European legislator discretionary choices that may contribute to the adoption of a discipline adapted to the specific aspects of the European systems. For operational risk, Final Basel III significantly amends the current framework by providing a single standardised approach for calculating the capital charge that replaces the previous approaches, of which three were standardised and one was based on internal models, which showed complexity, inadequacy in identifying the risk profiles of the banks, and significant problems of comparability across banks and systems.

Regarding market risks, the path has been very complex. The new standards, which were approved in an initial version in 2016 and in 2019, change the treatment of risks according to a more appropriate approach to identify all the risks related to the trading activities of banks. The increased focus on risk has resulted from the increased complexity of the rules, as the new standard has been defined based on the operations of major international banks.

The transposition of the rules in Europe entails the application of the principle of regulatory proportionality, adequately balancing the simplicity and specificity of the measures, the provision of strict conditions for the use of internal models, but which must not constitute regulatory barriers accessible only to the major international banking groups, and simplified solutions for those banks that carry out insignificant trading activities, even though less sensitive to risk.

This complex evolution inevitably affects the adaptation of Islamic banking and financial products for access to the European market. There are several practical implications that cannot be solved by interpretation, but require a specific European legal framework on which to base national implementing rules, a model that is already applied to conventional banks for harmonisation requirements.

Therefore, a special regulation is needed for the coexistence in the Banking Union of European banks and Islamic banks and financial products, at least to obtain banking authorisation for legal certainty and uniformity of solutions. In addition, national supervisors could assess the admissibility of Shariahcompliant banking contracts offered by conventional banks through the combination of several schemes that allow for effects that comply with domestic banking laws. The adoption of legal rules will, of course, depend on political choices, both at supranational and internal level.

Economic studies examine the relevance of Islamic banking activities also for some interesting sustainability aspects and this is a concrete basis for legal considerations to request regulatory interventions at European and national level.

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6424 - Published by University of Turin https://www.ojs.unito.it/index.php/EJIF/index

Adaptation, not immediate and simple, to international standards cannot be considered the only solution because of the implications, also theoretical, of the relationship between two sectors that are based on some very different principles.

Finally, we must mention the various similarities between the principles of Islamic finance (e.g., in the case of sukuk) and the principles of public-private partnership for infrastructures [72], regarding risk sharing, the lender as partner, the exclusion of speculative intentions and certainty in the definition of contractual performance. The comparison could be the topic for interesting studies in the future.

Moreover, the United Nations published a study in 2016 highlighting the possible contribution of Islamic finance to promoting the sustainable development goals of the 2030 Agenda to finance infrastructure. However, sustainable finance requires rules and techniques to manage risk and technology to enable knowledge of standards and accounting rules for the maximum transparency in transactions, and control of compliance with Islamic law as well as compatibility with the conventional system: objectives that are very complex to achieve [73].

# REFERENCES

[1] R. Wilson, "Islamic banking in the United Kingdom", M. F. Khan and M. Porzio (eds.), Islamic Banking and Finance in the European Union: A Challenge, Cheltenham: E. Elgar, 2010, 212 ff.

[2] V. J. Sadeghi and P. P. Biancone, "Shariah Compliant International Entrepreneurship: A Study of Islamic Finance in Europe", *European Journal of Islamic Finance*, 2017, 8, 4 ff.; P. P. Biancone and M. Radwan, "European Companies: Evaluation for Sharia Compliance "Opportunities and Challenges", *European Journal of Islamic Finance*, 2017, 5, 1 ff.

[3] M. F. Khan, "Islamic banking in Europe: the regulatory challenge", M. F. Khan and M. Porzio (eds.), Islamic Banking and Finance in the European Union, cit., 61 ff; C. Porzio, "Islamic banking versus conventional banking", ivi, 91 ff.; L. Donato and M. A. Freni, "Islamic banking and prudential supervision in Italy", ivi, 189 ff.; V. Santoro, "The riba prohibition and payment institutions", ivi, 222 ff.; R. Wilson, "The interface between Islamic and conventional banking", M. Iqbal and D. T. Llewellyn (eds.), Islamic Banking and Finance, Cheltenham: E. Elgar, 2002, 198 ff.; Y. Abdul-Rahman, The Art of RF (Riba-Free), Islamic Banking and Finance: Tools and Techniques for Community-based Banking, Hoboken: Wiley, 2014. [4] See the Koran (Sura II, 29). For Islamic law and economy principles, see A. Predieri, "Il diritto apicale metastatale. Analogie fra diritto comunitario e diritto islamico", Diritto dell'Unione Europea, 1996, 677 ff.; H. Askari and R. Taghavi, "I fondamenti di principio di un'economia islamica", Moneta e Credito, 2005, LVIII, n. 232, 175-178; A. Predieri, Shari'a e Costituzione, Roma:

Laterza, 2006; A. Habib "Product development in Islamic Banks", Edinburgh: Edinburgh University Press, 2011; L. M. Al-Rimawi, Raising Capital on Arab Equity Markets, Legal and Juridical Aspects of Arab Securities Regulation, Croydon: Kluwer Law International, 2012, 102 ff.; I. Zamir and M. Abbas (eds.), Economic development and Islamic finance, Washington: The World Bank, 2013; E. Francesca, Economia, religione e morale nell'Islam, Roma: Carocci, 2013; M. Donini and D. Scolart, La sharì'a e il mondo contemporaneo, Roma: Carocci, 2016; P. P. Biancone, La banca islamica, Torino: Giappichelli, 2017; F. Miglietta and P. P. Rampino, Gli strumenti di finanza islamica a sostegno del Sistema Italia, Milano: Pearson Italia, 2017, 7 ff.; S. I. u. D. Asad, "Islamic Finance: Concepts, Transactions, Developments, and Critique", W. Decock and V. Sagaert (eds.), Islamic Finance, Between Religious Norms and Legal Practice, Cambridge: Intersentia, 2019, 39 ff.

[5] K Khan, "How 'Islamic' is Islamic Banking?", *Journal* of Economic Behavior & Organization, 2010, 816-817; J. D. Fry and J. M. Taylor, "Foreign Direct Investment in Arab Countries: A Guide to Better Understanding Islamic Financial Doctrine", The International Bureau of the Permanent Court of Arbitration (Ed.), Strengthening Relations with Arab and Islamic Countries through International Law, Croydon: Kluwer Law International, 2002, 302-304; T. V. Russo, I contratti Shari'a compliant, Valori religiosi e meritevolezza degli interessi. Contributo allo studio, Napoli: ESI, 2014, 43 ff.

[6] G. M. Piccinelli, Banche islamiche in contesto non islamico, Materiali e strumenti giuridici, Roma: Istituto per l'Oriente, 1996, 22-23; G. M. Piccinelli, "Operazioni islamiche di provvista e di gestione del risparmio: il modello del cliente-socio", G. Gimigliano and G. Rotondo (a cura di), La banca islamica e la disciplina bancaria europea, Milano: Giuffrè, 2006, 17 ff.

[7] Porzio C., "Intervento", Gimigliano G. and Rotondo G. (a cura di), La banca islamica e la disciplina bancaria europea, cit., 84-85; S. I. u. D. Asad, "Islamic finance: concepts, transactions, developments, and critique", W. Decock and V. Sagaert (eds.), Islamic Finance, cit., 49-50; A. Gatto, "Principi e tecniche di finanza islamica", Economia, impresa e mercati finanziari, 2008, 31 ss.; C. Porzio (a cura di), Banca e finanza islamica, Roma: Bancaria Editore, 2009, 28-29; F. Miglietta and M. G. Starita, "Una tassonomia dei contratti", ivi, 31 ff.; M. K. Lewis, "Adapting understanding of riba to Islamic banking: some developments", M. Ariff M. and M. Iqbal (eds.), The Foundations of Islamic Banking, Cheltenham: E. Elgar, 2011, 54 ff.; M. K. Lewis, "A theoretical perspective on Islamic banking and financial intermediation", M. K. Lewis, M. Ariff, S. Mohamad (eds.), Risk and Regulation of Islamic Banking, Cheltenham: E. Elgar, 2014, 35-37; F. Khan, "How "Islamic" is Islamic Banking?", cit., 808-810; H. Ahmed, "Basel III liquidity requirement ratios and

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6424 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index

Islamic banking", Journal of Banking Regulation, 2015, 253-254.

[8] S. I. u. D. Asad, "Islamic finance: Concepts, Transactions, Developments, and Critique", W. Decock and V. Sagaert (eds.), Islamic Finance, cit., 52-53.

[9] M. Ariff and M. K. Lewis, "Similarities and differences in Islamic and conventional banking", M. K. Lewis, M. Ariff, S. Mohamad (eds.), Risk and Regulation of Islamic Banking, cit., 56 ff.

[10] R. Wilson, "Christian and Islamic Perspectives on Contemporary Finance", W. Decock and V. Sagaert (eds.), Islamic Finance, cit., 14 ff.

[11] New rules were approved in December 2017 that should have come into force in 2022 with more stringent requisites of capital and new patrimonial indexes of the banking assets to face the market risks. The entry into force, initially provided for the 1st of January 2019, was postponed to the 1st of January 2022, to allow the adjustments in a longer time with a transitional regime until the 1st of January 2027, date of full application. See "Basel III: Finalising post-crisis reforms. December 2017" with further official documents at www.bis.org. For the Covid pandemic, the entry into force of the "Final Basel III" standards has been postponed until 1st of January 2023. See A. Pilati, Banca d'Italia, "Final Basel III e oltre: inquadramento evolutivo, prime analisi d'impatto, prossimi passi", 22 giugno 2021; A. Louhichi et al., "The regulationrisk taking nexus under competitive pressure: What about the Islamic banking system?", Research in International Business and Finance, 2020, vol. 51, January. About measures of banking supervision to address the consequences of the pandemic, see K. Lackhoff (ed.), Banking Supervision and COVID-19, A Handbook, München: Verlag C.H.Beck, 2021.

[12] According to IFSB Stability report 2021.

[13] G. Samuel, "Comparative law and its methodology", D. Watkins and M. Burton (eds.), Research Methods in Law, Second Edition, London: Routledge, 2017, 122-145; L. J. Moran, "Researching the visual culture of law and legal institutions: some reflections on methodology", *Journal of Law and Society*, 2021, 48, 44-58.

[14] A. F. Aysan, M. Disli, M. Duygun, H. Ozturk, "Religiosity versus rationality: Depositor behavior in Islamic and conventional banks", *Journal of Comparative Economics*, 2018, 1 ff.; S. G. Prandi and D. Colecchia, "Pricing of Islamic Banking and Conventional Banking: An Empirical Study", *European Journal of Islamic Finance*, 2021, 18, 1 ff.

[15] R. Sbia, H. Hamdi, B. K. Onur Tas, S. Al Rousan, "Gulf Cooperation Council Stocks and the Effect of Domestic Monetary Policy Shocks", *International Journal of Economics and Financial Issues*, 2016, 6(2), 629 ff.

[16] M. Ariff and M. K. Lewis, "Similarities and differences in Islamic and conventional banking", cit., 62 ff.; M. K. Lewis, "Ethical principles in Islamic business

and banking transactions", M. Ariff and M. Iqbal (eds.), The Foundations of Islamic Banking, cit., 60 ff.; A. M. El Tiby, Islamic Banking, New York: Wiley, 2011, 73 ff.; H. Ahmed, "Basel III liquidity requirements ratios and Islamic banking", cit., 254-255.

[17] G. M. Piccinelli, Banche islamiche in contesto non islamico, cit., 31-32; Khan F., "How 'Islamic' is Islamic Banking?", cit., 815-816; T. V. Russo, I contratti Shari'a compliant, cit., 143 ff.

[18] Wilson R., "Islamic Banking in the United Kingdom", G. Gimigliano and G. Rotondo (a cura di), La banca islamica e la disciplina bancaria europea, cit., 215 ff.; M. Abu-Alkheil and H. – P. Burghof, W. A. Khan, Islamic Commercial Banking in Europe: A Cross-Country and Inter-Bank Analysis of Efficiency Performance, *International Business & Economics Research Journal*, 2012, n. 6, 647 ff. In 2004 the Financial Service Authority issued the banking authorisation and in December 2014, the name "Islamic Bank of Britain" was changed to "Al Rayan Bank", see *www.alrayanbank.co.uk*.

[19] R. Wilson, "Competition in Islamic banking", R. Wilson (ed.), Islamic Financial Markets, London: Routledge, 1990, 19 ff.; A. Khan, M. K. Hassan, A. Paltrinieri, S. Bahoo, "Trade, financial openness and dual banking economies: Evidence from GCC Region", *Journal of Multinational Final Management*, 2021, 62, 1-16.

[20] H. Ahmed, Product Development in Islamic Banks, Edinburgh: Edinburgh University Press, 2011, 202 ff; H. Ahmed, M. Asutay, R. Wilson, "Reflecting on Islamic Banking and Financial Crisis: Risks, Reputation and Stability", H. Ahmed, M. Asutay, R. Wilson (eds.), Islamic Banking and Financial Crisis: Reputation, Stability and Risks, Edinburgh: Edinburgh University Press, 2014, 5 ff.

[21] R. Costi, "Conclusioni", Gimigliano G., Rotondo G. (a cura di), La banca islamica e la disciplina bancaria europea, cit., 240 ff.; Ariff M., Lewis M. K., "Similarities and differences in Islamic and conventional banking", cit., 57 ff.

[22] K. Ginena, "Shari'ah risk and corporate governance of Islamic banks", Corporate Governance, 2014, vol. 14, 1, 98 ff. About banking supervision and ethical-religious criteria, see N. Trad, A. M. Trabelsi, J. F. Goux, "Risk and profitability of Islamic banks: A religious deception or an alternative solution?", European Research on Management and Business Economics, 2017, 40 ff.; C. W.H. Cheong, "Risk, resilience, and Shariah-compliance", Research in International Business and Finance, 2021, 55, 1 ff.; M. K Hassan, A. Khan, A. Paltrinieri, "Liquidity risk, credit risk and stability in Islamic and conventional banks", Research in International Business and Finance, 2019, 48, 17 ff. See IFSB-19, Guiding Principles on Disclosures Requirements for Islamic Capital Market Products (Sukuk and Islamic Collective Investment Schemes), April 2017, at www.ifsb.org.

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6424 - Published by University of Turin https://www.ojs.unito.it/index.php/EJIF/index

[23] The General Secretariat, based in Riyadh, prepares reports, studies and accounts for the GCC; it also drafts rules and regulations and is responsible for assisting Member States in implementing decisions taken by the Supreme and Ministerial Councils, at *www.gcc-sg.org*; M. Sturm and N. Siegfried, "Regional Monetary Integration in the Member States of the Gulf Cooperation Council", ECB Occasional Paper Series, n. 31, June 2005, 25 ff., at *www.ecb.europa.eu*.

[24] M. Sturn and N. Siegfried, "Regional Monetary Integration in the Member States of the Gulf Cooperation Council", cit., 32 ff; A. Alkholifey and A. Alreshan, "GCC monetary union", at *www.bis.org*. See Gulf Cooperation Council (GCC) and the EU, External Action, at *www.europa.eu*; ECB, "Economic integration in selected regions outside the European Union", Monthly Bulletin, October 2004, 67 ff.; Habib M. M. and Stra'sky' J., "Oil Exporters in Search of an External Anchor", ECB Occasional Paper Series, n. 958, November 2008, 7 ff., at *www.ecb.europa.eu*.

[25] M. Miraglia, "Islamic Banks: Criticality of the System and Profiles of Regulation, Perspectives of Harmonization", *Rivista trimestrale di diritto dell'economia*, 2011, 338 ff. See ECB, *Annual Report* 2005,152-153; ECB, Monthly Bulletin, 2008, 14 at *www.ecb.europa.eu*. For example, ECB Seminar, 19th January 2012, with GCC States central banks and monetary agencies, organised by the ECB and the Central Bank of United Arab Emirates (ECB, Report 2012, 154). See ECB, Report 2013, 169; Report 2014, 106; Report 2015, 101 and Report 2016, 104, at *www.ecb.europa.eu*.

[26] S. Antoniazzi, La Banca Centrale Europea tra politica monetaria e vigilanza bancaria, Torino: Giappichelli, 2013; Id., "L'Unione bancaria europea: i nuovi compiti della BCE di vigilanza prudenziale degli enti creditizi e il meccanismo unico di risoluzione delle crisi bancarie", 1-II, *Rivista italiana di diritto pubblico comunitario*, 2014, n. 2 and n. 3-4, 359 ff. and 717 ff.; Id., "The ECB's Banking Supervision and European Administrative Integration: Organisation, Procedures and Legal Acts", *Italian Journal of Public Law*, 2015, 318 ff.; R. Ayadi and W. P. De Groen, "Banking and Insurance in the GCC Countries: Is there Regulatory Convergence with the EU?", CES Research Papers, n. 4, 2013, 101, at *www.cesp.eu*.

[27] In United Kingdom, the Al Rayan Bank, The European Islamic Investment Bank, The Bank of London & the Middle East, Securities House, The European Finance House were established and in France, e.g., the National Bank of Kuwait, Tejerat Bank, Qatar National Bank; while in Germany: The Irani Bank Sepah, Kuveyt Turk. In addition, some conventional banks also offer Islamic financial products, e.g., Commerzbank, Deutsche Bank, Dresdner Bank (Germany), Crédit Agricole (France), HSBC Group, USB Group, Barclays, Lloyds (UK). See W. Mansour, M. B. Abdelhamid, O. Masood, G. S. K. Niazi, "Islamic banking and customers' preferences: the case of the UK", *Qualitative Research in Financial Markets*, 2010, n. 3, 185 ff.; D. Ecke, Islamic Banking: Grundlagen und Potenzial in Deutschland, Hamburg: Diplomica Verlag GmbH, 2012; Grewal K., "Islamic finance in the global financial system", F. Di Mauro et al., Islamic finance in Europe, ECB Occasional Paper n. 146, June 2013, 25 ff., at *www.ecb.europa.eu*; M. Ashfaq, Islamic Banking and Finance in Europe: the Case of Germany and United Kingdom, A Theoretical and an Empirical Analysis, Frankfurt: PL Academic Research, 2017.

[28] See *www.kt-bank.de*. The *KT Bank AG* is the first bank in Germany and the Eurozone that has introduced comprehensive financial products and services according to the "ethical, sustainable and transparent Islamic banking principles" for the Muslim community and for all other customers interested. This bank is a wholly owned subsidiary of the Turkish Kuveyt Türk Participation Bank established in Istanbul. Kuveyt Türk has been laying groundwork in Germany since 2004 by means a German branch that explained the Islamic model. In 2010, KT Bank AG entered the market and was granted, by the Federal Financial Supervisory Authority (BaFin), the authorisation for non-EEA deposit broking. Then in March 2015, the BaFin granted a full banking license under German law for the provision of deposit and credit business in Germany. KT Bank is also a member of the "Entschädigungseinrichtung deutscher Banken GmbH" (EdB), to secure clients' deposits up to € 100,000. On July 1st, 2015, it started business branches in Berlin, Frankfurt, Mannheim, Cologne and Munich, planning to offer Islamic products in other EU countries. The KT Bank AG is a subsidiary of the Kuveyt Türk Participation Bank A.S. established in 1989 in Istanbul, and is leading the Islamic banking market in Turkey for more than thirty years. Kuveyt Türk's shareholders include the Kuwait Finance House, the most important finance house in the GCC Region (62%) and the Turkish General Directorate of Foundations (18%).

[29] G. Gomel (coord.), Islamic Finance and Conventional Financial Systems. Tendencies of the market, profiles of supervision and implications for the activities of the central bank, Bank of Italy, Issues of Economy and Finance (Occasional Papers), n. 73, October 2010, at *www.bancaditalia.it*; N. Schoon, Modern Islamic Banking: products, processes in practice, Chinchester: Wiley, 2016, 158 ff.

[30] "Dubai Islamic Economy Development Centre Report" 2020/21 (SGIE), at *www.iedcdubai.ae*; S&P Global report "Islamic Finance 2021-2022: Toward sustainable Growth", at *www.spglobal.com*.

[31] IFSB Regulatory and Supervisory Measures to Mitigate the Impact of Covid-19: Recommendations Relating to the Takāful/Retāful Industry, 6 August 2021; A. A. Adewale, "Assessing the Stability of the Islamic

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6424 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index

Banking Industry Amid the Covid-19 Pandemic", IFSB Working Paper, WP-18/12/2020. For economic studies see M. M. Hamed, "The Role of Islamic Social Finance in Mitigating Humanitarian Crises", *European Journal of Islamic Finance*, 2020, n. 16, 1 ff.; Md. Tanvir Alam, "Role of Islamic Finance during COVID-19: A Study on Practical Implication of Zakat as Short-term Emergency Support System", *European Journal of Islamic Finance*, 2020, n. 16, 1 ff.

[32] "Eurosystem and GCC central banks and monetary agencies hold second high-level seminar", Rome, 29-30 June 2010; Eurosystem and GCC central banks and monetary agencies hold second high-level seminar, Mainz, 14 March 2008, at *www.ecb.europa.eu*. See G. Gomel (coord.), Finanza islamica e sistemi finanziari convenzionali. Tendenze di mercato, profili di supervisione e implicazioni per le attività di banca centrale, Banca d'Italia, Questioni di Economia e Finanza (Occasional Papers), n. 73, ottobre 2010, at *www.bancaditalia.it*.

[33] See the draft law at *www.camera.it*; A. Franco and C. Sallustio, "The taxation of sukuk in the Italian context: is Italy's tax system ready for Islamic financial instruments?", *European Journal of Islamic Finance*, 2017, n. 7, 1-6; F. Miglietta and P. P. Rampino, Gli strumenti di finanza islamica a sostegno del Sistema Italia, cit., 67 ff.

[34] IFSB Forum on Covid-19 and its implications on Institutions offering Islamic Financial Services, 26 August 2020; the IFSB's 2<sup>nd</sup> Innovation Forum Discussing "Digital Transformation of Islamic Finance and other Innovative Solutions for Post-Covid 19 Recovery". For new rules and standards see "Exposure Draft, Core Principles for Islamic Finance Regulation [Financial Market Infrastructures]", 10 September 2021; Core Principles for Islamic Finance Regulation (Financial Market Infrastructures), 26 December 2021; for further documents and draft rules see *www.ifsb.org*.

[35] The ISFB was established in Malaysia with the support of the International Monetary Fund on the 3rd of November 2002 (operative from the 10 March 2003), including banks of many countries (United Arab Emirates, Bahrain, Saudi Arabia, Oman, Qatar and Kuwait, Malaysia, etc.); see *www.ifsb.org*.

[36] Islamic Financial Services Industry Stability Report 2021, 101 ff.; see Reports 2015/2020.

[37] D. A. Asafa et al., "Effectiveness of Macroprudential Tools for Islamic Banking", IFSB Working Paper Series, WP-27/08/2021, 23 ff.

[38] A. A. Adeyinka, "Regulatory and Supervisory Issues in *Takaful* Windows", IFSB Working Paper Series, WP-16/12/2020; Report 2021, 6.

[39] Report 2021, 7-10; 40 ff.; 121; 118 ff.

[40] About the assessment of the soundness and resilience of the Islamic financial system and analytical data, see Report 2021, 39 ff.

[41] Report 2021, 128 ff.

[42] A. S. Gintzburger, "An Analysis of Global Trends and Regional Pockets in the Application of Islamic Financial Contracts in Malaysia and the Gulf Cooperation Council", M. K. Hassan and M. Mahlkhecht (eds.), Islamic Capital Markets, Chippenham: Wiley, 2011, 316 ff.; A. Chazi, A. Khallaf, Z. Zantout Z, "Corporate Governance and Bank Performance: Islamic Versus Non-Islamic Banks in GCC Countries", *The Journal of Developing Areas*, 2018, 52, 2, 109 ff; N. M. Nomran, R. Haron, R. Hassan, "Shai'ah Supervisory Board Characteristics Effects on Islamic Bank's Performance", *The international Journal of Bank Marketing*, 2018, 36 (2), 290 ff.

[43] A. Makhlouf, "The Transformation of Islamic Law: from Classical Fiqh to Financial Fiqh", W. Decock and V. Sagaert (eds.), Islamic Finance, cit., 125 ff.; M. K. Hassan, F. Miglietta, A. Paltrinieri, J. Floreani, "The effects of Shariah board composition on Islamic equity indices' performance", *Business Ethics*, 2018, 27, n. 3, 248-259; S. K. Kok et al., "The trade-off between knowledge accumulation and independence: The case of the Shariah supervisory board within the Shariah governance and firm performance nexus", *Research in International Business and Finance*, 2022, 59 (C).

[44] H. Ahmed, M. Asutay, R. Wilson, "Reflecting on Islamic Banking and Financial Crisis: Risks, Reputation and Stability", cit., 12 ff.

[45] Z. Man, The Islamic Banking System: The Experience of Malaysia, Islam and Finance, Istituto di Studi Sud Asiatici, Turino: Edizioni Fondazione G. Agnelli, 1991, 81 ff.

[46] Z. Shafii and S. Salleh, "Enhancing Governance, Accountability and Transparency in Islamic Financial Institutions: An Examination into Shari'a Internal Control Audit", H. Ahmed, M. Asutay, R. Wilson (eds.), Islamic Banking and Financial Crisis, Reputation, Stability and Risks, cit., 149 ff.

[47] H. Ahmed, M. Asutay, R. Wilson, "Reflecting on Islamic Banking and Financial Crisis: Risks, Reputation and Stability", cit., 13.

[48] A. R. Alotaibi and A. V. Mishra, "Time varying international financial integration for GCC stock markets", *The Quarterly Review of Economics and Finance*, 2017, 63, 66 ff; M. Darayseh and A. Chazi, "Bank Specifics, Economics Enviroment and Agency Theory: Determinats of Banking Performance in GCC", *The Journal of Developing Areas*, 2018, 52 (4), 199 ff.; A. Y. H. Saif-Alyousfi and A. Saha, "Determinants of banks'risk-taking behaviour, stability and profitability: evidence from GCC countries", *International Journal of Islamic and Middle Eastern Finance and Managament*, 2021, 14 (5), 874-907.

[49] A. S. Gintzburger, "An Analysis of Global Trends and Regional Pockets in the Application of Islamic Financial Contracts in Malaysia and in the Gulf Cooperation Council", M. K. Hassan and M. Mahlknecht (eds.), Islamic Capital Markets, Chippenham: Wiley, 2011, 316 ff.

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6424 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index

[50] T. Azrak et al., "Does information disclosure reduce stock price volatility? A comparison of Islamic and conventional banks in Gulf countries", *International Journal of emerging Markets*, 2021, 16 (8), 1769-1792.

[51] IFSB, Islamic Financial Services Industry, Stability Report 2017, 43 ff.; Strategic Performance Plan 2016-2018; IFSB-19 Guiding Principles on Disclosure Requirements for Islamic Capital Market Products, 2-4.

[52] See IFSB, TN-2, Technical Note on Stress Testing for Institutions Offering Islamic Financial Services (IIFS), December 2016; IFSB, Exposure Draft Technical Note on Stress Testing for Institutions Offering Islamic Financial Services (IIFS) [TN-2].

[53] See IFSB-19 Guiding Principles on Disclosure Requirements for Islamic Capital Market Products, 6 ff. and 14 ff.

[54] For example, the Dubai Financial Services Authority, an independent regulator within the Dubai International Financial Center, adopted in 2007 a Hedge Fund Code of Practice as a general regulation of hedge funds.

[55] IFSB, Islamic Financial Services Industry, Stability Report 2017, cit., 35 ff.; A. Msatfa, "Basel III in the Islamic Finance Industry", *Journal of Investing*, 2012, 165 ss.; O. M. Al-Hares, N. M. AbuGhazaleh, A. M. El-Galfy, "Financial Performance and Compliance with Basel III Capital Standards: Conventional vs. Islamic Banks", *The Journal of Applied Business Research*, 2013, 4, 1031 ss.; H. Visser, "Islamic Finance in Non-Muslim-Majority Jurisdictions: Regulatory Issues", *Islamic Finance*, 2019, 83 ff.

[56] Guiding Principles on Shari'ah Governance Systems for Institutions Offering Islamic Financial Services (*IFSB-10*); IFSB has organised its online executive Program, "Managing Digital Transformation Risks for Islamic Financial Institutions (IFIs), 30 September 2021; A. A. Adewale and R. Ismal, "Digital Transformation in Islamic Banking", IFSB Working Paper Series, WP-19/12/2020.

[57] A. Boumediene, "Is credit risk really higher in Islamic banks?", *The Journal of Credit Risk*, 7 (3), 2011, 97 ff.; T. Chamberlain, S. Hidayat, A. Khokhar, "Credit Risk in Islamic and Conventional Banking", *International Advances in Economic Research*, 2018, 24 (1), 99 ff.

[58] Z. Shafii and S. Salleh, "Enhancing governance, accountability and transparency in Islamic financial institutions: An examination into the audit of Sharia'a internal control system", *Malaysian Accounting Review*, 2010, 9 (2), 23 ff.; S. Z. Farook and M. O. Farroq, "Incentive-based regulation for Islamic banks", *Journal of Islamic Accounting and Business Research*, 2011, 2 (1), 8 ff.; E. Oz, Z. R. Khokher, M. M. Ali, R. Rosman, "Shari'ah Non-Compliance Risk in the Banking Sector: Impact on Capital Adequacy Framework of Islamic Banks", IFSB Working Paper Series, 5 March 2016, 1 ff. About risk regulation see N. Alam, "Impact of banking regulation on risk and efficiency in Islamic banking", *Journal of* 

*Financial Reporting and Accounting*, 2013, 1, 29 ff.; P. Abedifar, P. Molyneux, A. Tarazi, "Risk in Islamic Banking", *Review of Finance*, 2013, 2035 ff.; A. I. Maghyereh and B. Awartani, "The effect of market structure, regulation, and risk on banks efficiency", *Journal of Economic Studies*, 2014, n. 3, 405 ss.; A. R. Alotaibi and A. V. Mishra, "Global and regional volatility spillovers to GCC stock markets", *Economic Modelling*, 2015, 45, 38 ff.; A. Ali, "Analysis of the Determinants of Capital Adequacy Ratio: The case of Full-Fledged Islamic Banks in the Gulf Cooperation Council (GCC)", *European Journal of Islamic Finance*, 2019, 14, 1 ff.; A. A. Adewale and Others, "Risk-Based Supervision in Islamic Banking", IFSB Working Paper Series, WP-15/12/2020.

[59] IFSB-19 Guiding Principles on Disclosure Requirements for Islamic Capital Market Products, 13 ff. e 24 ff.; S. F. Najeeb and M. M. Mustafa, "Strengthening the Financial Safety Net: The Role and Mechanisms of Sharì ah-Compliant Deposit Insurance Schemes (SCDIS)", IFSB Working Paper Series, 6 March 2016, 1 ff.; IFSB, TN-3 Technical Note on Financial Inclusion and Islamic Finance, December 2019.

[60] S. Archer, R. A. A. Karim, V. Sundararajan, "Supervisory, regulatory, and capital adequacy implications of profit-sharing investment accounts in Islamic finance", *Journal of Islamic Accounting and Business Research*, 2010, n. 1, 10 ff.

[61] M. Casoria, "La regolamentazione concorrenziale nei paesi arabi del Golfo. Cronache di un enforcement abbozzato", *Mercato, concorrenza, regole*, 2018, 401 ff.

[62] The Central Bank of Bahrain adopted CBB Rulebook, http://cbb.complinet.com/cbb/microsite/cbb\_rulebook.htm, and the Central Bank of Oman adopted the Islamic Banking Regulatory Framework in 2012, at http://www.cbooman.org/news/IBRF.pdf.

[63] Dubai Financial Services Authority. About Islamic banking in UK: A. K. Aldohni, "Soft law, self-regulation and cultural sensitivity: The case of regulating Islamic banking in the UK", *Journal of Banking Regulation*, 2014, n. 2, 164 ff.; E. S. Housby, Islamic financial services in the United Kingdom, Edinburgh: Edinburgh University Press, 2011.

[64] H. Ahmed, "Basel III liquidity requirement ratios and Islamic banking", cit., 253 ff.

[65] Basel Committee on Banking Supervision, Assessment of Basel III - risk-based capital regulations - Saudi Arabia, September 2015; "Regulatory Consistency Assessment Program (RCAP)", Assessment of Basel III Liquidity Coverage Ratio regulations - Saudi Arabia, September 2015, at *www.bis.org*. In May 2016, the Capital Market Authority announced changes to incentivise further foreign investment (in *GCC Quarterly Review*, Q2, 2016). Since January 2017, listed companies are required to use the standards prepared by the International Financial Reporting

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6424 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index

Standards for the financial reporting framework (in *GCC Quarterly Review*, Q4, 2016, at *www.linklaters.com*).

[66] See Dow Jones Islamic Market Indices Methodology. December 2017, at https://us.spindices.com. In 1999, in Bahrain, the Dow Jones Islamic Market Indexes provided for the first shariah indices and offers a timely list of almost 70 shariah-compliant measures covering equities and fixed income securities around the World. See S&P Dow Jones Islamic Market World Index, 31 August 2021; M. Campra, S. Pucci, V. Brescia, "Can the Dow Jones Sustainable Index be useful for evaluating Dow Jones Islamic Market companies? European Journal of Islamic Finance, 2020, First Special Issue, 1 ff. The index measures the performance of stocks traded globally that pass rules-based screens for adherance to Shari'ah investment guidelines. About "shariah risk" see H. Visser, "Islamic Finance in Non-Muslim-Majority Jurisdictions: Regulatory Issues", cit., 88 ff. About the role of the Auditing Organisation for Islamic Financial Institution, see T. V. Russo, "I contratti Shari'a compliant, Valori religiosi e meritevolezza degli interessi", cit., 49 ff.

[67] M. Ashraf M. and A. Lahsasna, "Proposal for a new Sharì ah risk rating approach for Islamic banks", *International Journal of Islamic Finance*, 2017, 1, 87 ff.; A. Musleh Alsartawi, "Performance of Islamic banks: Do the frequency of Shari'ah supervisory board meetings and independence matter?", *ISRA International Journal of Islamic Finance*, 2019, vol. 11, No. 2, 303-321.

[68] The Central Bank of Bahrain has issued capital adequacy regulations (Circular n. 52/2017, which came into force on 1° February 2017 and fully implemented in 2019) in line with Basel III, see *GCC Quarterly Review*, Q1, 2017, at *www.linklaters.com*. About the role of central banks, see A. A. Hossain, Central Banking and Monetary Policy in Muslim-Majority Countries, Cheltenham: E. Elgar, 2015.

[69] E. Aracil, "Corporate social responsibility of Islamic and conventional banks: The influence of institutions in emerging countries", *International Journal of Emerging Markets*, 2019, 14 (4), 2019, 582-600; I. A. Oladapo et al., "Perception of stakeholders on governance dimensions of the Islamic banking sector", *International Journal of Emerging Markets*, 2019, 14(4), 601-619; S. El-Halaby et al., "The non-economic consequences of disclosure in Islamic banks", *International Journal of Emerging Markets*, 2018, 13(6), 1948-1968. About jurisprudence see S. Coopchik, "Judicial Decision-Making in Islamic Banking and Finance", *European Journal of Islamic Finance*, 2015, 2, 1 ff.

[70] See the Italian banking law "Testo Unico Bancario (T.U.B.)", especially artt. 10 et seq.; T. V. Russo, I contratti Shari'a compliant, cit., 140 ff. About collection of savings, the principle of the obligation to repay under art. 11 T.U.B. is contrary to the typically Islamic current accounts which exclude it, but combinations of contracts would be possible,

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6424 - Published by University of Turin <a href="https://www.ojs.unito.it/index.php/EJIF/index">https://www.ojs.unito.it/index.php/EJIF/index</a>

EJIF content is licensed under a Creative Commons Attribution 4.0 International License

obviously according to mechanisms which would have to be accepted by the supervisory authorities, see F. Miglietta and P. P. Rampino, Gli strumenti di finanza islamica a sostegno del Sistema Italia, cit., 69-71. On banking activities see G. Gimigliano, "Art. 10", in S. Bonfatti (a cura di), Commentario al Testo Unico Bancario, Pisa: Pacini, 2021, 41 ff., and banking authorisation see A. Benocci, "Art. 14", ivi, 74 ff.; C. Brescia Morra, Il diritto delle banche, Le regole dell'attività, Bologna: il Mulino, 2020.

[71] European Commission, "Banking Package 2021: new EU rules to strengthen banks' resilience and better prepare for the future", 27 October 2021, for the implementation of Basel III (also known as Basel IV) agreement that will involve the amendment of Directive 2013/36/EU, Reg. (EU) n. 575/2013 and separate proposals about the Capital Requirements Regulation (CRR) and Capital Requirements Directive (CRD IV), at *ec.europa.eu*. On amendments (Commission Delegated Regulation (EU), 20 October 2021, n. 439/2022) to the CRR see "European Banking Institute Report on Economic Policy and Financial Regulation Measures: International, EU and Euro Area Levels", 19 March 2022, at www.ebi-europa.eu.

[72] U. F. Moghul, A socially responsible Islamic finance: Character and the common good, London: Palgrave Macmillan, 2017; F. Miglietta and P. P. Rampino, Gli strumenti di finanza islamica a sostegno del sistema Italia, cit., 79 ff.; Forum Finanza Sostenibile (2019), "L'Unione europea e la finanza sostenibile, Impatti e prospettive per il mercato italiano", at *www.finanzasostenbile.it*.

[73] S. Khavarinezhad and P. P. Biancone, "Future Trends and Finance Approaches in Islamic Banking", *European Journal of Islamic Finance*, 2019, 12, 1 ff.; S. Khavarinezhad, P. P. Biancone, V. Jafari-Sadeghi, "Financing in the Islamic System and Sustainable Economic Development of Selected Islamic Countries", *European Journal of Islamic Finance*, 2021, 19, 18-23; V. Brescia, A. Adam Sa'ad, R. Bt Hassan, S. Musa Bin Syed Jaafar Alhabshi, F. Lanzalonga, "Exploring sustainability from the Islamic finance perspective", European Journal of Islamic Finance, 2021, 19, 45 ff.; F. Lanzara, "Islamic finance and Sustainable Development Goals. A bibliometric analysis from 2000 to 2021", *European Journal of Islamic Finance*, 2021, 18, 1 ff.

# Recent trend of deposits in Islamic banks

Davidia Zucchelli<sup>1</sup>

1. Senior Economist, Research Department, Intesa Sanpaolo and AIAF Analyst and research fellow at the Catholic University of Milan, Italy

Corresponding author: davidia.zucchelli@intesasanpaolo.com

Abstract— Bank deposits increased sharply in 2020, continuing in 2021, in countries with advanced economies and emerging countries alike. The pandemic caused savings to increase for precautionary reasons. At the same time, the lockdown compressed spending, but deposits are now a financial resource that could be used to boost the recovery of consumption and investments. There was an increase in deposits in Islamic countries, but this was less marked. The fall in GDP in 2020 in all of the Islamic countries considered, with the sole exception of Bangladesh (+5%), was met with an increase in deposits, but not such as to be thought of exceptional magnitude. An acceleration was recorded in Indonesia, Bangladesh and Saudi Arabia, with increases only moderately higher than in the previous year. Several drivers can explain this trend, some of which are attributable to Islamic principles. In assessing the role of Islamic finance and bank deposits, financial inclusion is also critical. Despite gradually increasing, the level of financial inclusion, which is high in countries with the highest income (the Gulf countries), shows a large margin for potential growth. Especially in the less rich Islamic countries, there exists a problem of mobilising resources and providing a custody function for the financial resources of households and businesses. The proliferation of fintech channels, primarily through mobile phones, can contribute significantly to financial inclusion, particularly among young people. Prospects are positive thanks to the confluence of some factors (such as the strong demand for both Sharia-compliant and ESG products).

# Keywords: bank deposit, Islamic finance, Islamic banks.

### I. SHARP INCREASE IN SAVINGS IN ADVANCED AND EMERGING ECONOMIES IN 2020

During 2020, there was a sharp increase in savings in countries with advanced economies and emerging countries alike, as deeply analysed by the IMF (2021) [1]. A large part of savings was banked, this being the most immediate way that businesses and households allocate excess liquid funds, while waiting to define their use more clearly. The main drivers supporting deposits were primarily the impossibility of making purchases due to the lockdown, as well as uncertainty about the

evolution of the pandemic and therefore the effects of the pandemic on the economy, labour, wages and salaries.

These liquid funds are expected to be used for the consumption of goods and services and investments, which are expected to gradually recover in 2021, as well as for financial products (government bonds, funds, etc.) or for house purchases, including, to some extent, in emerging countries.

The deposits in Islamic finance have some distinctive features that substantially differentiate them from the deposits collected by conventional banks. After considering the continued presence and weight of Islamic finance in the world, this paper outlines the technical peculiarities of the deposits and their objectives. An analysis of the financial statement data of a significant sample of Islamic banks in countries where Islamic finance is applied to a considerable extent, shows their recent dynamics and allows for assessment of the effects of COVID-19 in the sector. Lastly, the examination of the level of financial inclusion makes it possible to assess the spread of bank accounts and the ability of these systems to attract deposits and capture the development opportunities and the main evolutionary trends.

# II. THE MUSLIM PRESENCE IN THE WORLD AND ISLAMIC FINANCE

Various factors have contributed to the strengthening of the Muslim presence in many countries in recent decades. The first and foremost among these are globalisation and immigration, which has led many Muslims to settle in various western countries. According to the indications on the CIA website [2], Islam is the second most practiced religion globally (by approximately 24% of the population), after Christianity which, in its main denominations (Roman Catholic Church, Protestant Church and Orthodox Church), covers more than 30% of the population.

In the Emerging Europe, there is significant Muslim presence in Albania and Bosnia Herzegovina (over 50% of the population). In the countries of the former Yugoslavia, North

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6260 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index

Macedonia and Kosovo are also predominantly Muslim. A minority is present in Bulgaria.

A significant minority is present in Russia, estimated to be between 10 and 15%. Again, according to CIA's figures, is growing due to the greater birth rate of the Muslim minorities (such as the Chechens) compared to the Orthodox population.

In the world, the main countries with the largest Muslim populations are the Gulf countries, with Iran and Iraq, as well as in Asia, Pakistan, Bangladesh, Indonesia and Malaysia.

The spread of Islamic finance - defined as the activity of financial institutions that base their objectives and operations on the principles of the Koran - reflects the spread of the religion, but with significant particularities. 235 banks that cover this area are operating worldwide (according to the BankFocus Moody's BvD database), 9 of which are in the European Union (27 countries).

The Gulf countries account for the highest share of the total assets of the world's Islamic banks (45.4% at the end of 2019) according to the Islamic Financial Services Industry Stability Report (2020) published by the Islamic Financial Services Board [3]. The Middle East and South Asia (MESA) area covered a share close to 26% on the same date, while the South East Asia region accounted for 23.5%. Even when adding North Africa and Sub-Saharan countries, the African region's share of the world total is marginal (1.6%). Breaking down the individual countries further, Iran's share is the highest (about 29% of the total), followed by Saudi Arabia, which has a slightly lower share (25%).



Figure 1. Islamic Finance (breakdown of TA between geographical areas, 2019). Note: MESA includes Afghanistan, Bangladesh, Iran, Iraq, Jordan, Lebanon, the Maldives, Palestine, Pakistan and Sri Lanka. Source: our calculations based on IFSB figures.

It is important to consider the regional distinction. This is because Islam differs among geographical areas and countries (first of foremost with the distinction between Shiites and Sunnis), with a different interpretation of the provisions of the Sharia, the Islamic Law [4].

Figure 2. Islamic Finance (breakdown of TA, 2019). Source: our calculations based on IFSB figures.

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6260 - Published by University of Turin https://www.ojs.unito.it/index.php/EJIF/index

EJIF content is licensed under a Creative Commons Attribution 4.0 International License

Received: 22/11/2021 - Accepted for publication: 03/03/2022 - Published: 22/04/2022



The share that Islamic finance has compared to traditional finance in the individual countries is very significant. As highlighted by HRH Prince M. Al-Faisal Al-Saud (2000) [5] in providing a wide and in-depth examination of Islamic economic principles, the gradual spread of Islamic finance is part of the process defined in Muslim circles as 'Islamic resurgence', aimed at creating a new, united Islamic society that avoids 'fundamentalism' and extremist positions.

In Iran and Sudan, where the State has Islamised the financial sector, this share is 100%. Among the other countries, there is no perfect correspondence between the size of the sector (in USD Bn) and the share covered by Islamic banks. Saudi Arabia has a significant share of over 60% and high volumes of around USD 450Bn. On the other hand, in Kuwait the share of the total financial system is close to 50%, with a smaller volume (USD 100Bn). Other countries, such as Turkey and Indonesia, with volumes falling within the ranking of the major Islamic countries in the world, have modest shares compared to the conventional system (around 6%).



Figure 3. Islamic banks (TA and % share of the total assets of the country's banking system, 2019). Source: our calculations based on IFSB figures.

Egypt is a very important exception. With an almost total Sunni Muslim presence (with the Copts accounting for about 10%) of the population, the weight of Islamic finance in Egypt is still negligible (with a share of total world assets that fails to reach 1%), and the level of financial depth calculated using the main indicators referring to traditional finance also remains modest. A few Sharia-compliant banks were involved in foreign currency speculative transactions in the 1980s, which led to their closing. Subsequently, public banks have assumed a predominant role within the system, and they have held on to this role to date. In the country, only 3 Islamic banks are active, controlled by banks based in the Gulf, out of a total of 38 institutions. The structural Loan/GDP ratio also remains very low (just over 20% at the end of 2020) compared to peers, despite the presence of numerous international operators, while the ratio for deposits is higher (over 60% at the end of 2020).

Islamic banks in Bosnia accounted for a share of the country's total assets equal to about 3% at the end of 2019. Among other countries close to Europe, such as Russia, the Muslim presence is not reflected in a corresponding proliferation of Islamic financial instruments, as this presence consists of ethnic minorities whose role in the country's economy remains modest.

### III. DEPOSITS IN ISLAMIC BANKS: TECHNICAL CHARACTERISTICS

There are numerous differences between the principles of European/Western and Islamic finance. These differences pertain to the objectives that can be pursued and therefore their technical characteristics.

Article 14 of the Italian Consolidated Law on Banking (and the European Directive 2006/48, Article 4) provides authorisation to carry out banking activities which, in a "traditional" sense, is focused on the dual function of collecting deposits and loan granting, and Article 11 states "for the purposes of this legislative decree, the acquisition of funds with the obligation to repay...". M. Fahim Khan and M. Porzio (2010) [6], especially in the introduction, explain the way Islamic banks are incorporated into European regulations.

Like conventional banks, Islamic banks receive deposits and finance customers, businesses and households, though with significant specific characteristics. In almost all cases, the collection and financing scheme of Islamic banks does not entail the obligation to repay. The operations of these financial institutions are therefore more complex and this has made it necessary to develop specific international standards to be followed for the preparation of the financial statements of these institutions (P. Biancone, 2014) [7].

The Islamic finance model is based on a different type of financial intermediation activity of banks based on the principle of profit and loss sharing (PLS), i.e. participation in profits and losses deriving from its customers' investments. Income from the interest typical of Western financial systems is replaced by income from assets compatible with Sharia requirements. It is interesting to note in this regard that the bank must adequately distinguish the proceeds obtained from projects financed with its customers' resources, from the profits obtained from its own capital, as explained by Z. Ahmad Khan (2000) [8].

Like all traditional banks, the Islamic Bank transfers financial resources from sectors with financial surplus to sectors with a deficit, but it also participates in business projects and may also promote business initiatives and actively partner with corporate customers.

The assets and liabilities of an Islamic bank differ significantly from those of a traditional bank. In addition to

European Journal of Islamic Finance - ISSN: 2421-2172

DOI: 10.13135/2421-2172/6260 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index EJIF content is licensed under a Creative Commons Attribution 4.0 International License

Received: 22/11/2021 - Accepted for publication: 03/03/2022 - Published: 22/04/2022

working capital, the bank's assets include also equity investments in companies that entail the sharing of profits and losses, in accordance with Sharia principles (especially mudarabah, musharaka and murabaha contracts).

In liabilities, among funding instruments, G.M. Piccinelli illustrates (in Khan, Porzio, 2010) [9], Islamic banks offer three main categories of accounts, which essentially differ for the repayment method and the allocation of the amounts collected:

Non-remunerated on-demand deposits in current accounts, in turn concern three types of deposit accounts (current accounts): the amanah or trust deposit, in which the bank acts as trustee and the gard hassan or good loan, where the bank shall return only the principal; the wadiah, or safe deposit, is a custody contract whereby the bank returns the money at the customer's request. In all three cases, the bank obtains authorisation to invest the money deposited in Shariacompliant assets and does not pay out any form of interest or shares the profits with depositors. It only reserves the right to distribute gifts (hiba) in cash or in kind at its discretion, in order to render these deposits more attractive and competitive. They are mainly used by customers to manage current payments. Due to its nature, this type of account must be backed by reserves of a value equivalent to the deposit (100%), and represent about 10% of liabilities, an average that is, however, only indicative in a highly diverse context;

Investment accounts concern deposits for which the bank does not ensure reimbursement. Funds collected in these types of accounts are invested in risky activities managed by the bank, which acts as an entrepreneur (rabb-ul-mal) or as investor on the stock market. Depending on the degree of freedom the bank has insofar as the investment activity, the investment accounts may be divided into:

restricted investment accounts;

unrestricted investment accounts.

Banks have the highest investment freedom with the first category for which customers are only provided with the general information concerning the investment of their funds, while in the latter category the bank is bound by the depositor to invest only in some types of activities or to observe specific investment methods. In any case, the customers' return is based on the results achieved and therefore they become a sort of shareholder in some projects. They represent about 70% of the bank's total funds.

Special investment accounts or Profit-sharing investment accounts, i.e. non-guaranteed term deposits. These generally set a minimum initial amount and allow the customer to withdraw only periodically following adequate notice to the bank. They can guarantee an ex-post return through profit and loss sharing, i.e. the equitable sharing of profits and losses of the financial asset. The main profit and loss sharing include the aforementioned mudarabah and musharaka. These three types of deposits are recognised in the financial statements under liabilities together with reserves and share capital.

#### IV. PERFORMANCE OF DEPOSITS IN ISLAMIC BANKS

The analysis of data referring to Islamic banks - as highlighted by numerous scholars, as in particular I. Warde (2000) [10] and the central authorities - requires a number of important prerequisites. The relative "youth" of financial institutions and the products they offer, and therefore the uncertainties inherent in the interpretation of their technical and legal characteristics, also give rise to several accounting problems. First of all, operations have been ongoing only over several decades and they are handled differently by the various countries, which renders any analysis and comparison between them difficult. There are quite a few differences in the interpretation concerning the use and accounting of certain transactions.

Moreover, when interpreting the data, the financial and religious aspects are not always perfectly aligned. Regardless of whether or not they evaluate opportunities, Muslims concurrently rely on Islamic banks and conventional banks, and non-Muslims can bank with Islamic banks, to diversify their investments and financial operators that have specific skills and professionalism.

That said, the analysis of the financial statement data of a significant sample of Islamic banks, in countries with an Islamic majority, shows an acceleration in deposits during 2020. However, there is no real surge as was the case in many Western countries. In fact, the fall in GDP in 2020 in all of the Islamic countries considered, with the sole exception of Bangladesh (+5%), was met with an increase in deposits, but not such as to be considered of exceptional magnitude. An acceleration was recorded in Indonesia, Bangladesh and Saudi Arabia, with increases only moderately higher than in the previous year. On the other hand, Oman, Kuwait and Qatar slowed down.

Also, in the three other significant countries contained in the following table, the trend in deposits (over the entire banking system) does not appear to be particularly high, given a decline in GDP. The figures for Egypt are unique, with the resilience of real GDP (3.6%) and a significant increase in deposits, but in line with previous years (in nominal terms).

The comparison with the 2009 crisis is not helpful insofar as the interpretation, also due to the lack of data. As is known, the subprime mortgages storm had limited direct effects on Islamic banks, as they had no exposure to that segment. In 2009, deposits accelerated, especially in Malaysia and Egypt, and held up in Saudi Arabia. This seems to be in line with the data for the last two years, which may help to exclude the assumption of an abnormal trend in deposits in 2020.

The examination of some technical characteristics and functions of the deposits can help explain their growth, despite the decline in GDP.

In Western countries, one of the drivers underlying time deposit dynamics are interest rates. However, this is a factor

that did not play a significant role in 2020, since interest rates were very low and decreasing. On the other hand, the precautionary reasons for saving money and the forcedly postponed consumption due to the lockdown measures adopted by governments were decisive in supporting bank deposits in the context of the crisis.

	2008		2009		2019		2020	
	Dep. loc.curr.	Real GDP	Dep. loc.curr.	Real GDP	Dep. loc.curr.	Real GDP	Dep. loc.curr.	Real GDF
Indonesia	5.9	7.4	40.4	4.7	9.45	5	12.66	-2
Bangladesh	ND	6	n.d.	5	9.1	8.2	13.5	5.2
Malaysia	4	4.8	16.8	-1.5	12.2	4.3	6.2	-5.6
S. Arabia	24.7	6.2	22.5	-2.1	8.2	0.3	20.6	-4.1
UAE	27.4	3.2	7.7	-5.2	-2.7	1.7	13.2	-5.5
Oman	ND	8.2	ND	6.1	21.4	-0.8	15.7	-6.2
Kuwait	ND	2.5	ND	-7.1	15.5	0.4	11.7	-5
Qatar	ND	17.7	ND	12	16.1	0.8	10.1	-3
Egypt	6.3*	7.1	13.1*	4.7	17.01*	5.6	16.1**	3.6
Albania	2.6*	7.5	6.8*	3.3	3.8*	2.2	8.1*	-3.3
Bosnia H.	5.5*	3.5	3.8*	-2.3	8.4*	2.91	6.5*	-5.5

Table 1. Deposits in Islamic banks and real GDP in the major Islamic countries (yoy % change). Note: (\*) central banks' database referring to the entire banking system; (\*\*) Source: EIU; NA= not available. Source: our calculations on EIU and BankFocus BvD data.

In Islamic countries, the fact that the payment of interest is not permitted, though customers can participate in the bank's results may have prompted deposits by customers, households and businesses, on the basis of the results expected by the bank (according to the participatory model described in the previous paragraph). An empirical analysis regarding Malaysia and Turkey and elaborated by S. Cevik and J. Charap (2011) [11] revealed namely a high degree of correlation between conventional deposit rates and the rate of return Islamic profitand-loss sharing accounts.

At the same time, very low interest rates from conventional banks also in Islamic countries may have limited the diversion of at least a part of the deposits to these banks by Muslims who do not fully follow Sharia rules, with the aim of a return. As highlighted by B. Akhtar, W. Akhter, M. Shahbaz (2017) [12] with regards to Pakistan, it is observed that any changes in the rate of interest or in the rate of profit of Islamic bank significantly affect the level of deposit in conventional as well as that of Islamic banks. Consequently, customers of Islamic banks appear to be motivated by profit. The conventional bank interest rate has an impact on the customers of both conventional and Islamic systems. A boost in the interest rate of the conventional banks will increase the level of deposits at conventional banks and decrease the deposits of Islamic banks.

The fact that non-Muslim customers can use of the services of an Islamic bank, which in turn can offer its products to non-Muslim customers, is still up for discussion. Far from expressing an assessment of Muslim consistency, it represents a critical issue that is openly pondered by Muslim scholars. That many issues and doubts regarding the admissibility of certain behaviours in the light of Sharia remain open is actually part of the gradual process of understanding their Islamic identity. It is therefore not possible to determine how many customers of Islamic banks are actually Muslim. In E. Smolo, M. Šeho and M. Kabir Hassan (2020), among others, we read: "The term 'Islamic finance' may persuade that it is just for Muslim countries with greater number of Muslim populations.

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6260 - Published by University of Turin <u>https://www.ojs.unito.it/index.php/EJIF/index</u> EJIF content is licensed under a Creative Commons Attribution 4.0 International License Nevertheless, it is been contended that larger part of Islamic fund clients are really non-Muslims." [13]

A further factor that can explain the trend of deposits, which is strengthening, but to a rather modest extent, may be the need to avoid an excessive amassment or accumulation of financial resources, which is explicitly not permitted by Islamic law. In other words, money is only valuable as a useful tool for development and growth, and should circulate rather than being set aside.

Due to the crisis generated by COVID-19, a high level of liquidity has been created in Western countries. On the other hand, the need to implement strategies to mobilise financial resources persists in Islamic countries, especially the less rich ones.

In the current context of strong liquidity, Western banks are able to dispose of funds that are in excess of credit demand. The first consequence, which is rapidly spreading among banks, is the application of negative interest on deposits of a high amount (above 100,000 euro). But cases where the banks diverge their customers towards other forms of investment (to assets under management) also abound, as highlighted in I. Moise (2021) for example [14]. They also do this to reduce the effect of this excessive liquidity on their capital requirements and the charges into the deposit guarantee scheme.

In Islamic countries, these terms do not seem to apply. Especially in the less rich Islamic countries, there exists a problem of mobilising resources and providing a custody function for the financial resources of households and businesses. Both of these functions are of considerable importance in emerging countries, which are still working towards implementing their industrial and production structure. It is worth remembering that in Italy, it was deemed necessary to provide constitutional protection to savings. Article 47 reads "The Republic encourages and protects saving in all its forms. It regulates, coordinates and oversees the operation of credit".

In fact, the incentive to attract savings measured by the level of financial inclusion is a primary objective pursued by the central authorities, since this is an indispensable tool for the development of a country, as analysed by K.B. Jedidia and F. Boubakri (2018) [15] with particular regards to Tunisia. The Gulf countries also need financial resources for the creation of a growing diversification of the local production system, one that is not only based on oil revenues, but also on renewable energy.

Although it is not useful to explain the dynamics of deposits, it should also be remembered that in Islamic countries, particular attention is paid to the origin of the sums deposited, as the activities that gave rise to those amounts must be Sharia-compliant. If the sums held for deposit originate from activities that are not Sharia-compliant, the banks cannot withdraw and use them [16].

### V. LIQUIDITY (IS NOT) ACTIVELY SOUGHT

Islamic banks often find themselves holding excessive liquidity due to the numerous restrictions limiting the granting of credit (linked to the technical forms allowed and the risks taken). They route these funds to the interbank market, and often even to conventional banks, generating quite a bit of criticism from the most devout Muslims. In fact, the World Bank and IDBG (2018) [17], among international institutions, point out that funds originating in Islamic countries are mainly sent to banks in OECD countries, with estimates for sovereign funds equal to USD 3Trn. By contrast, Islamic capital markets include an outstanding amount of sukuks and wealth management funds totalling USD 671Bn.

However, liquidity management is structurally complex in Islamic banks, including because they cannot use funds from interest bearing accounts to subscribe for government bonds or hold reserves at central banks. These provisions were particularly binding especially with the application of Basel III provisions on liquidity. Furthermore, the possibility of using sukuks was allowed under certain conditions for refinancing purposes at central banks (eligibility).

### VI. NEITHER INTEREST (RIBA) NOR DEBT

The main prohibitions concerning the Islamic economy and finance, as mentioned above, include prohibition of the interest (riba). This is an aspect worthy of further analysis.

The term "riba" literally means "increase", "excess" and "growth" in Arabic. The notion of riba has been widely studied and discussed among Islamic schools of jurisprudence and a variety of interpretations and explanations have been attached to it. Moreover, all monotheistic religions have historically condemned usury. The creation of riba responds to the objective of achieving conditions of fairness and economic and social justice by preventing every form of privilege, through the earning of financial returns, without participation in the work effort involved. R. Hamaui, M. Mauri (2009) [18] offer an in-depth analysis on the interpretation of this ban and on certain related problems (in terms of operations, accounting, etc.).

Islamic reflection notes that the main reason for prohibiting interest is to protect the poor. Since money and assets generate interest, the poor cannot obtain interest and accumulate wealth, while the rich can generate wealth from wealth, through interest. However, this position too was not accepted without criticism and reservations. IIBI (1993) [19] raised several – still open - questions. Some aspects are still under discussion today. In particular, the question arises of whether for personal funds only the compound interest must be considered, thereby allowing simple interest, and whether the ban also applies to Muslims living in non-Islamic countries. Islamic banks have actually sought to diversify deposits, providing modest amounts with insurance as well.

In addition, according to Islamic thinking, earning interest discourages donations, as those who obtain interest are less motivated and encouraged to create value and therefore less motivated to build a cohesive society, for the benefit of everyone. It is also believed that interest-based finance leads to many inefficiencies, i.e. high debt and hazardous risks linked to trading on financial markets, which expose economies to instability. On the other hand, sharing risks and results allows the parties involved to minimise the risk and obtain a common benefit. The assumption is that all participate in the risk of

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6260 - Published by University of Turin https://www.ojs.unito.it/index.php/EJIF/index

default and are therefore encouraged to work together toward a common objective. The parties and therefore society as a whole are therefore protected from fraudulent activities and social tensions.

The prohibition of riba prohibits all forms of risk-free "increases" set beforehand, which Western thinking believes to be more "damaging" to investors. In fact, this view is opposite to the typical Western economic thinking, according to which participation in profits entails the assumption of risks that are higher than the interest rate applied, even if it is variable.

Also strictly related to the riba prohibition is the prohibition against using debt. While the level of debt currently plagues many Western countries, companies and states that have undertaken substantial measures to support the economy due to COVID-19, debt is absent from the Islamic system.

As is known, debt is favoured over equity investments in Western countries because it allows issuers to deduct interest paid for tax purposes, as they are costs, but also due to asymmetric information which discourages many investors from assuming capital risks. Leverage obviously mitigates this situation, but if it is too high, it may generate instability risks.

Financial theory has dealt with this issue widely. Here we will examine only some aspects that can explain at least part of the aversion to debt, compared to a partnership relationship with the investor.

First of all, collateral guarantees - which are prohibited by Islamic law - are required by conventional banks, but can lead to a less than optimal allocation of resources in the case of small and medium-sized businesses, even though they do pave the way for interesting projects.

The debt is also unsuitable for financing businesses, especially when the results are highly uncertain, or if the company is in the start-up phase. The debt finally weighs heavily on the economic return and therefore on the project's success.

Similarly, although a model for sharing profits and losses may lead to greater monitoring costs, including on account of the presence of the Bank's representatives in the Board of Directors, it is considered important in Islamic countries for the purpose of the proper functioning of the capital market as it favours social cohesion. When the relationship between entrepreneur and investors is direct and of an ongoing nature, it leads to better risk control and thus to better relations and financial results.

In order to finance themselves, in recent years Islamic banks have increasingly made use of sukuk, which is a form of securitisation of their loans, including on international markets. The peculiarities of this market segment do not fall within the objectives of this sheet.

### VII. FINANCIAL INCLUSION IN ISLAMIC COUNTRIES PAINTS A COMPOSITE PICTURE

An aspect that is fundamental in assessing the role played by Islamic finance and bank deposits - as in Western financial systems - is the level of financial inclusion. In other words, an essential requirement is the examination of the spread of current accounts, which are at the basis of both time deposits and loans, as highlighted by S.B. Naceur, A. Barajas, and A. Massara (2015) [20].

Financial inclusion - measured by the share of people and businesses that use the financial products offered by banks and other intermediaries - is the essential indicator of the proliferation and use of these financial products. Various international institutions - in particular the World Bank, IMF, OECD, IIF - are exploring this issue. The following analysis is based on the Global Financial Inclusion database (Global Findex) published by the World Bank on the basis of surveys conducted in 148 countries over several years (2011, 2014, 2017) [21]. A further update was expected in 2020, but it was postponed due to the COVID-19 crisis. The latest survey in 2017 involved over 150,000 people over 15 years of age. Availability and access to these products are not synonymous, since they may be available but not accessible due to various barriers (such as high costs, complexity of procedures, distance, required documents, etc.).

As shown in Figure N°. 4, the World Bank data confirms a gradual growth in the spread of bank current accounts on the three reporting dates, 2011, 2014 and 2017, in the major Islamic countries.

Inclusion is very high in the Gulf countries, involving more than 80% of the adult population in the United Arab Emirates and Kuwait. Also in Malaysia, the spread in percentage terms is very high (85%), but in the other Asian countries the percentages appear to be much more modest, though they are rising (around 50% in Bangladesh and Indonesia, and stable at 21% in Pakistan), mainly on account of the more limited income levels. The countries with the highest shares are High Income countries, as well as Iran among the Upper Middle-Income countries, while the other countries that have lower shares belong to the Lower Middle-Income category.

It should be noted that in all the countries listed in the chart - those where Islamic finance plays a significant role or prevails - the level of inclusion appears to be lower than the international average of the relevant income category, defined considering all financial intermediaries, mainly Western/conventional. Iran is a significant exception, since with a system wholly made up of Islamic banks, it covers about a third of the total assets of the world's entire Islamic sector while showing a very high level of inclusion.

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6260 - Published by University of Turin <u>https://www.ojs.unito.it/index.php/EJIF/index</u> EJIF content is licensed under a Creative Commons Attribution 4.0 International License



Figure 4. Adults with a bank current account (in some Islamic countries, % of adults, over 15 years of age). Notes: LMI=Lower middle income, UMI=Upper Middle Income, HI=High income. (\*) average of the country income category. Source: our calculations based on World Bank and Global Findex database figures.

Various studies - in particular A. Demirguc-Kunt, K. Leora, R. Douglas (2013) [22] - showed that, in general, in Islamic countries the degree of proliferation of savings instruments is lower than that of loans. This can be explained, on the one hand, by the special ability Islamic finance has insofar supporting small and medium-sized businesses. The demand for financial resources, especially in this category of businesses, may be encouraged when risk sharing is concerned, which is typical of operators in Islamic countries, where private equity and venture capital are extremely widespread. This result is significant. According to a study by the World Bank (2014) [23], the higher the number of Islamic banks per 100,000 inhabitants, the lower the share of companies that find it difficult to access credit. In Western countries, where financial systems have always been criticised for credit access by SMEs, the presence of adequate guarantees and the credit history of the borrower is a major consideration.

In terms of deposits, the reduced proliferation may be partly due to the fact that savers may be less incentivised to deposit their money (due to the lack of returns and forms of deposit insurance), but also because of a lower level of disposable income, as mentioned previously. Moreover, even if income is significant, as in the Gulf countries, the income concentration is very high and therefore the level of distribution of financial products (measured by the number of people) may be lower than that recorded in Western countries.

Egypt has additional peculiarities. Despite modest financial inclusion, the deposits/GDP ratio is high (around 60%), but the loans/GDP ratio is just over 20%. The role of public banks is crucial, both during the deposit-taking phase, by offering reinsurance to depositors, and in terms of assets due to the significant weight of subscribed government bonds, with the public system therefore engaged in incisive crowding-out action.

The breakdown by gender shows that it has been reasonable to expect a lower level of financial inclusion of women compared to men. However, it should be noted that also among women, financial inclusion has strengthened significantly in recent years, across all countries, especially in Egypt (from 7% in 2011 to 27% in 2017). The gender gap remained high in Bangladesh and Pakistan (both approximately 30pp in 2017). Surprisingly, these 2017 figures indicate that the gender gap was still equal to about 30pp in Turkey.



Figure 5. Women with a bank current account (in some Islamic countries, % of adults, over 15 years of age). Notes: LMI=Lower middle income, UMI=Upper Middle Income, HI=High income. (\*) average of the country income category. Source: our calculations based on World Bank and Global Findex database figures.

The main reasons why adults do not open a current account include a lack of money or the necessary documents, as well as the (physical) distance from financial institutions.

As highlighted in many studies (as in World Bank, 2020) [24], the proliferation of fintech channels can contribute significantly to financial inclusion. The Islamic population is also very young and includes half of the global population under the age of 34. Young people are not only IT proficient, they are also potential customers.

Among the various channels that can be used, the spread of fintech, especially through mobile phones, is expected in Islamic countries too, especially for payments. The World Bank estimates that two thirds of adults remain unbanked (around 1.7 billion), yet own a mobile phone that can be used to make payments. Most Islamic countries fall within the 50 countries with the highest penetration of mobile phones. Figure N° 6 shows the degree of smartphone penetration in a significant sample of Islamic countries. Indonesia has the highest share, accounting for 66% of the population, followed by Malaysia at 57.5% and Pakistan at 43% in 2018. Egypt has stopped at 28%. These percentages are high, but still far from the 96% average in advanced economies.



Figure 6. Smartphone penetration in some Islamic countries (2018). Source: Newzoo Global Mobile Market Report, September 2018.

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6260 - Published by University of Turin

https://www.ojs.unito.it/index.php/EJIF/index

### VIII. CONCLUDING REMARKS

Bank deposits increased sharply in 2020 and in 2021 in Islamic countries but not so strongly as in advanced economies. Nevertheless, some structural factors (such as financial inclusion) may support a furher increase in the next decade.

The numerous constraints that regulate the operations of Islamic banks may affect their capacity to expand. On this topic, the World Bank (2016) [25] offers a broad assessment of the role played by Islamic banks in the economies of their countries.

In 2020, Muslim countries were affected by the pandemic as well, with effects that weighed heavily on their economies, falling GDP, as mentioned above, with even the oil prices dropping. Now, as highlighted by rating agencies and international observers (as Moody's, 2021 [26], and A. Hauser, 2020) [27] prospects are still positive thanks to the confluence of some factors. First of all, demand for Sharia-compliant products is expected to remain strong, throughout the long term.

The fact that Sharia-compliant products will also be able to benefit from the growing focus on ESG products and sustainability, both of which have been historically safeguarded by Islamic principles, is important. K. Long (2020) [28] argues that the affinity between "Sharia-compliant" and "sustainability-compliant" finance is such that it has been proposed in the Islamic sphere to steer Islamic banks in a particularly determined manner towards this way of banking, which is highly appreciated by the community.

The support measures implemented by the central authorities also in these countries in response to COVID-19 will provide further support to "Islamic" activities. Moreover, in recent years numerous mergers have been carried out in the Gulf countries, with Islamic banks acquiring conventional banks, incorporating their business and expanding intermediated volumes represented by them, in many cases. All the mergers carried out in the Gulf in 2020 involved at least one Islamic bank.

As shown in Figure N° 7 covering total assets, the Islamic financial sector has posted an average increase of 6% over the last 5 years as compared to an average annual reduction of 1% in assets of conventional banks in Saudi Arabia.



Figure 7. Total Assets (CAGR, %, 2015-9/2020). Note: compound annual growth rate. Source: Moody's.

European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6260 - Published by University of Turin <u>https://www.ojs.unito.it/index.php/EJIF/index</u> EJIF content is licensed under a Creative Commons Attribution 4.0 International License

Received: 22/11/2021 - Accepted for publication: 03/03/2022 - Published: 22/04/2022

### REFERENCES

- [1] IMF, Global Financial Stability Outlook, April, 2021.
- [2] CIA website, available at https://www.cia.gov/.
- [3] Islamic Financial Services Board, "Islamic Financial Services Industry Stability Report 2020", 2020.
- [4] As is known, Islam is undergoing a constructive analysis of its own identity, deeply discussed among others in P. Nicelli, "Islam e modernità nel pensiero riformista islamico" [Islam and modernity in Islamic reformist thought], SanPaolo, 2009; and "Syed Muhammad Naquib Al-Attas and the Islamic Philosophy of Education", Encounter", Vol. 41/2, 2016).
- [5] HRH Prince M. Al-Faisal Al-Saud, "A State of Trusteeship", in Institute of Islamic Banking and Insurance, Anthology of Islamic Banking, 2000.
- [6] M. Fahim Khan, M. Porzio, "Islamic Banking and Finance in the European Union, A Challenge", Edward Elgar, 2010.
- [7] P. Biancone, "Scritti in onore di Pellegrino Capaldo" [Writings in honour of Pellegrino Capaldo], 2014. Based on the example of the IAS/IFRS for conventional banks, the standards issued by the AAOIFI provide the guidelines and criteria to be used in order to correctly and reliably represent the assets and income accrued during the year. However, this standardisation process requires numerous interventions by states with an Islamic majority, which must adhere to the principles issued by international Islamic institutions.
- [8] Z.Ahmad Khan, "Islamic Banking and its operations", Institute of Islamic Banking and Insurance, 2000.
- [9] G.M. Piccinelli, "The provision and management of savings: the clientpartner model", in Khan and Porzio (2010), 2010.
- [10] I. Warde, "Islamic Finance in the Global Economy", Edinburgh University Press, 2000.
- [11] S. Cevik and J. Charap, The Behavior of Conventional and Islamic Bank Deposit Returns in Malaysia and Turkey, IMF, WP/11/156. In 2010, a preliminary investigation showed furthermore that, three decades after its introduction, there remained substantial divergences between Islamic Banking and Finance's ideals and its practices, and much of Islamic Banking and Finance still remained functionally indistinguishable from conventional banking. See F. Khan, How 'Islamic' is Islamic Banking?, Journal of Economic Behavior & Organization, 76 (2010).
- [12] B. Akhtar, W. Akhter, and M. Shahbaz, (2017), "Determinants of deposits in conventional and Islamic banking: a case of an emerging economy", International Journal of Emerging Markets, Vol. 12 No. 2, pp. 296-309.
- [13] E. Smolo, M. Šeho and M. Kabir Hassan, "Development of Islamic Finance in Bosnia and Herzegovina", Journal of Economic Cooperation and Development, 41, 2020.
- [14] I. Moise, "Cash-rich banks seek to cut deposits, JPMorgan and Citi take step to avoid additional capital requirements", Financial Times, 5 May, 2021.
- [15] K.B. Jedidia and F. Boubakri, "Islamic Banking and the Perspectives of Savings Mobilization in Tunisia", Journal of Islamic Financial Studies, December, 2018.
- This provision is much more all encompassing than the anti-money [16] laundering legislation envisaged in Western countries, as a higher number of banned activities are envisaged. In the latter, if the sums derive from illegal activities, in accordance with anti-money laundering provisions, the bank shall block the account and report them to the competent authorities. As part of the beneficial owner audit information activity, the opening of ongoing relationships or the execution of occasional transactions is prohibited if: i) the customer refuses to provide information regarding the existence or not of beneficial owner; ii) the customer is unable to provide the information relating to the beneficial owner or cannot update such information and the information is not available from different and reliable sources; iii) the statement made by the customer and the evidence collected by the Bank (including as part of the credit process, in case of borrower) do not match and the customer confirms the indications provided without adequately showing the reasons why there are no beneficial owners or different persons than those declared to be beneficial owners are found to exist. In these cases, an assessment must be made regarding whether to initiate the suspicious

activity report procedure in accordance with the criteria defined in the anti-money laundering legislation in force.

- [17] World Bank and Islamic Development Bank Group, "Mobilizing Islamic Finance for Infrastructure Public-Private Parnerships", 2018.
- [18] R. Hamaui, M. Mauri, "Economia e finanza islamica" [Islamic economy and finance], Il Mulino, 2009.
- [19] Institute of Islamic Banking and Insurance (IIBI), "Islamic Banking and its problems", 1993, London.
- [20] S.B. Naceur, A. Barajas, and A. Massara, "Can Islamic Banking Increase Financial Inclusion?", IMF, WP/15/31, 2015.
- [21] World Bank, Global Financial Inclusion database (Global Findex), available at https://data.worldbank.org/products/data-books/little-databook-on-financial-development.
- [22] A. Demirguc-Kunt, K. Leora, R. Douglas, "Islamic finance and financial inclusion: measuring use of and demand for formal financial services among Muslim adults", World Bank, Policy Research Working Paper 6642/2013.

- [23] World Bank, "Global Financial Development Report 2014: Financial Inclusion", 2014.
- [24] World Bank, "Leveraging Islamic Fintech to improve financial inclusion", October 2020.
- [25] World Bank, "Islamic Finance. A catalyst for shared prosperity?", Global Report on Islamic Finance, 2016.
- [26] Moody's, "Growth prospects remain strong despite challenges from pandemic", February, 2021.
- [27] A. Hauser, "Why Islamic finance has an important role to play in supporting the recovery from COVID - and how the Bank of England's new Alternative Liquidity Facility can help", UK Islamic Finance Week 2020, December, 2020.
- [28] K. Long, "Islamic banking ripe for ESG rebrand", interview with Noripah Kamso, former chairwoman of Bank Rakyat in Malaysia, The Banker, 2020.



European Journal of Islamic Finance - ISSN: 2421-2172 DOI: 10.13135/2421-2172/6260 - Published by University of Turin <u>https://www.ojs.unito.it/index.php/EJIF/index</u> EJIF content is licensed under a Creative Commons Attribution 4.0 International License